Report No. FAA-EM-82-15

0

MA112

DTTC FILE COPY

EVALUATION OF VARIOUS NAVIGATION SYSTEM CONCEPTS

L Hogie K. Markin S. Toth



March 1.982

PREPARED FUR

U.S. DEPARTMENT OF TRANSPORTATION FEDERAL AVIATION ADMINISTRATION Office of Systems Engineering Management Washington, D.C. 20591 under Contract DTFA01-80-C-10030



Technical Report Documentation Page

.1

A The purpose of thi a particular set of nav rent airspace environme VHF Omnidirectional Ran tioning System (GPS), I (INS). In addition to tion system, considerat deficiencies existing i navigation systems are	is study is to identification systems and int. The navigation ge/Distance Measuring oppler navigation s detailed technical ion is also given to n the standards by measured.	tify the capabi d evaluate their n systems evaluating Equipment (system, and iner and operationation to the constrain which accuracy	lities and lim r performance ated are Loran VOR/DME), Glob rtial navigati l analyses of nts imposed by and effective	itations of in the cur- -C, Omega, al Posi- on system each naviga- and the ness of
A The purpose of thi a particular set of nav rent airspace environme VHF Omnidirectional Ran tioning System (GPS), I (INS). In addition to tion system, considerat deficiencies existing i navigation systems are	s study is to identify rigation systems and int. The navigation ge/Distance Measuring oppler navigation s detailed technical ion is also given to n the standards by measured.	tify the capabil d evaluate thei: n systems evaluating Equipment (system, and ine: and operational to the constrain which accuracy	lities and lim r performance ated are Loran VOR/DME), Glob rtial navigati l analyses of nts imposed by and effective	itations of in the cur- -C, Omega, al Posi- on system each naviga- and the ness of
A The purpose of thi a particular set of nav rent airspace environme VHF Omnidirectional Ran tioning System (GPS), D (INS). In addition to tion system, considerat deficiencies existing i navigation systems are	s study is to identify digation systems and nt. The navigation ge/Distance Measuring oppler navigation s detailed technical ion is also given to n the standards by measured.	tify the capabil d evaluate their n systems evaluating Equipment (system, and iner and operationation to the constrain which accuracy	lities and lim r performance ated are Loran VOR/DME), Glob rtial navigati l analyses of nts imposed by and effective	itations of in the cur- -C, Omega, al Posi- on system each naviga- and the ness of
A The purpose of thi a particular set of nav rent airspace environme VHF Omnidirectional Ran tioning System (GPS), I (INS). In addition to tion system, considerat deficiencies existing i navigation systems are	s study is to identify digation systems and nt. The navigation ge/Distance Measuring oppler navigation s detailed technical ion is also given to n the standards by measured.	tify the capabi d evaluate their n systems evaluating Equipment (system, and iner and operationation to the constrain which accuracy	lities and lim r performance ated are Loran VOR/DME), Glob rtial navigati l analyses of nts imposed by and effective	itations of in the cur- -C, Omega, al Posi- on system each naviga- and the ness of
A The purpose of thi a particular set of nav rent airspace environme VHF Omnidirectional Ran tioning System (GPS), I (INS). In addition to tion system, considerat deficiencies existing i	s study is to ident rigation systems and nt. The navigation ge/Distance Measuri oppler navigation s detailed technical ion is also given to n the standards by	tify the capabil d evaluate thei: n systems evalue ing Equipment (system, and ine: and operational to the constrain which accuracy	lities and lim r performance ated are Loran VOR/DME), Glob rtial navigati l analyses of nts imposed by and effective	itations of in the cur- -C, Omega, al Posi- on system each naviga- and the ness of
A The purpose of thi a particular set of nav rent airspace environme VHF Omnidirectional Ran tioning System (GPS), I (INS). In addition to	s study is to identify igation systems and nt. The navigation ge/Distance Measurit oppler navigation s detailed technical	tify the capabil d evaluate thei n systems evaluating Equipment (system, and ine and operational	lities and lim r performance ated are Loran VOR/DME), Glob rtial navigati l analyses of	itations of in the cur- -C, Omega, al Posi- on system each naviga-
A The purpose of thi a particular set of nav rent airspace environme VHF Omnidirectional Ran tioning System (GPS).	s study is to ident digation systems and nt. The navigation ge/Distance Measuri oppler navigation s	tify the capabi d evaluate thei n systems evalu ing Equipment (system, and ine	lities and lim r performance ated are Loran VOR/DME), Glob rtial navigati	itations of in the cur- -C, Omega, al Posi- on system
A The purpose of thi a particular set of nav rent airspace environme	s study is to identify is to identify is to identify is to identify and its state of the state o	tify the capabi d evaluate thei n systems evalu	lities and lim r performance ated are Loran	itations of in the cur- -C, Omega,
A The purpose of thi a particular set of nav	s study is to ident dgation systems and	tify the capabi d evaluate thei	lities and lim r performance	itations of in the cur-
A The purpose of thi	s study is to ident	tify the canabi	lities and lim	itations of
		•		
6. Abstract				
1				
· · · · · · · · · · · · · · · · · · ·				
S. Supplementery Notes	•	_		·····
Office of Systems Engin	eering Management	14	. Sponsoring Agency C	iode
Federal Aviation Admini	stration		-	
12. Sponsoring Agency Name and Address U.S. Department of Tran	sportation		Technical Rep	ort
		13	. Type of Ropert and F	Period Covered
Annapolis, Maryland 21	.401	''	DTFA01-80-C-1	.0030
ARINC Research Corporat	ion		Contract of Grant Ma	
9. Performing Organization Name and A	Idross	10	. Work Unit No. (TRAI	5)
L. Hogle, K. Markin, an	d S. Toth		1378-01-3-265	52 V
7. Author's)			Performing Organizati	en Report No.
EVALUATION OF VARIOUS 1	NAVIGATION SYSTEM C	ONCEPTS 6.	March 1982 Perferming Oreenizati	ion Code
4. Title and Subtitle		1 5 .	Report Date	
	AD-ALIZ	. 414	·····	

÷...

	1	.1.3.4	**	.	33		* •		9 # % 9	122		*		
Į]		11				11			, 1 		Ĭ	2	-22
nions from Motio	CENETH	53:	122	AIEA 0.4	2.2.2	(ASS (unidit)	1 222	VOLUME	8 - 7 - F	3,2	FRATURE (exect)		8 8 8	- 22 - 22
Appreziante Conve	Wan Ya Las	In ilonatoers Confirmatoers Confirmatoers		and the second second	seres mean square trianger hecters (3.00 e ²)	-	gene Litegrand Litegrand Litegrand Litegrand	ļ	milleiters Kiers Kiers	international designation of the second s		J. J.	¥ 4	
	ļ	8.	5	Ĩ	t î'r		•3-		1	-77		•		1
52		14 61 			₹ 14 18 10 10 10 10 10 10 10 10 10 10 10 10 10			• 70 • 70	• •	4 	•			נש ¹
. 	• 	l. la la	, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	" '''' ')	lului.	" ' '!'!' ' s		• • ••	' ' ' 	.ĺĺ.t.	 		1 	.uches ' ' '
•	ļ		55 e 5	•	<u>ት</u> ት	2	• 2 -		111		_77	:	·	. 186.
Messares			Contaments Contaments Anters Lidenters		Aparts Contaction Aparts Maters Aparts Maters Aparts Milandes				an (h-friger) An (h-friger) An (h-friger)		Liters Cubic meters Cubic meners	ļ		d tubles, see NBS Wax. P. 6.
nerions to Metric	a de la compañía de la	LENGTN	, 8 9	AREA	3 3 3 3	ASS (meight)	28 0.45 0.9	ADLUME	~ 5 \$	979 17-0 18-0	1.1 29.0 8'.0	ERATURE (exact)	19	ers.urs and nore derailed SD Calalog No. C13,10:29
Approximets Can	We to Lea		lstł		\$,1 ; }}!	=					Į.	ANU I		HERRICHAL, For other exact com- No. and Manustry, Price \$2,25, 5
	1	.1	1 a 7 i	•	ነኈኈ		3 #		::::	- 	1 27			-1 2,54 Unde ut New

METRIC CONVERSION FACTORS

ACKNOWLEDGMENT

The Federal Aviation Administration provided the overall guidance for this study. Material used in the preparation of this report was acquired through the cooperation of numerous Government agencies and private corporations. Appreciation is extended to the following organizations:

- American Airlines
- Bendix

E

an air a

E.

Ī

- Carlo

- Defense Mapping Agency
- Delco Electronics
- Eastern Airlines
- Honeywell
- Jeppesen Sanderson
- Litton Aero Products
- Magnavox
- Marconi Avionics
- Megatek
- MITRE Corporation
- Naval Ocean Systems Center
- Offshore Navigation, Inc.
- Pan American Airlines
- Rockwell/Collins
- State of Vermont
- Teledyne
- Transportation Systems Center of the Research and Special Programs Administration
- United States Coast Guard



SUMMARY

This report documents the work performed by ARINC Research under contract to the Federal Aviation Administration (FAA) to evaluate various navigation concepts. The purpose of this study is to compile and illuminate those technical and operational parameters that have the greatest impact on the compatibility of navigation systems operating in common airspace. The validity of analyses previously performed by recognized authorities in the field of navigation was assessed, and no need was found to conduct new analyses or to duplicate past analyses. The conclusions generated in this study are therefore based on existing reference material.

The navigation systems evaluated in this report, separately and in various combinations, are Loran-C, Omega, VHF Omnidirectional Range/Distance Measuring Equipment (VOR/DME), Global Positioning System (GPS), Doppler, and inertial navigation system (INS).

Although the primary focus of this study is on the technical and operational characteristics of navigation systems, economic and institutional issues are also discussed briefly. The technical performance capability of each navigation system is compared with existing accuracy requirements. Systems that are capable of satisfying the accuracy requirements are then evaluated to determine what limitations are imposed by signal coverage considerations. This analysis shows that, in terms of the currently used systems evaluated in this study -- Loran-C, Omega, VOR/DME, Doppler, and INS -- there is no unnecessary proliferation of navigation systems. This particular combination of navigation systems appears to satisfy current domestic, oceanic, and offshore navigation user requirements. No single system existing today as a fully operational system can meet all of these requirements.

Each system is also evaluated, separately and in various combinations, in an operational sense through the application of realistic flight procedure scenarios. These case studies identify a number of sources of conflict that could affect the integrity of position determination, pilot workload, Air Traffic Control (ATC) controller workload, and air safety. Although Loran-C, Omega, VOR/DME, and GPS navigation systems independently meet existing performance requirements as they apply to established VORreferenced airways, the errors associated with their use in an area

v

navigation application can combine in a way that causes lateral deviations about a geographic centerline exceeding current specifications for route width. The results of this study indicate that unrestricted use of area navigation systems in the current structure of the airspace is not advisable until some form of standardization is established in a number of areas, including the following:

- Path definition (e.g., great circle or rhumb line)
- Earth model (e.g., spherical, Clarke 1866, or WGS-72)
- Propagation models (i.e., sky waves and ground waves)

A REAL PROPERTY.

CONTENTS

L

I

I

and the second second second

an altern

ι

.

																							Page
ACKNOWLED	GMENT .	• • • • •	•	• •	•	•	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	iii
SUMMARY	• • • •		•	• •	•	•	• •	•	٠	•	•	•	•	•	•	•	•	•	•	•	•	•	v
CHAPTER C	NE: INT	RODUCTION	ι.		•	-	••	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	1-1
1.1	Backaro	und	•		•	•													•				1-1
1.2	Purpose					•			•					•	•			•	•				1-1
1.3	Scope		•			•				•			•	•		•			•		•		1-2
1.4	Technic	al Approa	ich			•		•	•		•		•				•	•		•			1~2
1.5	Report	Organizat	ion	•	•	•	• •	•	٠	•	•	•	•	•	•	•	•	•	•	•	•	•	1-2
CHAPTER 1	WO: NAV	IGATION S	Syst	EMS	OF	' II	NTE	RES	ST	•	•			•	•	•	•	•	•		•	•	2-1
2.1	Definit	ions	•		•	•		•	•	•		•	•	•	•	•	•	•	•		•	•	2-1
	211	Navigati	on																				2-1
	212	Dediones	ri aa	•••		•	•••	•	•	•	•	•	•	•	•	•	•	•	•	•	•		2-1
	2.1.2	Self-Cor	tya tai	ned	n. Na		· ·	101	•	•	•	•	•	•	•	•	•	•	•	•	•		2-2
	214	Area Nat	rias	+ i o			gu c	T.O.	•	•	•	•	•	•	•		•	•	•	•	•	•	2-2
	215	Hyperbol	ic	Nau	n. Nime	+ i d	•••	•	•	•	•	•	•			•			•		•		2-2
	2 1 6	Pho-Pho	Nav	ias	-30 +10	10 I. (•	•	•	•	•	•	•					•	•		•	2-4
	217	Pho-Pho-	Dho	Ne	via		· ·	•	•	•	•	•	•	•		•	•	•	•	•	•		2-5
	2.2.7	Navi gati	00	Aco			L 🗘 1 1	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	2-5
	2.1.0	Creat-Ci	.011 2011			e a a	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	2-6
	2.1.9		ne.		TRO	36	•	•	•	•	•	•	•	•	•	•	•	•	٠	•	•	•	2-6
	2.1.10	Geometri	e	4 1 1	+10	in d	 .#	Dre	•	•	on	: I	Gr	_E		٠	•	•	•	•	•		2-6
	2.1.12	Body-Ard			210 211 n		5 F			.04	.011	• •	92	ωE		•	•	•	•	•	•	•	2-7
	2.1.12	Locally	.0 0 T.0.1	-1	C00	ac. vali	o r Ine	1 au t a	Tre Tre			•	•	•	•	•	•	•	•	•	•	•	2-7
	2.1.14	Locally	Lev	el	Geo	gra	aph;	iC	Cc	OI	di	na	te	F	ra	me	•	•	•	•	:		2-7
2.2	System	Summaries	6 .	•••	•	•	• •	•	•	•	•	•	•	•	•	•	•	٠		•	•	•	2-7
	2.2.1	Loran-C			_			-					_					_			-		2-8
	2.2.2	Omega .	•		•		- •																2-8
	2.2.3	Differer	tia	io	med	a					-			-				-					2-8
	2.2.4	VOR/DMF					••			-		Ī	-					-		-			2-9
	2.2.5	GPS				•		-															2-10
	2.2.6	Doppler	•	•••	•	•		•		•	•	•			•	•	•	•		•	•	•	2-11

all and a state of the

ana an an anna a' ailteach an ailteach

CONTENTS (continued)

		Page	
		2.2.7 Conventional INS	
	2.3	System Features	
	·	2.3.1 Loran-C	
	2.4	Summary	
CHAP?	TER T	HREE: PERFORMANCE REQUIREMENTS OF NAVIGATION SYSTEMS 3-1	
	3.1 3.2 3.3 3.4	Overview	
		3.4.1International Standardization	
	3.5	Economic Considerations	
CHAPI	rer fo	DUR: TECHNICAL EVALUATION OF NAVIGATION SYSTEMS 4-1	
	4.1 4.2	Overview	
		4.2.1Transmitter Errors4-24.2.2Signal Detection Errors4-44.2.3Receiver Clock Errors4-54.2.4Signal Propagation Variations4-64.2.5Position Fix Calculation Errors4-74.2.6Total System Position Error4-7	
	4.3	Error Budget for Omega	
		4.3.1 Transmitter Timing Errors	

500

A State Street

viii

CONTENTS (continued)

E

		Page
	4.3.2	Signal Detection Errors
	4.3.3	Receiver Clock Errors
	4.3.4	Signal Propagation Variations
	4.3.5	Position Fix Calculation Errors
	4.3.6	Total System Position Error
4.4	Error 1	Sudget for VOR/DME
	4.4.1	Bearing Error Components
	4.4.2	Distance Measurement Errors 4-15
	4.4.3	Area Navigation Computation Error (E_C)
	4.4.4	Total System Error (Eg)
4.5	Error H	Budget for GPS
	4 5 1	
	4.5.1	Satellite Errors
	4.5.4	Signal Propagation Variations
	4.5.3	
	4.2.4	Total System Range Error
	4.5.5	Operational Considerations
4.6	Error B	Budget for Doppler
	4.6.1	Initialization Errors
	4.6.2	Calibration Error
	4.6.3	Doppler Signal Errors
	4.6.4	Instrumentation Errors
	4.6.5	Heading Reference Errors
	4.6.6	Total System Error
	Banan 5	A_{-25}
4./	Effor F	sudget for conventional ins
	4.7.1	Initialization Errors
	4.7.2	Gyro Errors
	4.7.3	Accelerometer Errors
	4.7.4	Sensor Mounting Orthogonality
	4.7.5	Gravity Anomalies
	4.7.6	Total System Error
4.8	Error E	Budget for Strapdown INS
	4.8.1	Gyro Errors
	4.8.2	Accelerometer Errors

ix

CONTENTS (continued)

															Page
	4.8.3	Sensor Mo	unting	Orthog	onal	lity		•		• •	•	•	•	•	4-31
	4.8.4	Gravity A	nomaly	• • •	•	• •	• •	٠	• •	• •	•	٠	٠	•	4-32
	4.8.5	Total Sys	tem Err	or	•	• •	• •	•	• •	• •	•	٠	٠	•	, 4-32
4.9	Summary	of Techni	c al Cap	abilit	ies	•	• •	•	• •	• •	•	•	•	•	4-32
CHAPTER F	IVE: OPI	ERATIONAL	evaluat	ION OF	NA	/IG/	TIO	n s	YST	ems	•	•	•	•	5-1
5.1	Interope	arability	Conside	ration		• •	• •	٠	• •	• •	•	•	•	•	5-1
	5.1.1	Coordinat	e Syste	ms	•	• •	• •	٠			•	•	•	•	5-1
	5.1.2	Route Str	uctures		•						•	•	•	•	5-2
	5.1.3	Separatic	n Stand	ards .	•			•		• •	•	•	•	•	5-6
	5.1.4	RF Enviro	nment .		•						•	•	•		5-6
	5.1.5	Equipment	Interf	ace .	•						•	•	•	•	5-6
	5.1.6	Flight Pr	ocedure	s.,	•	•	• •	•	• •	• •	•	٠	•	•	. 5 -7
5.2	Case Stu	udies		• • •	•	• •	• •	•	• •	• •	•	•	•	•	5-8
	5.2.1	Case Stud	y One .		•			•				•	•	•	5-8
	5.2.2	Case Stud	y Two .		•					• •	•		•		5-12
	5.2.3	Case Stud	y Three		•						•	•			5-21
	5.2.4	Case Stud	y Four		•				• •	• •	•	•	•	•	5-25
	5.2.5	Case Stud	y Five		•	•		•			•	•	•	•	5-27
5.3	Summary	of Case S	tudies		•	•	••	٠	•••	• •	•	•	•	•	5-31
CHAPTER S	IX: CON	CLUSIONS A	ND RECO	MMENDA	TIO	15	• •	•	•••	• •	•	•	•	•	6-1
APPENDIX A	A: DETAI	LLED SYSTE	M DESCR	IPTION	8			•		• •	•	•	•	•	A-1
ADDENTY		TONS .							• •				_		B-1
	ni allou	e a vite e e			• •	•	••	•	• •			•	•	-	
APPENDIX	C: COMPU	JTATIONS		• • •	•	• •	••	•	• •	• •	•	•	٠	•	C-1
APPENDIX I	D: REFEI	RENCES AND	BIBLIO	GRAPHY	•			•			•			•	D-1

CHAPTER ONE

INTRODUCTION

1.1 BACKGROUND

The continuing growth of aviation traffic, in both general aviation and commercial airlines, places increasing demands on the national navigation system. Higher densities of commercial and general aviation aircraft will demand more accurate positioning en route than is now required. To ensure a satisfactory integrated national navigation system capable of meeting anticipated requirements, the Federal Aviation Administration (FAA) is conducting strategic studies and planning with regard to possible navigation concept scenarios to be adopted between now and 1995. The strategic studies of the FAA require an accurate, detailed assessment of these concepts and of the capabilities, costs, and interdependencies of the possible navigation systems that may be applied to meet anticipated navigation requirements.

Consequently, the FAA contracted ARINC Research to study critical aspects of current and future navigation systems and practices. This report presents the current status of work involving the evaluation of various navigation system concepts in the context of the existing airspace environment.

1.2 PURPOSE

The Federal Radionavigation Plan (FRP) directs the Departments of Transportation and Defense to select a suitable mix of radionavigation systems that can meet the diverse technical, operational, and economic requirements imposed by users, manufacturers, and the Air Traffic Control system. The purpose of this study is to compile and illuminate those technical and operational parameters that have the greatest impact on the compatibility of navigation systems operating in common airspace. The validity of analyses previously performed by recognized authorities in the field of navigation was assessed, and no need was found to conduct new analyses or to duplicate past analyses. The conclusions generated in this study are therefore based on existing reference material.

1.3 SCOPE

This study reviews the technical and operational characteristics of Loran-C, Omega, VHF Omnidirectional Range/Distance Measuring Equipment (VOR/DME, Global Positioning System (GPS), Doppler, and inertial navigation system (INS). The systems are evaluated both independently and in various combinations. The evaluations are in the form of comparative analyses -the systems are compared with each other as well as with existing FAA technical and operational requirements. Conclusions are presented concerning the effectiveness of particular navigation systems in satisfying the stated requirements, together with recommendations for improving operational performance. Inadequacies in formulating accuracy are pointed out. Finally, the study recommends a combination of navigation systems that will provide the greatest potential for acceptance as the primary means for navigation in the context of the current airspace environment.

1.4 TECHNICAL APPROACH

Technical and operational capabilities of each navigation system under consideration were identified through literature searches and discussions with manufacturers and users. Performance requirements as specified by the FAA were obtained from the recently published FRP and from previously issued FAA advisory circulars. Technical capabilities of each system were then compared with the requirements. The results of these comparisons were used to categorize each system with respect to which flight phases could be accommodated in a purely technical sense, using both statistical and measured navigation system capabilities. The systems were then evaluated to determine their operational capability in specific case studies. The scenarios selected for the case studies were representative of typical situations that are encountered frequently. The operational capability of each navigation system in each scenario was defined on the basis of selected measures of performance. The degree of system interoperability was determined and discussed in terms of potential conflict in each scenario as a function of the relative differences in operational capability between system types. It was possible to interpret the capabilities and limitations of various system combinations by integrating the results of each case study with the results of the technical evaluation. This assessment led to recommendations of which system mixes should be supported by the FAA and also provided insight into the factors that tend to limit full exploitation of currently available navigation system capability. This report suggests steps that can be taken to overcome these limitations.

1.5 REPORT ORGANIZATION

Chapter Two defines terms used throughout the report and briefly describes the navigation systems discussed, together with a summary of the operational features of the airborne equipment associated with each of those systems. Chapter Three details the performance requirements specified by the FAA that any navigation system must satisfy to be acceptable.

Chapter Four contains detailed analyses of the technical capabilities of each navigation system and compares these capabilities with the requirements.

Chapter Five provides insight into the operational capabilities of each navigation system, both independently and in combination with other systems, through specific case studies.

Chapter Six presents conclusions and recommendations.

Appendix A provides detailed descriptions of each navigation system.

Appendix B contains the equations used in computing some of the navigation errors specified in the report, and Appendix C illustrates the application of these equations.

Appendix D lists the references cited in this report, together with other source material.

CHAPTER TWO

NAVIGATION SYSTEMS OF INTEREST

2.1 DEFINITIONS

L

The following sections define navigation terms that are used throughout this report. The document, American Practical Navigator (Reference 1), was used as the preferred source of definitions. Modifications were sometimes necessary, however, to reflect the specific context in which navigation terms are applied in this report.

2.1.1 Navigation

Navigation is a means by which an aircraft is given guidance to travel from one known position to another known position. This process involves referencing the actual aircraft position to a desired course. The desired course is a path, between the two known locations, defined by a course generation method (e.g., great-circle, rhumb-line).

2.1.2 Radionavigation

Navigation systems characterized by their use of radio waves generated externally from the aircraft to determine position are known as radionavigation systems. They include Loran-C, Omega, VOR/DME, Tactical Air Navigation (TACAN), and GPS navigation systems; these will be discussed in detail in this report. Other navigation systems that utilize the radio wave spectrum but are not included in this report are Loran-A, instrument landing system (ILS), microwave landing system (MLS), the Navy Navigation Satellite System (TRANSIT), radio beacons, Decca, and radar.

The Loran-A navigation system has been replaced by Loran-C; ILS and HLS are landing systems; and TRANSIT navigation does not provide continuous coverage and is therefore not suitable as a primary means of navigation. Although radio beacons may be an element of a navigation system mix, their lack of sufficient navigational accuracy precluder their use as a primary system. The Decca navigation system has coverage limitations that make it less suitable for general use than Loran-C, and radar is primarily used for ground surveillance, not navigation. Radar vectors are issued by air traffic controllers and accepted by pilots as a means of navigation. However, because they do not constitute a primary form of navigation, they are not discussed as such in this report.

2-1

A number of navigation systems are derivatives of a primary system. For example, basic VOR has led to the development of wide-aperture (digital) VOR, Doppler VOR, and precision VOR. The study of system derivatives is outside the scope of this report.

2.1.3 Self-Contained Navigation

Navigation systems that do not require the use of externally generated radionavigation signals on a continuous basis for determining position are known as self-contained navigation systems. They include Doppler navigation systems and inertial navigation systems (INSs); these systems will be discussed in detail in this report.

The accuracy of self-contained navigation systems is specified differently than for the radionavigation systems discussed in this report. Whereas the accuracy of the radionavigation systems is defined as a position-determination discrepancy, the accuracy of self-contained navigation systems is defined as a rate of position-accuracy degradation, which is a function of elapsed operational time for INS accuracy and a function of distance traveled from the flight origin for Doppler accuracy. INS accuracy is specified in nautical miles per hour (nm/hr), and Doppler accuracy is specified as percentage of distance traveled from flight origin.

2.1.4 Area Navigation

Area navigation is an application of the navigation process providing the capability to establish and maintain a flight path on any arbitrarily chosen course that remains within the coverage area of the type of navigation signals being used. This random navigation capability is generally referred to as RNAV. Loran-C, Omega, GPS, Doppler, and INS are inherently RNAV navigation systems. VOR/DME in its basic form is station-referenced and therefore is not an RNAV system. However, more sophisticated airborne VOR/DME processors provide RNAV capability.

2.1.5 Hyperbolic Navigation

Hyperbolic radionavigation systems determine position through difference measurements of signals from three transmitting stations. These measurements can be either time differences (the elapsed time between the arrival of signals from two stations) or phase differences measured between two signals.

A transmitter that emanates signals in all directions creates a circular wavefront, with the transmitter at the center of the circle. A series of circles is generated, each at a successively larger distance (wavelength- λ) from the transmitter. If signals are transmitted from two stations at the same time, they will meet along a line equidistant from the two stations. This line, called the centerline, is the perpendicular bisector of a line drawn between the two stations, called the baseline. Figure 2-1 illustrates this pattern.



Source: American Practical Navigator, Reference 1.

Γ.

ł,

Figure 2-1. HYPERBOLIC NAVIGATION GEOMETRY

2-3

Reception of the two signals from a point not on the centerline will result in a measured difference in time or phase. The locus of points at which the same difference can be measured defines a hyperbola. The time or phase difference measured corresponds to a distance difference. Each hyperbola in Figure 2-1 is a line of position (LOP), representing a constant range difference from the two transmitters. When a user is along the line connecting the two stations but is not between those two stations, the user is on a baseline extension. The farther a user is from the baseline along a given LOP, the larger the spacing, or gradient, between consecutive LOPs per unit measurement difference. The width of this spacing is referred to as a lane. In regions of a large gradient, a relatively small time- or phase-difference error will cause a relatively large position error. The gradient is so large along the baseline extensions that navigation accuracy is severely degraded in those areas.

Although a user can determine which LOP corresponds to his difference measurement, he is unable to locate his point of position along the LOP. A second LOP can be defined through the use of a third station, with a second baseline being drawn between the third station and either of the first two stations. The intersection of the two LOPs defines a position fix, illustrated in Figure 2-2.



Figure 2-2. HYPERBOLIC POSITION FIX

2.1.6 Rho-Rho Navigation

In hyperbolic navigation, time or phase differences are measured to determine aircraft position in relation to the distance between stations. The distance between the aircraft and a ground station is commonly referred to as range, or rho. Rho-rho navigation involves directly measuring the total time it takes for each of two signals to travel from separate transmitters

2-4

to the airborne receiver. Each time measurement corresponds directly to a range measurement and defines a circular LOP. This form of operation is referred to as the ranging mode. The intersection of two circular LOPs provides two possible position fixes; the ambiguity is easily resolved by knowledge of approximate position.

The gradient between LOPs in the hyperbolic mode increases in accordance with the divergence of the hyperbolas. In the ranging mode, however, the gradient between the circular LOPs is constant, as shown in Figure 2-1. Therefore, navigation at extended ranges from the baseline is possible with use of the ranging modes; the coverage area of the stations is thereby effectively increased. In addition, whereas hyperbolic navigation requires three stations, rho-rho navigation requires only two stations. Greater freedom in selecting stations allows for consideration of station-to-user geometry to improve accuracy. The fact that the user need be within range of only two stations rather than three is another factor in providing extended coverage beyond that possible for hyperbolic navigation.

A disadvantage of rho-rho navigation is the user requirement for a highly stable, and therefore costly, time source (local oscillator or clock) against which the time measurements are made. Any clock drift will directly affect the range measurement. Clock bias is not a factor in the hyperbolic mode, since time differences between signals are measured.

2.1.7 Rho-Rho-Rho Navigation

I

-

1 .

U

When a third station is used to complement the two range measurements of the rho-rho mode, three circular LOPs are generated. The three LOPs will intersect at a common point only if there are no clock errors. If the range measurements are adjusted until common intersection is achieved, clock drift can be estimated and applied to subsequent range measurements. This technique allows the use of a less expensive clock, but necessitates the use of three stations.

2.1.8 Navigation Accuracy

Navigation accuracy is usually expressed in terms of a probability that deviation will be less than some stated magnitude of error. Different terms are used to define error, depending on the application. The terms "standard deviation" and "root mean square" are equivalent when they refer to one-dimensional zero-mean errors. Navigation errors typically have components in at least two dimensions -- along-track and cross-track. Terms commonly used in referring to two-dimensional navigation accuracy include circular error probability (CEP), standard deviation, and dimensioned root-mean-square error or radial error (drms).

CEP defines the radius of a circle for which there is a 50-percent probability that all position measurements will be inside. Standard deviation, or sigma (σ), in a two-dimensional analysis involves a radial rather than a linear distribution of errors. Numerically, 1 σ corresponds to 39.95 percent ($2\sigma = 86.47$ percent) of a circular distribution, whereas it corresponds to 58.27 percent (20 = 95.45 percent) of a linear distribution. Reference to a circular standard deviation is meaningful only when the error figure, whose orthogonal axes are defined by one-dimensional errors (referred to as 0x and 0y), is a circle, a result that occurs only when 0x = 0y. Since 0x and 0y are generally not equal, reference to a position accuracy of 20 can cause confusion as to whether the number describes the 86-percent two-dimensional probability circle or the 95-percent onedimensional probabilities for each error axis.

The drms concept is introduced to give a single measure that will account for variations in both σ_X and σ_Y that generate an error ellipse rather than a circle. One drms is defined as the radius of a probability circle, with the magnitude of the radius the RSS of the one-dimensional 10 error components along the major and minor axes (σ_X and σ_Y) of the error ellipse. The major drawback of this measure is that the value of probability associated with a fixed value of drms varies with the eccentricity of the error ellipse. Use of 2 σ error components results in the computation of a 2-drms value. The 2-drms variation in probability is small (0.954 to 0.982).

Navigation accuracies or error components presented in this report are 20 (95.45 percent) values unless specified otherwise. Relative accuracy is defined as the accuracy with which a user can measure relative position with respect to another user of the same navigation system at the same time. Repeatable accuracy is the accuracy with which a user can return to a position whose coordinates have been measured previously with the same navigation system.

2.1.9 Great-Circle Course

The shortest path between two points on a spherical earth is a great circle, which is created by the intersection of the earth's surface and a plane defined by the center of the earth and the points of origin and destination.

2.1.10 Rhumb-Line Course

A rhumb-line path connects origin and destination along a path that maintains constant true course (i.e., the course crosses successive lines of longitude at a constant angle).

2.1.11 Geometric Dilution of Precision (GDOP)

The effect that geometry between the user and the signal transmitters has on accuracy is expressed in terms of geometric dilution of precision (GDOP). A GDOP of 1 yields the value of accuracy associated with the best geometry configurations. Progressively larger values of GDOP indicate worsening geometry and correspondingly degrade the achievable accuracy of the system.

For hyperbolic navigation, geometrical considerations include the crossing angle of intersecting LOPs and the spacing between consecutive

LOPs. The most favorable geometry occurs along baselines with the two intersecting LOPs being nearly orthogonal. As the crossing angle becomes more shallow, it becomes more difficult for the system to resolve the point of intersection accurately. Uncertainty in time or phase measurement translates into a position error, the magnitude of which is a function of the distance between LOPs (the lane width). A certain percentage of uncertainty in determining position within a lane will result in larger errors for wider lanes. The lane width widens between hyperbolic LOPs with increasing distance from the baseline, resulting in degradation of accuracy corresponding to typical GDOPs of 2 to 3.

For rho-rho or rho-rho-rho navigation, the only geometrical consideration is the crossing angle of intersecting LOPs. Since the spacing between the consecutive circular LOPs is constant when ranging techniques are used, accuracy is not degraded by varying gradient between LOPs. The most favorable crossing angle geometry occurs when the two intersecting LOPs are nearly orthogonal. Since the use of ranging techniques allows freedom to select those stations that optimize the crossing angle of the LOPs, the corresponding GDOP can be reduced nearly to 1 under most conditions.

2.1.12 Body-Axis Coordinate Frame

The body-axis coordinate frame is a right-handed system that is fixed relative to the frame of the aircraft, with its origin at the aircraft's center of gravity. The X axis points outward through the nose of the aircraft. The Y and Z axes point respectively out the right wing and out the bottom of the aircraft.

2.1.13 Locally Level Coordinate Frame

Pitch, roll, and yaw are measured with respect to a locally level coordinate frame. This right-handed coordinate system has its origin at the aircraft's center of gravity. The yaw axis is coincident with the local vertical and points downward. The roll axis points in the direction of aircraft heading, and the pitch axis points 90 degrees to the right of aircraft heading.

2.1.14 Locally Level Geographic Coordinate Frame

This coordinate frame is also locally level but is referenced to geographic north. The axes point in the east, north, and "up" directions, regardless of the aircraft attitude or heading.

2.2 SYSTEM SUMMARIES

State of the second second

The following sections briefly summarize the systems evaluated in this report. Detailed descriptions are provided in Appendix A.

的人们也能够认为我,我们这些事实我们的没有的问题,我不能没有的人的。 第二章

2.2.1 Loran-C

Loran-C is a hyperbolic radionavigation system that uses ground waves at low frequencies to obtain an operating range of 600 to 1,500 nautical miles (nm) independent of line-of-sight. It also uses pulse techniques to avoid skywave contamination. Accuracy of Loran-C is heavily dependent upon GDOP factors at the user's location within the coverage area. Loran-C is capable of achieving absolute 2-drms accuracies of 463 meters (0.25 nm) or better. The repeatable and relative accuracies of Loran-C are usually between 18 and 90 meters (0.01 and 0.05 nm). The Loran-C system currently consists of 16 chains operating throughout the world, comprising a total of 51 transmitting stations. Two-thirds of the conterminous United States and Alaska is currently within the Loran-C coverage area; there is no Loran-C coverage in the sourthern hemisphere. Loran-C coverage is concentrated in the coastal areas of the Atlantic and Pacific oceans and the Norwegian and Mediterranean seas.

2.2.2 Omega

Omega is another hyperbolic navigation system, but it utilizes sky waves rather than ground waves to give each transmitter an operating range of about 5,000 nm. The accuracy of the Omega system is limited by the accuracy of the propagation corrections that are applied to the received signal; a predictable accuracy of two to four nm is the design goal of the system. Statistical studies conducted in the North Atlantic show that rms positional accuracies of one to two nm are possible.

Omega signals transmitted from eight stations provide nearly worldwide coverage. There are also a number of United States Navy very low frequency (VLF) communication transmitters located around the world that can be used as supplementary signal sources for Omega navigation. Use of VLF communication stations improves the dependability and accuracy of continuous coverage during Omega/VLF navigation. A wider choice of candidate stations provides greater probability that GDOP can be minimized by judicious station selection. The extent of infusion into the marketplace of the Omega/ VLF concept is making Omega/VLF the dominant form of civil Omega navigation.

Another recent development in the evaluation of Omega navigation has been the increasing use of ranging techniques rather than the hyperbolic technique in the airborne processor. The ranging mode requires reception from only two stations (rather than the three required in the hyperbolic mode) to obtain a position fix and provides capability for increased coverage and accuracy. However, it is necessary to employ a highly accurate, and therefore expensive, clock for rho-rho measurement. As explained previously, using a third station in a rho-rho-rho mode permits use of a less expensive clock.

2.2.3 Differential Omega

Differential Omega is a concept that supplements the standard Omega signal received by the airborne unit with differential correction terms

that compensate for localized variations in signal propagation characteristics. A more complete description of the differential umega concept is contained in Section 2.8 of Appendix A. Use of differential Gmega techniques has been demonstrated in marine applications to provide a 20 accuracy capability of 0.2 to 0.4 nm, but only when within 50nm of the ground monitor station that provides the differential corrections. The level of accuracy diminishes with increasing distance from the monitor station.

Most data currently available on the performance capability of differential Omega are associated with marine usage in coastal regions. The testing of differential Omega in airborne applications has not been sufficient to provide conclusive evidence of expected performance. The lack of data in this regard precludes evaluation of differential Omega in this report as a currently existing means of airborne navigation. From the data that do exist, however, there are indications that differential Omega could be suitable as a means of providing navigational capability where other navigation systems are not suitable, thereby becoming an element of a mix of systems.

2.2.4 VOR/DME

The international standard en-route navigation system used within the conterminous United States is VOR/DME. VHF Omnidirectional Range provides the azimuth relative to the VOR ground station, and Distance Measuring Equipment (DME) furnishes a measurement of distance from the aircraft to the DME ground station. VOR and DME are usually collocated as a VOR/DME facility. TACAN is a combination of omnibearing and distance-measuring functions. TACAN is a military system, with the azimuth portion of a TACAN facility not widely used by nonmilitary users. The distance-measuring function of TACAN, however, is the international standard DME and is therefore accessible to anyone with a DME interrogator. So that both military and civil aircraft can navigate using the same airway network, TACAN facilities are collocated with VOR facilities. A VOR collcated with a TACAN is called a VORTAC. Since there is no difference in operation or performance between a VORTAC and a collocated VOR/DME, VORTACs will be included in the VOR/DME system classification in this report.

VOR and DME navigation aids (navaids) can be used in any of the following three configurations:

- VOR only
- VOR/DME
- DME/DME

The VOR system is the basis for defining airways and is therefore an integral part of air traffic control procedures. Use of VOR in the absence of DME results in navigation based solely on directional information. This singular capability is the basis upon which Victor airways were established --VOR-to-VOR navigation. Two VOR stations can be used to define a radial intersection reporting point. By charting the bearing information obtained

LISS CALL

from the two VOR stations, a position fix can be determined. The combination of VOR and DME at a single site provides the capability for position fixing by means of a single facility. This configuration forms the standard International Civil Aviation Organization (ICAO) short-range navigation system. The use of dual DME offers a significant improvement in position-determination accuracy in areas that have suitable dual-DME signal coverage.

The accuracy of VOR is the basis of the design specification for United States air traffic control standards and procedures. The magnitudes of both accuracy and signal coverage of VOR and DME are a function of aircraft altitude and distance from the station. Line-of-sight limitations restrict coverage to 30 nm or less at ground level, a distance that progressively increases with altitude to an upper limit approaching 200 nm.

VCR signal coverage is a function of the class of VOR station: terminal, low altitude, or high altitude. A terminal VOR provides coverage for an altitude range of 1,000 to 12,000 feet at radial distances out to 25 nm. A low-altitude VOR provides coverage from 1,000 to 18,000 feet at radial distances out to 40 nm. The coverage provided by a high-altitude VOR is a function of aircraft altitude, as shown below:

overage (of radial distances of:	is provided at altitudes of:
	40 nm	1,000 feet to 14,500 feet
	100 nm	14,500 feet to 18,000 feet
1.111	130 nm	18,000 feet to 45,000 feet
	100 mm	45,000 feet to 60,000 feet

These operational coverage areas can be expanded under special circumstances by extending the noninterfering coverage radius to no more than 110 nm at altitudes below 18,000 feet, or 185 nm above 18,000 feet.

2.2.5 GPS

The NAVSTAR Global Positioning System (GPS) is a proposed space-based radionavigation system that is intended to provide accurate navigation and position information to all properly equipped users. The fully operational system will enable continuous worldwide navigation, regardless of weather conditions. Current concepts are based on an 18-satellite constellation -- a reduction from the 24 in the original specification. Using signals from four satellites, a user can obtain three position dimensions (latitude, longitude, and altitude), determine time, and derive velocity.

Although GPS is a military system, its potential use for civil navigation is a major topic of discussion and study. Current plans call for exclusive military use of the precision code (P-code), which, in the context of an 18-satellite constellation, enables predictable positioning accuracy of 25 meters (0.013 nm) horizontally and 30 meters vertically (95 percent probability), depending on the capability of user equipment and the user-to-satellite geometries. The navigational accuracy to be made available by the military to civilian users of the coarse acquisition

2-10

(C/A code is uncertain; speculation ranges from 50 to 500 meters (0.016 to 0.268 nm).

Three-dimensional navigation coverage requires four GPS satellites to be in view of the user. The accuracy obtained depends on the geometry involved. A substantial increase in GDOP is predicted because of the reduction in the specified number of satellites from 24 to 18. The effect of increased GDOP on civilian use of GPS is somewhat dependent on which areas of the world are most adversely affected.

2.2.6 Doppler

A Doppler navigation system provides self-contained navigation capability for use in over-water and remote environments, by using inputs from a Doppler radar and a heading reference. The Doppler radar measures groundspeed, and the heading reference determines the aircraft's heading. These inputs are then used by the Doppler navigation system to continuously compute and display current aircraft position relative to the flight path origin.

The Doppler radar transmitter directs radar signals toward the ground; the signals are scattered by the ground surface and are then detected by the Doppler radar receiver. The frequency is shifted in the scattered radar signals by an amount related to aircraft velocity. The Doppler radar then determines the groundspeed of the aircraft by measuring this frequency shift in the detected signals. The heading reference typically used in a civil Doppler navigation system measures aircraft heading with a magnetic compass.

The cross-track accuracy of Doppler navigation is highly dependent on the heading reference accuracy. Doppler navigation accuracy degrades as a function of distance traveled from the flight path origin. Cross-track accuracy for current civil Doppler navigation systems is typically three percent of flight path distance, and along-track accuracy is one percent.

An operational problem that occurs over calm water is an occasional loss of signal due to insufficient scattering of the radar signals in the direction of the aircraft.

2.2.7 Conventional INS

e - ..

Conventional INS also provides self-contained navigation capability, using measured aircraft acceleration to determine aircraft position. For computational convenience, the accelerometers used to measure aircraft acceleration are mounted on a gimbaled platform mechanized to maintain continuous alignment with the locally level geographic coordinate frame, regardless of aircraft orientation. Current position is determined by integrating the measured aircraft acceleration to obtain velocity and integrating a second time to obtain position displacement. Gyros are used to measure the deviations of the platform orientation. These measurements are then applied to the platform to correct its orientation. The accuracy of conventional INSs is highly dependent on the characteristics of these gyros. Since the error associated with the gyro measurements tends to

2-11

and the state of the

increase as a function of time elapsed during operation, the position determination of conventional INSs used in civil applications has typical rates of accuracy degradation of one to two nm per hour of use; conventional INSs being developed for use in military applications have demonstrated rates of accuracy degradation as low as 0.08 nm per hour, using electrostatically suspended gyros.

2.2.8 Strapdown INS

Strapdown INS will also provide self-contained navigation capability when commercially available. It is similar in concept to conventional INS; it differs in the manner in which the gyros and accelerometers are mounted. In strapdown INS, the accelerometers and gyros are mounted in fixed reference to the aircraft frame, and as a result measure acceleration and angular components in body-axis coordinates. These components must be transformed into the local coordinate frame before velocity and position are computed. Angular data provided by gyros are used in the transformation. Several types of gyros have been considered for this application, including ring laser gyros, tuned rotor gyros, and electrostatically suspended gyros, which are each described in Section 8.3 of Appendix A.

Strapdown INS has fewer moving parts and is therefore more reliable than conventional INS. Because of the coordinate transformations, however, strapdown INS has more computational requirements. These computations have recently become relatively easy to perform with the development of minicomputer technology.

The strapdown INS now being developed for use in civil applications combines a strapdown inertial reference system (IRS) using ring laser gyros, and a flight management computer (FMC). The strapdown IRS provides the FMC with current aircraft position. The FMC performs waypoint storage, flight path generation, and steering command computations. The term "strapdown INS" as used subsequently in this report refers to this IRS/FMC combination.

Accuracy of current civil strapdown INSs is primarily dependent on the characteristics of the ring laser gyros used. Currently achieved rates of position accuracy degradation are typically one to two nm per hour of use for these systems. Military strapdown INSs being developed have demonstrated rates of accuracy degradation as low as 1.1 nm per hour.

2.3 SYSTEM FEATURES

Airborne navigation units have evolved and continue to evolve in accordance with the needs of the various user communities. Consequently, the man-machine interfaces of airborne equipment differ from each other as manufacturers provide special features desired by particular users.

2.3.1 Loran-C

Use of Loran-C for airborne navigation evolved as a supplement to conventional VOR/DME in areas of poor VOR/DME signal coverage. The inability of VOR/DME to provide offshore signal coverage made Loran-C particularly appropriate for offshore helicopter operations. The Coast Guard uses Loran-C for over-water search and rescue missions, and offshore oil well operators use it for navigating between land and offshore drilling platforms. Since these users require the capability of precise position location, manufacturers have emphasized special features such as stored programmable search patterns and highly accurate ground position and track determination. Until Loran-C is more widely accepted for use along conventional domestic airways, there will be little incentive for manufacturers to include a data base of VOR/DME navaids to enable the construction of navaid-oriented area navigation routes.

2.3.2 Omega

Airborne Omega navigation is used primarily to serve the needs of international air carriers. Neither VOR/DME nor Loran-C can provide coverage over mid-ocean. Transoceanic navigation is typically performed by either inertial navigation systems (INSs) or Omega. The inertial systems, however, are limited in accuracy by the accumulation of gyrodrift errors. Omega is frequently used, therefore, to provide periodic updates to the INS and to serve as a back-up navigation system. However, Omega is also certified for use as a primary means of oceanic navigation. The ability of Omega to serve as a worldwide navigation system has prompted some manufacturers to provide capabilities that make it operationally efficient in almost any environment. Omega navigation systems may (1) include a data base of airports and navaids for use in flight planning (2) provide the capability for an extensive waypoint list that is useful for long transoceanic or transcontinental flights, and (3) allow selection of trackhold autopilot coupling.

2.3.3 VOR/DME

VOR/DME was developed in response to the basic need for an extensive network of omnidirectional navigation aids. As a result, the VOR/DME navigation system provided the means of successful navigation within the United States.

Some airborne VOR/DME navigation units can provide RNAV capability, but others cannot. A VOR-only unit is limited to a non-RNAV application, but single and multiple VOR/DME units are available for either non-RNAV or RNAV applications. The RNAV capability adds to the cost.

A multiple VOR/DME RNAV unit that utilizes two VOR/DME navaids (primary and secondary) offers increased automation as well as a substantial improvement in accuracy as a result of dual-DME processing capability. Since RNAV systems can be geographically oriented (e.g., referenced to latitude and longitude), VOR/DME RNAV units can generally maintain position in terms of latitude and longitude displacement in relation to the coordinates of the reference VORTAC. To provide this capability, a navaid data base can typically be provided, containing the latitude and longitude coordinates of all relevant VORTAC navaids. The navaid information is entered into the data base through either preprogrammed memory elements provided by the manufacturer or manual input by the pilot. Access to this data base enables the use of automatic station-selection algorithms, which eliminate the need for both pilot selection of appropriate navaids and input of corresponding navaid frequencies.

2.3.4 GPS

The satellite-based GPS is still in the development stage. Development of GPS for civil aviation applications is motivated by a desire to replace currently existing navigation systems with a single system that can meet the requirements of all users throughout the world. Although no production units are commercially available yet, studies have been performed, based primarily on cost considerations, on the likely configurations and capabilities of airborne GPS receivers. The need to offset the initial high cost of new and complex technology will result in minimizing optional capabilities. Consequently, the "low cost" GPS receiver is projected to have functional capabilities comparable to the less expensive RNAV VOR/DME systems.

2.3.5 Doppler

Doppler navigation systems were developed in response to the need for navigation capability in areas with inadequate or nonexistent VOR/DME coverage. They were first used in transoceanic navigation before airborne INS or Omega units were available. These early Doppler navigation systems do not use great-circle course generation and do not compute current latitude and longitude. Doppler navigation systems now being developed will include more convenient man-machine interfaces.

2.3.6 Conventional INS

Conventional INS was developed to provide an accurate and convenient transoceanic navigation capability. INS has the additional capability of providing circraft attitude information for use by the autopilot and other aircraft instruments. Because of accumulated error in an INS due to gyro drift, an independent INS cannot currently provide the level of accuracy required for many over-land applications. However, by providing the INS with periodic updates from a VOR/DME or other radionavigation system, INS can be used for over-land navigation. This feature has prompted manufacturers to include a data base of airports and navaids and an extensive waypoint list capability in recent INSs.

2.3.7 Strapdown INS

Strapdown INS was developed in response to the high maintenance cost of conventional INS. Strapdown INS is capable of providing the same capabilities as conventional INS with the same level of accuracy, but with lower maintenance costs. The current configuration of strapdown INS is an inertial reference system (IRS) used in conjunction with a flight management system (FMS) to provide navigation capability.

2.4 SUMMARY

E

Table 2-1 summarizes the functions and features generally available in typical 1980-model airborne navigation units. The table should not be interpreted to suggest that a particular navigation unit always provides the features indicated or that features not included could not be accommodated. The differences between the man-machine interfaces of various RNAV units primarily reflect marketing decisions on the part of the manufacturers. There is no technical impediment to providing all navigation units with the same level of capability in terms of the man-machine interface. In fact, as the market potential for the various types of navigation system units becomes more obvious, units are being offered with more features in common.

în.	DIA 2-1. BURNARY OF	TYPICAL P	NEVINE OF A	VAILABLE AI		GATION UNITE		
	Non-INAV				MINY			
Unit Penture	VOR/DHE	VOR/DHE	Loren-C	Omge	ore	Deppler	Conventional INS	Strepton SHS
·			Enpusa					
Plight Planning	1	1		1	T			
By latitude/longitude	•	•	l x	 x -	×	x	x ;	x
By place/bearing/	•	{ x	x	×		×	•	•
alsonne waypelass By nevera alpha	•	×	×	x	•	•	•	•
By designators By desired magnetic	TO VORTAC ONLY	× I	x	x	•	×	•	•
heading By in-Slight updating	x	×	x	*	×	×	×	x
Flight Information						j		
Current position Time (GNT)				X	X		X	ž
Date Memotic veriation	•	•	•	×) ×) Ó	ĕ	ě
Nenuel Automatic prestored	•		X	X	x	:	1	•
Actual heading							, i i i i i i i i i i i i i i i i i i i	-
Nanuel Automatic sensor	:	*	X	X		A A A A A A A A A A A A A A A A A A A		X
True airspeed				1	{			
Nenuel Automatic consor		•	× •	X			X X	×
Station Selection and Deselection								
Automatic Monual	e X	X X	X	X	x X	N/A N/A	1/A 1/A	11/A 11/A
		¢¢	erating Hod	÷4				
Navigation	I			1	Γ			
Direct to			1	1	j			
Great circle Shumb line	× ●	X	ĕ	×	×	X X	× •	×
Log change								
Automatic	i i) x	x i	x) x	X	X	X
Parallel offset Dead reckening	•	×	×) ×	•	×	x	•
With sensor updates	•	•	•	x	•	•	٠	٠
(e.g., TAS, HDG) Without sensor updates	•	•	×	×	•	•	•	٠
Autopilot Coupling]				
Track hold Turn anticipation	Along radial	X	:	×××	x	*	×	x
		Diej	alay Paramet	ers	1	-		
Current Position	[T						
Latitude/longitude Bearing/distance to or from stored	:		x	X	X X	X	X	×
Waypoint Bearing/distance from	x	x		•		N/A	N/A	N/A
signal source Time to waypoint	•	x	×	x	x	•	X	•
Current Meading						i		
lingne tic True				X		:	X X	X X
nurse audre	L	1. •	•	X	•	X	X	X

(continued)

Table 2-1. (continued) Hon-RKAV **BHAV** Unit Peatures Conventional Strapdown ZNS VOR/DHE VOR/DRB Queya Leran-C GPE Doppler ZHE Display Parameters (continued) Current Track Desired cogle Actual angle To/From VORTAC only To/From VORTAC only XX X X X ð X X x X X X X Angle error Gross-track deviation To/From VORTAC only XX X XXX XXX ÷ x XXX X XX X X X X Groundspeed X × Atmospheric Conditions Wind speed : : XX X X : : XX X X Wind direction Marnings and Indications Operator error • • X X x x X Malfunction Approaching or missed XX XX XXX x X XX X waypoint Input and Output (I/O) Interfaces Directional Gyro • x ٠ X • X • . Course Deviation ٠ • x x x X x x Indicator • Autopilot X x ٠ x ٠ x x • ٠ Data Bus, Data Link X X × x x . Flight Management ۲ x x x x . . x Computer Mechanical Characteristics Receiver/Processor Unit VOR DHE (RPU) Weight (pounds) Volume (cubic feet) 11.0 11.0 17.0 \$7.0 44.0 3 5 26.0 12.0 õ.1 0.1 0.3 0.4 0.7 0.3 0.6 1.0 Control/Display Unit (CDU) Weight (pounds) Volume (cubic feet) 0.4 4.0 4.0 | Integrated | in RPU 5.0 5.0 ÷ Integrated 0.1 in RPU 0.1 . Cost Range (\$ thousands) 1 to 7 | 3 to 10 2 to 60 10 to 30 20 to 40 5 to 15 48 100 75 to 90 *Insufficient data. Legend: This table is based on the following units: X - Available Not available Hon-RHAV VOR/DHE GPS Collins VIR-351 Narco DHE-890 RMAV VOR/DHE Design Studies Doppler Bendix DRA-12 Colling ANS-31, ANS-351 Marconi AD660 Bendix MP-2041A King MMR-665A, KNE-80 Foster AirDate VNAV-541, AD-611 Conventional INS Litton LTN-51, LTN-72, LTN-72R Delco Carousel IV, Carousel IV-A J.E.T. DAC-2000 Strapdown INS Litton LTN-90 Locan-C Teledyne TDL-711, TDL-424 Offshore Navigation ONI-7000 Honeywell RLG IRS Ô Nega Litton LTM-211 GJobal Cmega ONS-VII GJobal Nev ONS-500A Series 2 Collins LDN-80, LRN-85 Canadian Marconi CNA-771

T

I

٩,

٠.

ż

i

WHEN THE REAL PROPERTY OF THE PARTY OF THE P

CHAPTER THREE

PERFORMANCE REQUIREMENTS OF NAVIGATION SYSTEMS

3.1 OVERVIEW

The process of selecting a navigation system or a particular mix of navigation systems for adoption as the United States standard involves a series of comparisons. The essential question to be answered is: Which candidate system(s) can provide a level of navigation capability that surpasses all others? The comparison cannot be based on a singular issue such as system accuracy, however. The criteria that must be used are the following:

- Technical Institutional
- Operational Economic

Technical considerations are primarily concerned with integrity, reliability, coverage, and accuracy; the means of achieving accuracy is an operational issue. The intent of this report is to evaluate only those elements of the technical and operational considerations which contribute to navigation system accuracy. Both technical and operational considerations are related to the environment in which the navigation system will be used. The airspace environment has been divided into two basic phases of air navigation:

- Approach/landing
- En route/terminal

The approach/landing phase includes flight operations generally conducted within 10 nm of the runway in preparation for touchdown. Two categories of approach are defined within this phase: nonprecision and precision.

The en route/terminal phase of flight is divided into categories defined by the following particular geographic areas and operating environments:

- Oceanic en route Remote areas
- Domestic en route Helicopter operations
- Terminal

3-1

كالالا فيطلقهم والتركية وتشارعه المدرية تصادره

and a second second second second second and second s

Institutional considerations are concerned primarily with the effects and resolution of political issues such as international standardisation; distribution of costs; and system ownership, control, operation, and maintenance. Economic considerations, for both ground and airborne facilities and equipment, include the initial investment; operating, maintenance, and replacement costs; and amortization of the capital investment.

3.2 TECHNICAL CONSIDERATIONS

The system-use accuracy necessary to meet current route requirements is summarized in Table 3-1, which is taken directly from Volume II of the July 1980 Federal Radionavigation Plan (FRP). "System-use accuracy" is defined by ICAO to be the square root of the sum of the squares (root sum square [RSS]) of the following error contributions:

- Ground station error
- Airborne receiver error
- Display system error
- Flight technical error

System-use accuracy is a measure of the ability of a navigation system user to remain within a specified distance (route width) from a desired point (track). Navigation accuracy is generally expressed in terms such as following:

- Prodictable accuracy (also called absolute accuracy): the accu A consociated with predicting position with respect to geograph C ordinates
- Reputtion is y: the accuracy with which a user can return to a position of the same navious of the same

Accuracy defines the a fraction between a measurement value and a reference value; precision defines the degree of refinement to which the value can be expressed.

System-use accuracy correspond. In predictable, or absolute, accuracy in a statistical sense. The interprivate in 2-4rms accuracy values (shown in Table 3-1) is that the along-truck and cross-track error components cannot combine to yield a result enceding the value given for 95 percent of the measurements taken.

Tab	<i>le 3-1.</i> com	NTROLL	ED AIRSPACE N	AVIGATION ACCU	RACY TO MEET	CURRENT REQUI	STREAM
į	1		Altitude (Filght Level)	Traffic Density		Access 2 des	System-Use Accement
En Route/	Cent.		FL 275 to 400	Kene	8		12.e. 1
Terrine	Demostic		FL 100 to 000	191	*	2	7,280
	1		A	Normel	•	9891	3,000
			500 - 16,000 ft."	1	•	889	3,000
	Torning		500 - 10,000 ft.*	1	•	-	1,800
	Remote		500 - 00,000 ft.*	Low	839		3,000 to 14,400
			200 - 2000 ft. •	Lew (Off-Shene)	Net Determined	1000 1 2000	3,000 to 7,200
			508 - 3000 ft.°		•	8	1,200
	Rea Pacific]	~	\$	<u>2</u>
		-			± 9.1 meen [2]	± 3 mm ⁽³⁾	
		2		Morte			
)		1. 1. 1.		± 4.6 meter	± 1.4 meters	
		I X			± 4.1 motors	± 0.5 metre	
					1	artere .	
(1) The deal	bution of this error	ar is detail	And in the "Report	of the Limited North) Admits Report	Air Murigadian	
(1) the call	an is lateral pode na is vertical pode artera	ion 2 sig tion 2 sig	na accuracy in met	an far Pracision Appr um far Pracision App	parts and Landing. process and Landing.		

I

7

,,,,

.

1

18

Source: Federal Radionavigation Plan.

1

調

3-3

Table 3-1 does not reflect the system-use accuracy requirements for area navigation systems. FAA Advisory Circular (AC) 90-45A provides the requirements for approval of area navigation systems for use in the United States National Airspace System (NAS). Figure 3-1, reproduced from AC 90-45A, presents the maximum allowable errors for VOR/DNE-based area navigation systems. The figure indicates allowable cross-track and along-track errors as a function of distance from the VORTAC reference. Rather than defining

DISTANCE ALONG TRACK FROM TANGENT POINT

		0	5	10	. 15.	20	25	30	35	40	50	60	70	80	90	100	110	120	130	130	100
	0(n trk) (alg trk)		;\$:\$	13	1.3	¹ :\$	1:8	1:1	1:]	1:1	1:1	1:1	3:3	3:1	1:6	313	318	\$:\$	2:3	13.4
	b(m trn) (alg trk)	:1	3	:1	1:1	1.4	1:8	1:1	1:1	1:1	1:1	1:1	3:1	1:3	3:3	3:1	3:1	3:3	1:8	1:1	****
	10(x trk) (#18 trk)	:1	:1	:1	13	1:8	1:1	1:3	1:1	1:3	1:3	3:8	1:3	\$:0	1:1	3:1	3:1	3:3	1:8	1.1	42.4
	15(m LTh) (#1m LTh)	1:1	11	1.1	1:1	1:1	13	1:2	:::	ł::	2:3	}: †	\$:\$	3:9	3:3	3:1	- 3:3	3:2	1:1	2:3	12.4
	20(2 tra) (818 tra)	13	1:2	1.2	1:1	1:3	1::	<u>:</u> ;;	1:1	1:\$	}: †	3:5	1:1	1.1	3:1	3:2	3:2	3:3	11	2.8	. *:: :
5	25(n trh) (alg trk)	1:1	1:0	1:}	ł:}	1:1	1:1	1:1	2.0	3:1	1:3	3:3	1:}	3:8	3:3	1:3	3:3	1:3	13	. :::	43
	JO(R LPK) (alg trk)	1.9	:: ‡	1:3	1:4	1:1	1:1	1:1	13	1:3	13	3:3	2:3	3:1	3:1	33	3:7	1:1	¥.2 4.3	3:6	12.3
10	JS(# LPk) (Alg LPk)	1:2	1:3	1:3	1:3	3:3	2.0	£3	13	3:1	1:1	3:8	3:1	3:2	3:8	3:3	?.î	3:1	8.2 4.6	3:3	12:3
IST	40(# trk) (Alg trk)	13	1:5	1:1	1: :	1:2	<u>}</u> ;}	1:3	::	1:1	3:8	3:2	3:4	3:2	3:1	1:0	2:3	1:5	1:1	3:3	43
AR 0	30(n trk) (alg trk)	3:1	3:3	\$:3	3:3	3:2	3:3	3:3	3:8	3:2	3:4	\$:¥	4:7	13	23	: ;}	1:)	1:1	· • • • • • • • • • • • • • • • • • • •	3:3	17:8
D	60(a tra) (41g tra)	3:3	3:2	2.0	3:1	3:2	3:3	3:3	3.8	3:8	2 :1	13	2:2	\$: \$::2	\$:8	3:1	3:4	3:8	2:5	43
	70(± trk) (alg trk)	2:2	1.1	2:2	::	1:1	1:3	1:3	2:2	3.4	2:3	1:1	3.0	}: }	33	\$:}	3:3	3:3	8:3 6:0	::3	43
PCK.	80(s trk) (alg trk)	3.0	3.3	2.4	2.7	\$: 1	3:1	3:1	3:1	3:1	\$:3	\$:\$	3:3	3:3	\$.2 5.8	*: 8	2:1	8.0 8.3		9.8 6.9	12.8
	90(x trk) (alg trk)	2.# 5.0	3: 4	3:2	3.0	3:7	3:3	3:\$	3:1	3:8	\$:}	4:8	2:2	3:8 6:2	::	::	13	8.1 6.8	1.0	7:2	43
	100(m trk) (81g trk)	3.0	<u>}:1</u>	3:1	2:3	3:3	2:3	13	3:2	4.0	4.3	.5.0 6.0	:;}	\$.0 \$.8	: ;;	3:1	3:1	9:2	7: 5	10.0	**:3
	110(± trk) (alg trk)	3:3	3:1	3.4	2:3	::;	3:\$	2:5	\$:Ì	;: }	; :1	}:]	1:3	\$: ?	; ;;	7.2	3: 8	;;	: :7	10.1	8).9
	120(x trh) (A1g trk)	3:2	}: }	3:3	7:5	<i>}:</i> }	; ;;	\$; 2	7:5	\$: ?	3:3	}: ‡	}: ‡	\$.3 8.0	: :1	1:1	1:3	: :3	9.0 8.6	10.2	13:1
	130(m trk) (Alg trk)	3:1	\$.0 \$.1	4.0 4.1	:: 1	1:1	<u>8</u> 3	`1 ;3	Å:\$	1:1	13	1:1	\$: \$	1:1	2:9	2:3	1:1	9:0	; ;}	10.3	13:1
	140(x trk) (41g trk)	: ;}	: }	1:7	1:7	1:7	1:1	1:8	1:1	3.0 8.8	3:3	3:8	; ;}	\$: <u>?</u>	; ;;	3:2	9 :3	8.8 9.6	; ;}	10.3	13:1
	130(x trk) (41g trk)	\$:3	;;	\$:5	\$:3	\$: \$	\$:2	3:0 9:4	3:2	3:3	; ;;	\$.0 9.\$; ;}	1:2	;; ;	10:0	10.1	9.0 10.2	10:3	10.6	11:2
																					•

	TO FIND THE CROSS-TRACK AND ALONG-TRACK ELECTRONIC SYSTEM ERROR (LESS PILOTAGE) AT A POINT, ENTER TABLE WITH PERPENDICULAR DISTANCE AND DISTANCE ALONG TRACK FROM TANGENT POINT, I.e., when the Perp Dist is 30 and the along-track dist is 50, the cross-track error is ± 3.3 NM and the along-track error is 2.5 NM.	ERROR ELEMENTS GROUND VOR DME AIRBORNE VOR DME DME	I, 9 ⁰ 0. I NM 3. 0 ⁰ 3% or 0. 5 NM
Tangent Pt	ALONG TRACK DIST	PILOT	ZERO

Source: FAA Advisory Circular 90-45A.

Figure 3-1. VOR/DME/TACAN STATION-REFERENCED AREA NAVIGATION ERROR TABLE (95% PROBABILITY)

3-4

distance from the VORTAC as radial distance, Figure 3-1 resolves the distance into two orthogonal components (illustrated at the bottom of the figure). The radial distance is the hypotenuse of the right triangle thus formed.

As an example of how Figure 3-1 is used, consider an aircraft flying along a flight path that is offset from the VORTAC by 30 nm. This offset distance, sometimes referred to as the abeam distance, is defined as the perpendicular distance from the VORTAC to the flight path. At the time of interest, assume that the aircraft is 50 nm from the abeam point, from which the abeam distance to the VORTAC is measured. The block drawn in Figure 3-1 indicates the allowable error associated with an abeam distance of 30 nm and an along-track distance of 50 nm from the abeam point: ± 3.3 nm cross-track and 2.5 nm along-track. The allowable error for any distance can be found in this manner.

Table 3-2 summarizes the requirements specified for non-VOR/DME-based two-dimensional (2-D) RNAV systems. All error values correspond to 95percent confidence levels.

Table 3-2.	ALLOWABLE RNAV SYST	e error bu Tems (nau:	JDGETS FOI FICAL MILI	r non-vor/ es)	'DME-BASEI	D
		Flight Ph	ase Allow	able Erro	r Budgets	
Type of Error	En R	oute	Term	inal	Nonpre Appr	cision oach
	Cross- Track	Along- Track	Cross- Track	Along- Track	Cross- Track	Along- Track
Equipment	1.50	1.50	1.12	1.10	0.33	0.30
Flight technical	2.00	0.50				
Total system error (RSS)	2.50	1.50	1.50	1.10	0.60	0.30

3.3 OPERATIONAL CONSIDERATIONS

Operational requirements for navigation systems have been published in the FRP, Volume II, Section 2.1.1, titled "Aviation Requirements." These requirements are quoted below:

> "A. The system must be suitable for use in all aircraft types which may require the service without unduly limiting the performance characteristics of those aircraft types, e.g., maneuverability and fuel economy.

- "B. The system must be safe, reliable, available and capable of providing service over all the used airspace of the world, regardless of time, weather, terrain, and propagation anomalies.
- "C. The overall integrity of the system, including the presentation of information in the cockpit, shall be as near 100 percent as is achievable and to the extent feasible should provide flight deck warnings in the event of failure, malfunction, or interruption.
- "D. The system must have a capability of recovering from a temporary loss of signal in such a manner that the correct current position will be indicated without the need for complete resetting.
- "E. The system must automatically present to the pilot adequate warning in the case of malfunctioning of either the airborne or source portions of the system, and assure ready identification of erroneous information which may result from a malfunctioning of the whole system or incorrect setting.
- "F. The system must provide in itself maximum practicable protection against the possibility of input blunder or misinterpretation of output data.
- "G. The system must provide adequate means for the pilot to check the accuracy of airborne equipment.
- "H. The system must employ navigational information source equipment which automatically and radically changes the character of its indication in case a divergence from accuracy occurs outside safe tolerance.
- "I. The system must employ navigational information source equipment which provides immediate and positive indication of malfunction.
- "J. The navigational information provided by the system must be free from unresolved ambiguities of operational significance.
- "K. Any source-referenced component of the total navigation system shall be capable of providing navigational information simultaneously and instantaneously to all aircraft which require it within the area of coverage.
- "L. The navigation system must be capable, in conjunction with other flight instruments, of providing to the pilot in convenient, natural, and rapidly assimilable form in all circumstances, and the appropriate phases of flight, information directly
applicable to the handling of the aircraft, for the purpose of:

- 1. Continuous track guidance.
- 2. Continuous determination of distance along thack.
- 3. Continuous determination of position of aircraft, as resolved by the navigation system.
- 4. Position reporting.
- 5. Manual or automatic flight.

The system shall also provide for input and utilization of the above in conveniently operable form; and must permit design of indicators and controls which can be directly interpreted or operated by the pilot at his normal station aboard the aircraft.

- "M. The system must be capable of being integrated into the overall ATC, communications, and navigation system.
- "N. The system should be capable of integration with all phases of flight, including the precision approach and landing system.
- "O. The system must permit the pilot to determine the position of the aircraft with an accuracy and frequency such as to ensure that the separation minima used can be maintained at all times, execute accurately the required holding and approach patterns, and to maintain the aircraft within the area allotted to the procedures.
- "P. The system must permit the establishment and the servicing of any practical, defined, route structure for the appropriate phases of flight as required.
- "Q. The system must have sufficient flexibility to permit changes to be made to the air-route structure and siting of holding patterns without imposing unreasonable inconvenience or cost to the providers and the users of the system.
- "R. The system must be capable of providing the information necessary to permit maximum utilization of airports and airspace.
- "S. The system must be cost-effective to both government and users.

- "T. The system must employ equipment such as to minimize susceptibility to interference from adjacent radio-electronic equipment and shall not cause objectionable interference to any associated or adjacent radio-electronic equipment installation in aircraft or on the ground.
- "U. The system must be free from signal fades or signalto-signal plus noise ratios below which the system cannot operate in the operating area.
- "V. The system avionics must be comprised of the minimum number of elements which are simple enough to meet, economically and practically, the most elementary requirements, yet be capable of meeting, by the addition of suitable elements, the most complex requirements.
- "W. The system must be capable of furnishing reduced service to aircraft with limited or partially inoperative equipment.
- "X. The system must be capable of integration with the flight control system of the aircraft to provide automatic tracking.

"No one set of aviation navigation operational requirements, even though they meet the basic requirement for safety, can adequately reflect the many different combinations of operating conditions encountered in various parts of the world, in that the requirements applicable to the most exacting region may be extravagant when applied to others."

3.4 INSTITUTIONAL CONSIDERATIONS

In addition to the technical and operational considerations, a number of institutional issues will play a major role in determining which mix of navigation systems will be used in the future. Although it is not within the scope of this study to evaluate in detail the influence that institutional issues will have on the selection process, some factors are considered to be significant enough to be discussed briefly.

3.4.1 International Standardization

The United States (and consequently the FAA) is an active participant in the international aviation community as represented by ICAO and other organizations. One function of ICAO is to facilitate international aeronautical activity and cooperation through coordination and recommendation of standardized equipment and procedures. In 1959, ICAO approved VOR/DME as the international standard for aeronautical navigation through 1984. Of the 2,609 VOR/DME facilities that currently exist worldwide, 1,152 are registered in the ICAO Air Navigation Plans as international navigation aids. An additional 466 international facilities are scheduled for installation according to these plans. Consequently, there is strong international sentiment for maintaining VOR/DME as the international standard at least through 1995 -- a proposal that ICAO has recently adopted. Past experience has demonstrated that a particular type of navigation aid (e.g., Loran-A) continues to be used for 15 years or more after its official protection ceases. This being the case, it could be assumed that VOR/DME will continue to be used worldwide into the year 2010.

ICAO has no plans, formal or informal, to endorse either Loran-C, Omega, or GPS as an alternate or replacement international standard aeronautical navigation aid. Loran-C coverage areas are unlikely to extend beyond what is currently planned, which still excludes the southern hemisphere. Omega already provides essentially worldwide coverage, and the necessary international agreements for maintenance and operation of the transmitting stations have been negotiated and accepted. GPS is currently a United States military-sponsored satellite navigation system; its implementation does not require international cooperation other than for frequency allocation.

One of the central issues concerning international, and frequently domestic, endorsement of a particular navigation system standard involves the responsibility of control. The organization that controls the system can theoretically limit access to the system. Because VOR/DME, Loran-C, and, to some extent, Omega are systems where the control responsibility is distributed among many nations and organizations, they will likely remain available to all users of all nations. GPS, however, is a system that inherently requires centralized control. This control will have to be vested in either a single country or an international organization. If the former occurs, international support is likely to be difficult to obtain. If the latter occurs, the system may become unappealing to its most prominent supporters in the United States, and its acceptance may be delayed while institutional arrangements are prepared for international GPS management.

Intertwined closely with the issue of system control is allocation of operation and maintenance responsibility and cost. The distributed regional systems such as VOR/DME and Loran-C are well suited for the international environment, since the using, controlling, operating, and maintenance functions are typically the responsibilities of the country in which the facilities are located. Omega also presents no difficulties in this context, because the operational and maintenance considerations have already been agreed to by treaty. GPS, however, is likely to encounter significant institutional difficulties in allocation of operation and maintenance responsibility and cost.

3.4.2 Transitional Impact

National governments as well as the various user communities have significant capital investments in the existing VOR/DME-based navigation system. These investments must be protected over an extended period of time while any new system is being phased in, to allow for full amortization of the old system. The institutional difficulties of effectively maintaining standards for two navigation systems over a period of possibly 15 to 20 years may be so cumbersome and uneconomical as to force retention of the VOR/DME system until the successor system can have a guaranteed useful life of at least 30 to 50 years.

3.4.3 Special Interest Groups

The acceptability and desirability of any single navigation system or combination of systems will ultimately be determined by the user community, either directly or indirectly. For example, the airlines provide the impetus for the application of Omega to en-route navigation in the continental United States, offshore oil prospectors generate the market for Loran-C, and private pilots continue to provide demand for VOR/DME. User preferences are evidenced through public statements of trade associations and, more important, through equipment purchases.

3.5 ECONOMIC CONSIDERATIONS

Both the Government, as the provider of navigation capability, and the user, as the purchaser of equipment for aircraft, are concerned about the life-cycle cost of the navigation system or combination of systems selected for future use. Consequently, numerous cost analyses have been performed for alternative navigation scenarios. The following two reports have quantified the total Government and user life-cycle costs of various navigation system combinations:

- Economic Requirements Analysis of Civil Air Navigation Alternatives, Reference 29
- Economic Analysis of Future Civil Air Navigation Systems, Reference 30

In broad terms, these studies have concluded that neither Loran-C, GPS, nor Omega can compete in cost with the single VOR receiver for low-budget aviation. However, if area navigation becomes the standard of the air traffic control system, the specific combination of systems used will have relatively little cost impact as long as the existing user capital investment in VOR/DME is utilized concurrently with any newly introduced system (e.g., VOR/DME with Loran-C or VOR/DME with GPS). If the existing VOR/DME network is discarded in favor of a totally new navigation concept such as GPS, the cost to retrofit all existing aircraft with the new navigation equipment will exact a significant cost penalty from the users, as compared with the other approaches. The quantitative conclusions of these two reports are summarized in Table 3-3.

Table 3-3. LIFE-CYCLE-COST COMPARISONS OF NAVIGATION SYSTEMS		
Navigation System(s) Used	Total Life-Cycle Cost to U.S. Government and Users Normalized to Baseline	
	SCI*	MITRE Corp.**
VOR/DME (RNAV not required) in U.S.; Omega for global navigation	1.00 (baseline)	1.00 (baseline)
VOR/DME with Loran-C (RNAV required) in U.S.; Omega for global navigation	Not Studied	1.13
VOR without DME (RNAV not required) in U.S.; GPS for RNAV and global navigation	1.20	1.08
RNAV Loran-C in U.S.; Omega for global navigation	1.20	1.24
RNAV GPS both in U.S. and for global navigation	1.40	1.40
*Systems Control, Inc., Economic Requirements Analysis of Civil Air Navigation Alternatives, Reference 29. **MITRE Corp., Economic Analysis of Future Civil Air Navigation Systems, Reference 30.		

1

1.

CHAPTER FOUR

TECHNICAL EVALUATION OF NAVIGATION SYSTEMS

4.1 OVERVIEW

The technical performance of a navigation system is strongly influenced by the properties of the received signal as well as the techniques used by the airborne receiver in processing the signals. Significant signal properties include stability, coverage, and availability. Signal acquisition and tracking continuity depend on both signal and receiver characteristics.

Accuracy of hyperbolic radionavigation systems such as Loran-C and Omega is primarily dependent on the accuracy and precision with which time or phase differences can be measured. When ranging techniques are used for Loran-C and Omega navigation, achievable accuracy is a function of the time synchronization of the transmitters, user clock biases, accuracy of propagation modeling, and the geometry of the user's position in relation to the transmitting stations.

Accuracy of VOR/DME is primarily influenced by the inaccuracies of VOR. The dominant VOR errors are caused by multipath effects. Multipath signals are caused by reflecting objects near the transmitter that scatter the original signal. These multipath signals create distortions in the desired signal resulting from different paths being simultaneously traveled by the signal between transmitter and receiver. The result is scalloping in the VOR bearing indications.

Accuracy of the space-based GPS radionavigation concept depends on accurate and continuous knowledge of the spatial position of each satellite in the system with respect to time and distance from the user.

Accuracy of Doppler navigation systems degrades as a function of the distance traveled from a known position. The accuracy capability is mainly dependent on the accuracy of the heading reference used.

Accuracy of conventional and strapdown INS degrades as a function of time. The accuracy is principally influenced by accumulated gyro drift.

Although compliance of a navigation system with technical performance requirements such as accuracy is necessary, it is not a sufficient condition for acceptability. Also important are operational considerations relating to the ability of the system to intoract with the pilot and ATC to meet and maintain operational standards of safety and accuracy. The summary of features of typical navigation units presented in Table 2-1 provides an indication of the operational compatibility of various navigation systems in terms of the system-user interface. The following sections present the technical capabilities of each system in terms of an error budget.

4.2 ERROR BUDGET FOR LORAN-C

The error budget for Loran-C is broadly divisible into five categories as follows:

- Transmitter errors
- Signal detection errors
- Receiver clock errors
- Signal propagation variations
- Position fix calculation errors

Table 4-1 presents a summary of the effect of specific errors on the navigational accuracy of various Loran-C system use categories. As the table shows, inaccurate prediction of signal propagation variations is the main contributor to the total RSS system accuracy, excluding flight technical error. Although it may appear that hyperbolic navigation is more accurate than either rho-rho or rho-rho-rho navigation, that is not the Case in an operational environment. When user distance from a Loran-C baseline is large in relation to the length of the baseline, the user is in a high GDOP situation. The relatively short baselines of Loran-C chains result in typical GDOPs of two to four when hyperbolic navigation is used. Use of rho-rho or rho-rho-rho techniques effectively eliminates GDOP considerations. Each error source is discussed in the following sections.

4.2.1 Transmitter Errors

Loran-C receivers presume that the received signal originated at a particular instant at a specific location. Jitters in the timing of the Loran-C pulse or errors in transmitter location are interpreted by the receiver as variations in signal propagation time. These propagation variations in turn directly translate into position fix errors. The errors presented are based on an assumed signal propagation velocity of 161,829 nm/sec.

4.2.1.1 Timing Error

Loran-C transmitters are synchronized with each other via extremely accurate internal atomic clocks and information from various monitor

Table 4-1. POSITION BRROR BUDGET SUMMARY FOR LORAN-C			
Source of Error	Effect of Error (inm) on Navigation System Use		nm) on Nuse
× .	Hyperbolic	Rho-Rho	Rho-Rho-Rho
Transmitter Errors			
Timing Location	0.03 8 × 10 ⁻⁴	0.03 8 × 10 ⁻⁴	0.03 8 × 10 ⁻⁴
Signal Detection Errors			
Zero crossing detection Phase comparison quantization	2.4×10^{-3} 3×10^{-3}	2.4×10^{-3} 3×10^{-3}	2.4×10^{-3} 3 × 10^{-3}
Cycle jump*	1.62	1.62	1.62
Recaiver Clock Errors **	Ч. У У		
Initialization Drift	Negligible 1.3 × 10 ⁻⁶	0.06 3 × 10 ⁻³ †	0.06 1.3 × 10 ⁻⁶
Signal Propagation Variations			
Prediction Random*	0.06 0.01	0.06 0.91	0.06 0.01
Position Fix Calculation Errors*			
Earth model Dead reckoning	0.2 5.5 × 10 ⁻⁴	0.2 5.5 × 10 ⁻⁴	0.2 5.5 × 10 ⁻⁴
Total System Position Error (RSS 2 drms)			
GDOP = 1 GDOP = 4	0.07 0.26	0.09	0.09
*Excluded from RSS calculation. **Quartz clock unless otherwise noted. †Rubidium clock or better.			

stations. The most critical types of timing errors include lack of proper signal phase or envelope coherence among stations, as follows:

- Master station timing error
- Master-slave timing error

IJ

Π

- Slave-slave timing error
- Chain-chain timing error

The United States Coast Guard, which operates the Loran-C system in the United States, has specified the following four operational modes for transmitters:*

- "Optimum achieved 95% of the time; full power timing precision of ± 40 h seconds;
- "Precision achieved 97% of the time; half power timing precision of ± 40 n seconds;
- "Enhanced achieved 98.6% of the time; full power timing precision of ± 200 n seconds; and
- "Standard achieved 99.7% of the time; half power timing precision of ± 200 n seconds."

For this discussion, the standard operating mode is used to determine transmitter errors. Therefore, the sum of all timing errors within the transmitters is assumed to be less than ± 200 nanoseconds (ns), corresponding with a propagation distance error of ± 0.03 nm (± 55.6 meters).

4.2.1.2 Station Location Error

Published locations of Loran-C stations are accurate to 0.1 arc second. Any variation of actual signal origin from these published position coordinates is reflected directly into an erroneous range measurement from the. station. An accuracy of 0.1 arc second is the same as an error of ± 0.05 arc second, which corresponds to a position fix error of $\pm 8 \times 10^{-4}$ nm (± 1.5 meters) when assuming 60 nm/deg.

4.2.2 Signal Detection Errors

Each Loran-C pulse contains 20 cycles of radio frequency (RF) energy and exhibits a slowly rising and decaying amplitude envelope shaped much like a teardrop. The energy contained in the first two cycles of the pulse is insufficient for a good signal-to-noise ratio measurement. The fourth cycle is likely to be contaminated by sky waves. Therefore, receivers typically are designed to measure pulse arrival at the third cycle. The most accurate technique for measuring time of signal arrival is to detect the third pulse zero crossing and compare that time to a local oscillator. The exact time is measured in terms of a phase relationship between the local oscillator and the received signal. The following sections describe the three sources of error that contribute to Loran-C position uncertainty.

4.2.2.1 Zero Crossing Detection Error

The zero crossing is derived by comparing the squared-off Loran RF signal with the local oscillator clock. The squared-off Loran signal is derived by means of such a device as a saturating amplifier. There is

*A Comparative Analysis of Area Navigation Systems in General Aviation, Reference 28. approximately 1 degree of error in identifying the precise moment of zero crossing. This 1-degree error in a 100-kHz signal corresponds to 0.03 microsecond, or a propagation distance error of $\pm 2.4 \times 10^{-3}$ nm (± 4.4 meters).

V

4.2.2.2 Phase Comparison Quantization Error

The phase difference between the local oscillator and the received Loran pulse is usually measured with a digital counter that runs several times faster than the basic RF carrier rate of 100 kHz. Current engineering designs typically provide resolution of 40 ns (± 20 ns error), corresponding to a propagation distance error of $\pm 3 \times 10^{-3}$ nm (± 5.6 meters).

4.2.2.3 Cycle Jump Error

Under certain atmospheric propagation conditions, differences between the phase and group velocities in the RF signal affect the shape of the Loran-C pulse so that it is distorted by the time it reaches the receiver. This pulse distortion may deceive the receiver into miscounting the true cycle crossings and thus locking onto the second or fourth cycle of the Loran pulse. This phenomenon is called cycle jump. Most Loran receivers have special circuitry and logic to prevent this from happening and to warn the operator if it does occur. The position error introduced by cycle jump is the equivalent propagation time of one Loran-C cycle, or 10 microseconds, corresponding to a propagation distance error of ± 1.62 nm.

4.2.3 Receiver Clock Errors

There are two types of clock errors. The first type is a function of initial synchronization, and the second is a function of stability.

4.2.3.1 Clock Initialization Error

In the hyperbolic mode, clock initialization errors have minimal effect on position accuracy, because only the time difference between incoming signals is measured, resulting in the cancellation of initialization errors. In the ranging mode, initialization errors are significant, because they add a constant error to the position fix. The accuracy of the initial clock synchronization is limited by knowledge of the exact distance from the receiver at the time of initialization, knowledge of the propagation characteristics at the time of initialization, and the receiver's signal detection accuracy at the time of initialization. These different types of error are presented in Table 4-2.

4.2.3.2 Clock Drift Error

If the local oscillator is unstable and drifts in relation to the transmitter clocks, the airborne system perceives the drift as a shift in time difference and computes an erroneous position.

Table 4-2. RECEIVER INITIALIZATION ERRORS			
Source of Error	Derivation	Report Section	Magnitude (±nm)
Range from transmitter	Limitation of accuracy of transmitter and receiver loca- tions to ±0.05 arc second	4.2.1.2	8 × 10 ⁻⁴
Propagation	Errors in prediction	4.2.4	0.06
Signal detection	Phase comparison quantization	4.2.2	3×10^{-3}

In the hyperbolic mode, only short-term clock stability is important. With a quartz crystal clock, drift is about 1.7×10^{-11} seconds for a representative measurement interval of 0.05 seconds. This drift corresponds to a propagation distance error of $\pm 1.3 \times 10^{-6}$ nm (± 0.002 meters).

In the rho-rho mode, the clock is free-running during each leg of the flight and thus may not be resynchronized for two to six hours. During a six-hour flight, a quartz clock may drift 7.2 microseconds, corresponding to a propagation distance error of ± 0.6 nm. A rubidium clock is more stable and will reduce this error to 3.6×10^{-8} seconds, or $\pm 3 \times 10^{-3}$ nm (± 5.6 meters).

In the rho-rho-rho mode, a third Loran-C station is used to estimate and compensate for the error resulting from clock drift. Therefore, the effect of drift will be minimal for any high-quality clock used in the receiver.

4.2.4 Signal Propagation Variations

Accurate modeling of the propagation characteristics of the Loran-C signal enhances the accuracy of the system. Differences in propagation characteristics between the model and the received signal result from errors in predictive modeling and the occurrence of random, non-predictable variations in signal propagation.

4.2.4.1 Prediction Error

The propagation velocity of electromagnetic waves varies slightly as a function of the medium through which the waves are passing. For Loran-C, the characteristics of the medium depend on surface effects and atmospheric conditions. Surface effects are important, because Loran-C receivers derive range information from ground waves traveling along the surface of the earth. Propagation models that assume that the ground waves travel over water will introduce position errors when used over land. The uncertainty in determining the correct propagation velocity from the transmitter to the user is called prediction error. Estimates of the magnitude of prediction error are typically around ± 0.4 microsecond, corresponding to a propagation distance error of ± 0.06 nm (± 111.1 meters). Those Loran-C units that store propagation corrections to account for the land/sea shift minimize this error.

4.2.4.2 Random Error

Ľ

Many sources of error in signal propagation cannot be adequately predicted. Their effect on position accuracy is primarily a function of the limit of signal detection possible, as a consequence of noise. The operational limit of most radionavigation systems, including Loran-C, is determined by the signal-to-noise ratio (SNR) at which the receiver must operate. Analyses have shown that the unpredictable error in an excellent SNR environment (1:1) is about 0.05 microsecond, or $\pm 4 \times 10^{-3}$ nm. In a typical SNR environment (1:3), unpredictable errors increase to a value of about 0.14 microsecond, corresponding to a propagation distance error of ± 0.01 nm (± 18.52 meters).

4.2.5 Position Fix Calculation Errors

The ability of Loran-C units to determine geographical position is dependent on the accuracy with which the unit converts transmitter-to-user range information to earth coordinates. Inaccuracies in the interpretation of the range measurements, due either to the effects of sudden changes to the input or improper earth modeling, will degrade position fix accuracy.

4.2.5.1 Earth Model

Once the range from the Loran-C transmitter has been determined, the geographical position of the user is computed on the basis of a mathematical model of the earth. Position fix errors resulting from the lack of agreement between earth models can be as great as ± 0.2 nm (± 370.4 meters).

4.2.5.2 Dead Reckoning

Initiation of a maneuver or a sudden change in the wind vector introduces an error rate of 5.5×10^{-3} nm per second (based on an along-track wind change of 20 knots), which decays to zero as the tracking loop of the receiver compensates for the change in dynamics during successive update cycles. Between 0.1-second measurement updates of Loran-C, an error of 5.5×10^{-3} nm per second results in a position fix error of 5.5×10^{-4} nm (±1.0 meters).

4.2.6 Total System Position Error

The individual error contributions are combined in RSS fashion to compute the total system position errors (2 drms) for hyperbolic, rho-rho, and rho-rho-rho implementations. The RSS totals corresponding to a GDOP of 1 assume best geometry conditions. A typical total error corresponding to a GDOP of 4 is also shown in Table 4-1 for hyperbolic navigation. It is assumed that a GDOP of 1 can be achieved under most conditions when rhorho or rho-rho-rho navigation is used with Loran-C.

4.3 ERROR BUDGET FOR OMEGA

Errors for Omega are divided into the same five categories as defined for Loran-C:

- Transmitter errors
- Signal detection errors
- Receiver clock errors
- Signal propagation variations
- Position fix calculation errors

A summary of the affect of these errors on Omega navigation accuracy is presented in Table 4-3. As the table shows, propagation modeling errors overwhelmingly predominate and limit the overall accuracy of position fixing. The baselines between Omega transmitters are much longer than those for Loran-C; therefore, the typical GDOP associated with Omega hyperbolic navigation is 2, whereas it is 4 with Loran-C. The errors are discussed in the following sections.

4.3.1 Transmitter Timing Errors

Since the Omega system assumes perfectly synchronized and phase-locked VLF radio transmissions, any jitter will result in an erroneous position fix. The eight Omega transmitters are phase-locked by means of monitor stations and use very accurate atomic clocks at each station for frequency standards. Timing errors between stations are limited to 1 microsecond, corresponding to a propagation distance error of ± 0.08 nm (± 148.2 meters).

4.3.2 Signal Detection Errors

Position fixing for Omega in the hyperbolic mode requires comparison of the phase values of the received signals from several transmitting stations. In the ranging mode, the phase of the received signals is measured directly and not compared. Errors in phase detection and phase measurement quantization are described in the following sections.

4.3.2.1 Zero Crossing Detection Error

The phase of the Omega signal is measured by hard-limiting the signal and comparing signal zero crossing with reference clock zero crossing. The time difference between the two is converted to phase value.

Assuming a saturation ratio of 10:1, the transition from one limit to the other will take nearly 12 degrees $(2 \times \sin^{-1} [0.1])$ for a sinusoidal Omega signal). With 100:1 saturation, the transition is 1 degree. A 1-degree phase detection error corresponds to a position fix error of ± 0.02 nm (± 37.0 meters). (At an Omega frequency of 10.2 kHz, 16 nm corresponds to 360 degrees).

Table 4-3. POSITION ERROR BUDGET SUMMARY FOR OMEGA			
Source of Error	Effect of Error (inm) on Navigation System Use		
	Hyperbolic	Rho-Rho	Rho-Rho-Rho
Transmitter Timing Errors	0.08	0.08	0.08
Signal Detection Errors			
Zero crossing detection Phase measurement quantization	0.02 0.04	0.02 0.04	0.02 0.04
Receiver Clock Errors*			
Initialization Drift	Negligible 1.25 × 10 ⁻⁴	0.02 0.005**	0.02 2.5 × 10 ⁻⁴
Signal Propagation Variations			
Prediction Random [†]	1.6 2.5	1.6 2.5	1.6 2.5
Position Fix Calculation Errors†			
Earth model Dead reckoning	0.2 0.028	0.2 0.028	0.2 0.028
Total System Position Error (RSS 2 drms)			
GDOP = 1 GDOP = 2	1.6 3.2	1.6 	1.6
*Quartz clock unless otherwise noted. **Rubidium clock or better. †Excluded from RSS calculation.			

1

ł

4.3.2.2 Phase Measurement Quantization Error

T

¥*

Omega receiver: using hyperbolic techniques typically quantize phase in units of 1/100 of a phase-difference cycle (centicycle). Resolution accuracy along the baseline at a frequency of 10.2 kHz is therefore 0.08 nm, or ± 0.04 nm (± 74.1 meters).

4.3.3 Receiver Clock Errors

As with Loran-C, there are two types of clock errors for Omega. The first type is a function of initial synchronisation, and the second is a function of stability.

4.3.3.1 Clock Initialization Error

In the hyperbolic mode, clock initialization errors have minimal effect , on position accuracy, because only the phase difference between incoming signals is measured, resulting in the cancellation of initialization errors,

In the rho-rho mode, initialization errors are significant, because they add a constant error to the position fix. The clock is initialized by the use of either the four Omega navigation frequencies or differences between these frequencies. Frequency differences are used to resolve initial lane ambiguity. Nondifferenced frequencies subsequently are used for final correction of clock error.

If exact position and propagation characteristics are known at the time of initialization, and if the clock synchronization circuits provide comparable performance to the phase detection circuits, phase ambiguity of 1 degree can be expected in clock synchronization. This ambiguity corresponds to a position fix error of ± 0.02 nm (± 37.0 meters).

4.3.3.2 Clock Drift Error

If the local oscillator is unstable and drifts in relation to the transmitter clocks, the system perceives this drift as a phase shift and computes an erroneous position. Table 4-4 summarizes the effect of clock drift on navigation accuracy for different types of clocks.

The magnitude of position error is related to the elapsed time between independent measurements that enable timing synchronization. In a hyperbolic implementation, the elapsed time between phase measurements of the same frequency, but from two signals, is constant, although the value is dependent on which Omega station combinations are being tracked. It is therefore possible to synchronize the clock in accordance with this elapsed time. An elapsed time of five seconds is used in Table 4-4 to represent a worst-case station combination. For such short time intervals between timing synchronizations, the clock need have only good short-term stability. In fact, the effect of drift is minimal if any high-quality clock is used in the Omega receiver.

In the rho-rho mode, in which phase comparisons that allow timing synchronization are not utilized, the clock is free-running during each leg of the flight and thus may possibly not be resynchronized for two to six hours. Because of this extended operating period, clocks that are typically used are not stable enough to provide accurate rho-rho navigation over extended time periods.

Table 4-4. ESTIMATE OF CLOCK STABILITY EFFECTS ON POSITION ERROR FOR OMEGA NAVIGATION				
		Position Error (nm)		
Clock Type	Short-Term Stability*	Hyperbolic (5-Second Period)	Rho-Rho (6-Hour Period)	Rho-Rho-Rho (10-Second Period)
Quartz	3 × 10 ⁹	2.5×10^{-4}	1.8	5×10^{-4}
Rubidium	6 × 10 ¹¹	1.5×10^{-6}	1×10^{-2}	3×10^{-6}
Cesium	5 × 10 ¹²	1.5×10^{-7}	1 × 10 ⁻³	3×10^{-7}
Hydrogen Maser	1 × 10 ¹⁴	1 × 10 ⁻⁸	7 × 10 ⁻⁵	2×10^{-8}
*Short-term stability is expressed as one part per the value given. By dividing the time period of interest by the short-term stability value and then multiplying the result by the speed of light, the position error is determined.				

In the rho-rho-rho mode, a third Omega station is used to estimate and compensate for the error resulting from clock drift. This readjustment occurs every 10 seconds. Therefore, the effect of drift will be minimal for any high-quality clock used in the receiver.

4.3.4 Signal Propagation Variations

The accuracy of Omega is limited by the accuracy of the propagation corrections that are applied to the received signal. The corrections are developed on the basis of predicted signal behavior. Random, nonpredictable variations in signal propagation introduce errors that compound those inherent in the predictive modeling of signal propagation.

4.3.4.1 Prediction Error

Estimates of the error due to propagation modeling range from 3.2 to 8 nm. For this analysis, 3.2 nm, specified as ± 1.6 nm, is assumed to be a typical propagation prediction error.

4.3.4.2 Random Error

Many influences on signal propagation that occur infrequently cannot be predicted -- for example: sudden phase anomalies (SPAs), produced by sudden ionospheric disturbances (SIDs); and polar cap absorption (PCA) events, caused by the concentration of solar protons in the vicinity of the earth's magnetic poles. These anomalies can cause position errors of 2 to 8 nm -- an average error of 5 nm.

4.3.5 Position Fix Calculation Errors

Geographical position is determined by converting to geographical coordinates the position of the user relative to the Omega stations, as determined by the receiver. This position fixing process is susceptible to errors in earth modeling and uncertainty in assumed position as a function of dynamic response to sudden changes in input.

4.3.5.1 Earth Model

As in Loran-C navigation, Omega navigation requires the use of a mathematical model of the earth. The possibility of a position fix error of ± 0.2 nm is used in this analysis.

4.3.5.2 Dead Reckoning

If an error rate of $\pm 5.5 \times 10^{-3}$ nm per second is used (based on an along-track wind change of ± 20 knots), a 5-second dead-reckoning period before the first update will result in a position fix error of ± 0.028 nm. Some Omega/VLF units use processing techniques that reduce the period of dead reckoning. By processing the VLF signals in parallel rather than in series with the Omega signals, a more continuous update capability can be achieved.

4.3.6 Total System Position Error

The individual effects of the various sources of error are combined in RSS fashion to determine the total system position error (2 drms) for Omega. RSS totals corresponding to a GDOP of 1 are given in Table 4-3 for hyperbolic, rho-rho, and rho-rho-rho implementations. A typical total error corresponding to a GDOP of 2 is also shown for hyperbolic navigation. As discussed for Loran-C, when ranging techniques are used, it is assumed that a GDOP of 1 can also be achieved by Omega when supplemented with VLF.

4.4 ERROR BUDGET FOR VOR/DME

Errors for VOR/DME are divided into the following categories:

Bearing error components

- •• Ground component radial error
- •• Airborne component radial error
- · Course selection error
- Distance measurement errors
 - •• Ground component distance error
 - Airborne component distance error
- Area navigation computation error
- Total system error

Table 4-5 summarizes the effect of specific errors on navigation accuracy for the VOR/DME navigation system use categories. Each error source is discussed in detail in the following sections.

١

Source of Error	Effect of Error on Navigation System Use	
	Basic VOR/DME	RNAV VOR/DME
Bearing Errors		
Ground component Airborne component Course selection Distance Errors	±1.4 ^c ±3.0° (±1.0°**) ±2.0° (±0.5°**)	±1.4° ±3.0° (±1.0°**) ±2.0° (±0.5°**)
Ground component Airborne component	±0.1 nm Between 0.1 nm and 1% of range	±0.1 nm Between 0.1 nm and 1% of range
Area Navigation Computation Error	Not applicable	±0.5 nm
Total System Error (RSS 20)		
Radial	±3.9° (±1.8°**)	<u>+3.9° + 0.5 nm</u> (<u>+1.8° + 0.5 nm**)</u>
Distance	Between 0.5 nm (0.1 nm**) and 3% (1%**) of range	Between 0.5 nm (0.1 nm**) and 3% (1%**) of range

4.4.1 Bearing Error Components

3

Γ.

Some of the errors listed in the VOR/DME error budget can be attributed directly either to the VOR or DME element of the system. Bearing error reflects the error introduced specifically through use of VOR. The three individual components contributing to bearing error are described in the following sections.

4.4.1.1 Ground Component Radial Error (E gr)

Radial signal error is the difference between the nominal magnetic bearing from the VOR transmitter to a point of measurement and the bearing indicated by the signal at the same measurement point. Ground component radial error consists of (1) certain constant elements, such as course displacement errors and most site and terrain effect errors, which may be considered fixed for long periods of time, and (2) certain random variable errors that can be expected to vary about the essentially constant value. The ground component radial error is associated with the VOR transmitter and nominal signal path errors, but excludes other error factors.

Siting and propagation errors are the principal VOR signal errors. Alignment of electronic radials with magnetic radials, and deviations of signal because of roughness, scalloping, multipath effects, signal bending, and refraction, are mostly uncontrollable factors that contribute significantly to siting errors. Signal propagation variations in the atmosphere have been demonstrated to contribute to a course error of 0.2 degree.

Extensive data collection by the FAA indicates an E_{gr} error value of ±1.4 degrees (95 percent probability).

4.4.1.2 Airborne Component Radial Error (E ar)

Airborne component radial error is the error attributable to the inability of the airborne equipment to translate correctly the bearing information contained in the radial signal. This element embraces all factors in the airborne component that introduce errors in the information presented to the pilot. (Errors resulting from the use of compass information in some VOR and TACAN displays are not included.)

An E_{ar} value of ±3.0 degrees (95 percent probability) was used in defining the national aviation standard for the VORTAC system. The use of digital signal processing has reduced this error to a range between ±0.75 and ±2 degrees, according to various equipment manufacturers. A value of ±1 degree is specified in Table 4-5 as obtainable.

4.4.1.3 Course Selection Error (CSE)

Course selection error (CSE) is the accuracy limitation of the omnibearing selector (OBS) resulting from the resolution of the device and the error inherent in translating the pilot input to the avionic comparator. This comparator derives the difference between the actual computed course and the selected course entered by the pilot. This difference is usually displayed on a course deviation indicator (CDI). Errors of the CDI are considered to be part of $E_{\rm ar}$.

A requirement of ± 2.0 degrees for CSE was established on the basis of accuracies achievable with analog dials. The advent of digital processing and displays permits pilots to set a desired course to within the precision made available, either whole degrees, tenths of a degree, or better. In this analysis, ± 0.5 degrees will be assumed achievable.

4.4.2 Distance Measurement Errors

Distance measurement errors reflect the errors introduced specifically through use of DME. The two components contributing to distance measurement errors are described in the following sections.

4.4.2.1 Ground Component Distance Error (Egd)

According to current requirements, ground component range accuracy must be maintained to within 0.1 nm.

4.4.2.2 Airborne Component Distance Error (Ead)

The airborne DME unit measures and displays the slant range distance between the aircraft and the ground station. The accuracy of this information must be maintained to within 3 percent of the actual distance, or ± 0.5 nm, whichever is greater. The digital revolution has markedly improved the accuracy of airborne DMEs beyond standard requirements. Manufacturers commonly quote airborne equipment accuracy of ± 0.1 nm to 1 percent of DME distance for general aviation equipment.

4.4.3 Area Navigation Computation Error (Ec)

When RNAV computation is used, an additional error contribution is specified and combined with the basic VOR/DME system error. The additional maximum RNAV equipment error allowed, per FAA AC 90-45A, is ± 0.5 nm.

Computation error includes error components contributed (1) by any input, output, or signal conversion equipment used, (2) by any computing element used, (3) by the display as it presents either aircraft position or guidance commands (e.g., course deviation or command heading), and (4) by any course definition entry devices used. For systems in which charts are incorporated as integral parts of the display, E_c necessarily includes charting errors to the extent that these errors actually result in errors in controlling the position of the aircraft in relation to a desired path over the ground. To be consistent, for symbolic displays not employing integral charts, any errors in waypoint definition directly attributable to errors in reference charts used in determining waypoint positions should be included as components of E_c . This type of error is difficult to quantify; in general practice, highly accurate published waypoint locations are used to the greatest extent possible to avoid charting errors (and to reduce workload).

4.4.4 Total System Error (Es)

Assuming that the variable errors from various sources discussed are normally distributed and independent, the error components are combined in RSS fashion as follows:

System radial error
$$(E_{sr}) = \sqrt{E_{gr}^2 + E_{ar}^2 + CsE^2 + FTE^2 + E_c^2}$$

System distance error $(E_{gd}) = \sqrt{E_{gd}^2 + B_{ad}^2}$

4.5 ERROR BUDGET FOR GPS

GPS errors are divided into the following categories:

- Satellite errors
- Signal propagation variations
- Receiver errors

These errors directly affect the range measurements from each visible satellite. User position is determined by the processing of independent range measurements.

Two levels of accuracy are provided by NAVSTAR GPS, corresponding to two different signal codes -- the precision code (P-code) and the less accurate coarse/acquisition (C/A) code. The range error budget for GPS, summarized in Table 4-6, reflects error magnitudes relevant to use of the C/A code only, since that is the only code to be made available for civil aviation applications. Position error is a function of the combined effects of the errors arising from each range measurement.

Although the results shown in Table 4-6 are of interest, they are not necessarily indicative of the position fix accuracy that a civilian GPS user can expect when GPS becomes fully operational. The denial of accuracy that may be imposed by the Department of Defense will result in a degraded capability of determining position, possibly to a 2-drms accuracy of no better than 500 meters (0.27 nm).

The individual range error sources are discussed in the following sections.

4.5.1 Satellite Errors

The satellites are the source of the GPS navigation signals. Range from a satellite is determined on the basis of time measurements. The satellite clock provides the standard against which time intervals are measured. Satellite position is computed from the ephemeris parameters transmitted in the GPS signal. Perturbations in the orbit of a satellite will cause a position deviation relative to the ephemeris data. The effect on position accuracy of errors associated with the satellites is described in the following sections.

4.5.1.1 Clock Error

The rubidium clocks currently used in the satellites are periodically updated by the ground control facility to maintain an accuracy better than 1 meter (10).

Table 4-6. RANGE ERROR BUDGET SUMMARY FOR UNRESTRICTED GPS (C/A CODE)		
Source of Error	Effect of Error (in Neters) on Navigation System Use	
Satellite Errors		
Clock Ephemeris Orbital perturbation	1,0 1.5 1.5	
Signal Propagation Variations		
Atmospheric delay Multipath	3 20	
Receiver Errors		
Measurement noise Range quantization Navigation algorithm	10.5 2.66 1.0	
Total System Range Error (RSS)		
lo 2 drms	23.1 (0.01 nm) 46.2 (0.025 nm)	

4.5.1.2 Ephemeris Error

T

I

1

 \prod

11

Each satellite transmits its position to a user receiver in the form of predicted ephemeris data. Errors in the ephemeris translate into position fix errors. Techniques for charting the ephemeris are believed to be accurate to within 1.5 meters (1σ) .

4.5.1.3 Orbital Perturbation Error

Influences such as variations in the earth's gravity gradient cause space vehicle perturbations. Such perturbations are projected to be 1.5 meters (10).

4.5.2 Signal Propagation Variations

The GPS signal propagates from the satellite to the user through several mediums, including a vacuum and an anisotrophic atmosphere. Consequently, the propagation velocity varies as the signal passes through the ionosphere and troposphere and may experience multipath effects from reflection and refraction at various boundary layers and from objects near the receiver. The uncertainty inherent in the modeling of these propagation characteristics contributes to the range error budget for GPS.

4.5.2.1 Atmospheric Delay Error

The time delay of the GPS signal passing through the ionosphere can be compensated for by one of two techniques. The first and most accurate means of calculating the delay involves comparing two frequencies of signal transmission, exploiting the fact that the overall delay is nearly inversely proportional to the square of the frequency. When dual-frequency measurements are not available, as is the case for the C/A signal, a second technique, modeling, provides the only means of compensation. The magnitude of the error depends on time of day, solar activity, geomagnetic latitude, and other factors, which result in uncertainties of 1 to 30 meters. On the basis of projections of recent studies, an average value of 3 meters seems reasonable.

4.5.2.2 <u>Multipath Error</u>

The effects of multipath cannot be characterized through modeling, because the creation of multipath signals depends on the nature and location of reflective objects in relation to the receiver antenna. The magnitude of multipath errors has been estimated to be between 1.2 and 2.7 meters (10) for P-code operation. The degree of signal interference due to multipath effects is inversely proportional to the code rate used. Since the code rate of the C/A signal is one-tenth that of the P-code, the multipath errors corresponding to the C/A code would be between 12 and 27 meters. For this study, an average magnitude of 20 meters is used.

4.5.3 Receiver Errors

and the second second

In processing the received signal, the GPS receiver contributes to position fix error through suboptimal code lock-on caused by noise in the signal, internal accuracy limitations resulting from quantization effects, and inaccuracies inherent in the navigation solution.

4.5.3.1 Measurement Noise Error

The ability to determine GPS range accurately from a signal measurement depends on both the modulation of the selected code and the signal quality. In a signal environment characterized by a carrier-to-noise density ratio (C/N_O) of 30 db-Hz, a code measurement error of 10.5 meters (10) has been projected for C/A code operation. This level of accuracy degrades rapidly as the C/N_O worsens (approximately 22 meters at C/N_O = 25 dB-Hz), imposing a requirement for external inputs when the C/N_O falls below 20 dB-Hz. The addition of an external input, such as velocity, to the GPS position measurement process does not improve the accuracy of the GPS signals, but enhances the ability of the navigation algorithm to determine position.

4.5.3.2 Range Quantization Error

The user receiver generates replicas of the C/A codes and crosscorrelates these locally generated signals with the signals received from the satellites. A tracking loop is used to establish synchronization

between the two signals. The code tracking loop establishes maximum correlation between the received signal code and the internally generated reference code, defining the quantisation or the receiver range measurement. Code tracking errors directly affect the signal phase measurements, which are resolved into a position fix. Current designs for code tracking loops provide a code resolution capability of 1 to 1.6 percent. Each bit in the C/A code is 978 nanoseconds long. At the speed of light, the duration of the code bit, or chip width, corresponds to 293.2 meters. A resolution capability of 1.6 percent results in a range quantization of 4.6 meters. A quantization error of 2.66 meters (10) is specified when uniform distribution of the error is assumed over the range quantization value.

4.5.3.3 Navigation Algorithm Error

Implementation of a navigation algorithm contributes some error, because of computer limitations, mathematical approximations, algorithm uncertainties, and timing delays inherent in the sequential nature of the computations. The magnitude of this error is estimated to be about 1 meter.

4.5.4 Total System Range Error

I

The effects of the individual range error sources are combined in RSS fashion to compute the total system range error (1σ) . The corresponding 2-drms value for total error is approximated as the 10-error value multiplied by a factor of 2.

4.5.5 Operational Considerations

In addition to the errors contributed by the various system components, two other factors that are not error sources in the conventional sense strongly influence the overall accuracy of GPS. These two factors, geometric dilution of precision (GDOP) and denial of accuracy, are included as operational considerations because they are not limitations in the same technical sense as the factors discussed in the previous sections.

4.5.5.1 GDOP

The concept of GDOP was initially developed in connection with Loran navigation, used in characterizing other radionavigation systems, and then extended to GPS. As applied to GPS, the value of GDOP is a composite measure that reflects the influence of satellite and user geometry on the accuracy of the navigation position fix.

The following parameters are contained in the GDOP composite:

- HDOP Horizontal dilution of precision (two dimensions)
- VDOP Vertical dilution of precision
- TDOP Time dilution of precision
- PDOP Position dilution of precision (three dimensions), or $\sqrt{(HDOP)^2 + (VDOP)^2}$

 $GDOP - \sqrt{(PDOP)^2 + (TDOP)^2}$

Extensive analyses have been conducted to determine values of these GDOP parameters corresponding to various satellite geometries throughout the world. Unfortunately, most of the results published to date have been based on a constellation of 24 satellites rather than the configuration of 6 orbital planes and 18 satellites currently proposed. The rms values* for the GDOP parameters that were determined for the 24-satellite system can be thought of as minimum values for an 18-satellite system, as follows:

GDOP	Parameter	RMS Value
	PDOP	2.60
	HDOP	1.45
	TDOP	1.20
	VDOP	2.20
	GDOP	2.90

By multiplying the total 1 σ range error (shown in Table 4-6) by the value of the GDOP parameter of interest, the magnitude of the position error can be determined. When values for PDOP or HDOP are applied to the range error, the result is a radial error (1 σ) in either three (PDOP) or two (HDOP) dimensions.

4.5.5.2 Denial of Accuracy

GPS is capable of providing extremely accurate global positioning of about 50 meters (2 drms). The possible impact on national security in allowing unrestricted access to the accuracy of GPS has prompted extensive discussion of methods for denial of accuracy. The method that will most likely be used will deny the signal accuracy by altering the ephemeris and clock correction terms in the satellite navigation message to create range errors of an order of magnitude not yet agreed upon. It is believed, however, that the level of accuracy to be made available to nonmilitary users of GPS will be somewhat better than 500 meters (2 drms); the actual level will be based upon national security considerations.

4.6 ERROR BUDGET FOR DOPPLER

The error budget for Doppler navigation systems is divisible into five basic categories as follows:

- Initialization errors
- Calibration errors
- Doppler signal errors

*Derived from "Principle of Operation of NAVSTAR and System Characteristics," Reference 3. Instrumentation errors

T

Heading reference errors

Table 4-7 presents a summary of the effect of specific sources of error on the navigation accuracy of a Doppler navigation system. The various sources of error generally affect cross-track and along-track error independently. Therefore, the error budget is separated into cross-track and along-track components to illustrate the specific effects of the error sources.

Table 4-7. ERROR BUDGET FOR DOPPLER NAVIGATION SYSTEMS		
Error Source	Effect on Navigation Accuracy (Percentage of Distance Traveled from Origin Unless Otherwise Indicated)	
	Cross-Track Distance Error	Along-Track Distance Error
Initialization Errors		
Course determination	1.75	
Course selection	0.436	
Distance setting		0.500 nm
Departure point		0.049 nm
Calibration Errors Scale factor Beam direction	Negligible 0.058	0.100 0.128
Doppler Signal Errors		
Fluctuation	~~	0.146
Altitude holes		Negligible
Over-water effects		1.05
Instrumentation Errors	0.175	0.040
Heading Reference Errors	2.62	
Total Error (RSS 20)	±3.18% of distance traveled	0.502 nm ±1.07% of distance traveled
Error for 300-nm leg	9.54 nm	3.25 nm

As the table shows, error in the heading reference system input to the Doppler navigation system is the chief contributor to the overall navigation accuracy, affecting mainly the cross-track accuracy. Since heading reference error is an angular error, the corresponding cross-track distance error accumulates as a linear function of the distance traveled from the origin. The along-track distance errors are generally a result of errors in ground velocity as determined by the Doppler radar. Ground velocity errors,

which are caused by Doppler signal errors and calibration errors, are generally fractional errors rather than bias errors. A given percentage error in ground velocity results in a corresponding error in along-track position that accumulates as a linear function of the distance traveled from the origin. Both along-track and cross-track errors can be conveniently expressed as a percentage of distance traveled.

4.6.1 Initialization Errors

Commercial Doppler navigation systems generally establish a flight path by selecting a desired constant magnetic course and distance between an initial position and the next waypoint location. Since current position is determined by reference to an initial position, any errors in establishing the initial position result in position errors during flight.

4.6.1.1 Course Determination Error

The constant magnetic course computed for input to the Doppler navigation system in defining a flight segment depends on knowledge of magnetic variations. A representative accuracy for values of magnetic variation is ± 1.0 degree. This corresponds to a cross-track error of 1.75 percent of distance traveled.

4.6.1.2 Course Selection Error

Any error in the mechanical tuning of the desired course (ε_{cs}) causes a cross-track distance error (σ_{cs}) that increases with the distance traveled from the initial position, as follows:

$$\sigma_{c_{e}} = \text{distance traveled} \times \sin \varepsilon_{c_{e}}$$

A typical Doppler navigation system provides 1/2-degree increments in course selection. The resulting mechanical tuning error is $\pm 1/4$ degree and the associated cross-track distance error is the following:

 $\sigma_{cs} = .436$ × distance traveled

4.6.1.3 Distance Setting Error

The resolution in the distance selector is 1 nm. The resulting error in along-track distance is ± 0.5 nm, which does not increase as a function of distance traveled. By carefully selecting waypoints to be spaced at intervals of whole nautical miles, this error can be significantly reduced.

4.6.1.4 Departure Point Error

Because of the method often used in minimizing the effect of altitude holes (Section 4.6.3.2), the Doppler radar receiver cannot track the reflected signal below an altitude above ground level of about 50 feet. The departure point is therefore not known as accurately as it could be if

the radar receiver tracked the reflected signal while on the ground. A published estimate of the position error is 300 feet or 0.049 nm. This error does not increase with distance traveled.

4.6.2 Calibration Error

The Doppler radar is calibrated in two respects:

- Scale factor
- Beam direction

Any errors in the calibration result in position determination errors.

4.6.2.1 Scale Factor Error

Calibration of the relationship between the measured Doppler shift and the corresponding velocity to be computed involves the determination of a scale factor expressed in units of Hertz per knot. The accuracy of the scale factor is a function of the frequency stability of the Doppler transmitter. The velocity error corresponding to a typical scale factor accuracy is .1 percent of the actual velocity. This corresponds to an along-track position error of .1 percent of the distance traveled from the initial position. The cross-track error caused by scale factor error is negligible.

4.6.2.2 Beam Direction Error

The accuracy with which the directions of the radar beams are known affects the accuracy with which the measured velocity components can be resolved into the local coordinate frame. The resulting velocity error then affects the position determination accuracy, which accumulates with distance traveled. Errors in the knowledge of beam direction result from antenna installation alignment error and temperature effects in certain antenna types. Typical beam direction errors result in along-track position errors of 0.128 percent and cross-track position error of 0.058 percent of the distance traveled from the initial position.

4.6.3 Doppler Signal Errors

Several conditions affect the characteristics of the reflected Doppler signal. The following three conditions can significantly affect the navigation accuracy of the Doppler navigation system:

- Fluctuation error
- Altitude holes
- Over-water effects

4.6.3.1 Fluctuation Error

The frequency spectrum of the reflected Doppler radar signal has a random noise-like distribution of frequency components. The mean frequency in the received signal is estimated and compared with the transmitted frequency in order to determine the Doppler shift. The randomness of the return signal results in a position determination error that accumulates as a function of the square root of distance from the origin. To represent the worst-case conditions, this error is generally expressed as a linear function of distance traveled from the origin. A typical 20 along-track error due to fluctuation error is .146 percent of the distance traveled from the origin.

4.6.3.2 Altitude Holes

A complete blanking of the received radar signal can occur at altitudes at which the round-trip signal time is a multiple of the half-wave period of the FM continuous wave signal. In a Doppler system using pulsed transmission techniques, altitude holes occur where the round-trip signal time is a multiple of the pulse repetition period. This momentary loss in signal can lead to erroneous navigation data. However, this problem has been eliminated through the use of techniques that alter the modulation frequencies at selected altitudes. Therefore, in many currently used Doppler navigation systems, altitude holes do not produce significant position errors.

4.6.3.3 Over-Water Effects

Source of errors when traveling over water include the bulk motion of the water surface caused by currents, particle motion created by wind (sea spray), and sea shift. Bulk water motion and sea spray effects occur randomly; therefore, the resulting errors are not listed in Table 4-7 as generally predictable contributions to the error budget. Sea shift is related to the varying value of the scattering coefficient over calm water.

The scattering coefficient defines the amount of the transmitted radar signal that is reflected back to the Doppler receiver and is related to the angle of incidence of the transmitted signal on the reflecting surface. Since the angle of incidence varies slightly across the beam width, the scattering coefficient is also varied across the beam width. Over land, this effect is negligible, because the scattering coefficient is nearly constant for a wide range of incidence angles. However, the scattering coefficient over calm water is very sensitive to the angle of incidence. This tends to distort the shape of the frequency spectrum distribution envelope in the received Doppler radar signal. The mean frequency and the corresponding Doppler shift are therefore erroneously offset. The resulting computed velocity error can cause a position error as great as 5 percent of the distance traveled from the origin in very smooth sea conditions.

To reduce this effect, some designs compensate the velocity offset with a value preselected to represent the most likely sea conditions. When smooth sea conditions exist, this compensation is manually selected with a land/sea switch. A residual sea shift error using this method is typically between 0.6 and 1.5 percent of the distance traveled from the origin. An average value of 1.05 percent is shown in Table 4-7.

Another method, called lobe-switching, has been used in military systems to achieve position errors as small as .056 percent. This method, however, is not widely used in civilian applications.

4.6.4 Instrumentation Errors

Instrumentation errors are a result of data conversion and processing techniques. Along-track distance errors are a result of quantization and processing errors in ground velocity measurements; and cross-track distance errors are due to quantization and processing errors in drift angle determination. Typical instrumentation errors cause groundspeed errors of about .04 percent of the total groundspeed. This corresponds to an along-track position error of .04 percent of the distance traveled from the origin. Typical drift-angle instrumentation errors are about 6 minutes of arc. This corresponds to a cross-track error of about .175 percent of the distance traveled from the origin.

4.6.5 Heading Reference Errors

The accuracy of magnetic compasses used with Doppler navigation systems can vary significantly. A representative value for magnetic compass error is ± 1.5 degrees. This corresponds to a cross-track position error of 2.62 percent of the distance traveled.

4.6.6 Total System Error

Assuming that the errors from the various sources discussed are normally distributed and independent, the error components are combined in RSS fashion for both along-track and cross-track total errors. These errors are expressed as a percentage of the total along-track distance traveled from the origin. For example, the accumulated position error for a 300-nm leg is shown in Table 4-7 to be 9.54 nm for cross-track and 3.25 nm for along-track.

4.7 ERROR BUDGET FOR CONVENTIONAL INS

The error budget for conventional INS can be divided into five categories, as follows:

- Initialization Sensor mounting
- Gyros Gravity anomaly
- Accelerometers

T

Table 4-8 summarizes the effect of the various sources of error on conventional INS navigation accuracy. All error components shown in Table 4-8 represent 20 values. The main source of navigation error is related to gyro errors. Other significant sources of error are uncertainty in orthogonality of the sensor axes due to mounting errors, and initial platform azimuth error, which is due to gyro drift error during the initial alignment mode.

Table 4-8. ERROR BUDGET FO	OR CONVENTIONAL INS
Source of Error	Effect of Error on Navigation System Use (nm/hr)
Initialization Error	
Position entry Platform attitude Platform azimuth	Negligible 0.04 0.75
Gyro Errors	
Mass imbalance Bias drift Torquer scale factor	0.70 0.54 0.54
Accelerometer Error	
Random bias Scale factor	0.09 0.08
Sensor Mounting Orthogonality	0.96
Gravity Anomalies	0.10
Total Position Error (RSS 2 drms)	1.61

There are various types of errors commonly associated with inertial navigation systems:

- Bias errors
- Time-dependent errors
 - •• Linear
 - •• Oscillatory
 - ••• 84.4 minute (Schuler)
 - ••• 24 hour

Since the gyros and accelerometers are mounted on a platform continuously oriented with respect to true north, some of the error effects due to sensor uncertainties are related to the direction of travel. The error effects cited in the following discussion are typical values for a west-toeast flight over a transatlantic route. However, most manufacturers of conventional INS consider the relationship between the direction of travel and the overall position error to be insignificant.

The effects of some of the error sources are sensitive to aircraft maneuvers. For example, the error effects of gyro mass imbalance, nonorthogonality in gyro and accelerimeter mounting, and accelerometer nonlinearity are functions of changes in aircraft altitude and velocity. The values in Table 4-8 are typical error effects for a flight path that includes typical civil aircraft maneuvers. The values were derived by using a statistical model of the relationship between sensor errors and their effect on position accuracy. The model used was chosen solely on the basis of availability of data and is not necessarily the most representative of actual INS performance. This model represented all errors as being linearly time-dependent, which is why the values in Table 4-8 are expressed in nautical miles per hour (nm/hr). The model assumes that the propagation of those errors which cannot be characterized as being linearly time-dependent can be bounded by a rate of degradation of position accuracy, which can also be expressed in nm/hr.

n aga an a'

ายี่

「「「「「「「」」」」

4.7.1 Initialization Errors

Before the system can be used for navigation, it must be initialized. The initial alignment mode, referred to as initialization, comprises the following three actions:

- Initial position is entered.
- Platform attitude is aligned with the local horizontal plane (perpendicular to the gravity vector).
- Platform azimuth is referenced to true north.

Initialization errors are due to uncertainties in each of these processes. Whereas the initial position is entered manually into the system, the platform attitude and azimuth are aligned automatically by measuring gravity with the accelerometers and earth rotation with the gyros. Uncertainties in both attitude and azimuth are mainly due to gyro drift. A gyro drift rate of .003 degrees per hour (rms) is typical for systems in current use and is used in the following discussions.

4.7.1.1 Initial Position Error

Initial position is entered into the system in terms of latitude and longitude during the initial alignment mode. The resolution provided by current designs is .1 (or \pm .05) arc minutes in both latitude and longitude. This resolution corresponds to .05 nm in north position error and .04 nm (at 37° latitude) in east position error. These position errors either remain constant or decrease with time and do not represent a significant portion of the total error. These are not included in the root sum square total position error.

4.7.1.2 Platform Attitude Error

A 2-sigma error in the initial platform attitude of .2 arc minutes about both horizontal platform axes corresponds to a gyro drift rate of .003 degrees per hour (rms). This attitude error causes .04 nm/hr error.

4.7.1.3 Platform Azimuth Error

An initial platform azimuth error of 2 arc minutes, resulting from a gyro drift rate of 0.003 degrees per hour (rms) during alignment, produces a 0.75 nm/hr error during navigation.

4.7.2 Gyro Errors

Three types of gyro errors significantly affect the navigation error of a conventional inertial navigation system:

- Mass imbalance drift
- Bias drift
- Torquer scale factor instability

4.7.2.1 Mass Imbalance Drift

Mass imbalance drift is a result of an imbalance in the gyro's spinning mass and is a function of aircraft acceleration, and it is therefore related to aircraft maneuvers. Mass imbalance drift is expressed in terms of degrees/hour/g, where g is equivalent to the acceleration due to gravity. A typical mass imbalance drift of .5 degrees/hour/g causes a .70 nm/hr error.

4.7.2.2 Bias Drift

Bias drift is a result of unwanted torques introduced into the spinning mass by suspension wires, pivot friction, and back reactions of angular sensor pick-offs and torquer coils. This error takes on a random value each time the INS is turned on and also varies randomly as a function of time. Bias drift error has an impact on the initial platform azimuth accuracy (Section 4.7.1.3) and also directly affects the navigation accuracy. A bias drift of .003 degrees per hour (rms), which is typical of systems in current use, produces a .54 nm/hr error.

4.7.2.3 Torquer Scale Factor

A gyro torquer, which is a component of the gyros, is used to precess the gyro's spinning mass electrically. The resulting angular error measured by the pick-offs in the gyro is used to reposition the INS platform. The torquer is thus used in an INS to introduce earth-rotation and transportrate compensation into the orientation of the INS platform. Since this process is open-loop, any fluctuations in the torquer scale factor results in angular drift in the platform orientation. The angular drift then produces position errors. Fluctuations in scale factor of .02 percent are typical of systems in current use. This uncertainty corresponds to a .54 nm/hr error.

4.7.3 Accelerometer Errors

The most significant effects of accelerometer error can be attributed to random bias error and scale factor error.

4.7.3.1 Random Bias Error

The acceleration bias in an accelerometer takes on a random value each time the INS is turned on. The estimated mean bias is usually compensated for in either the computer or the accelerometer itself. The remaining portion of the bias produces navigation errors. An rms residual bias of $0.000025 \times g$ (g is the acceleration of gravity at sea level - approximately 32.2 feet per sec²) is typical of systems in current use. This bias corresponds to a .09 nm/hr error.

4.7.3.2 Scale Factor Error

An accelerometer scale factor fluctuation of .05 percent is typical for INSs in current use. This uncertainty produces an 0.08 nm/hr position error.

4.7.4 Sensor Mounting Orthogonality

The uncertainty in the alignment of the input axes of the gyros and accelerometers with respect to the platform due to mounting inaccuracies causes navigation errors that are sensitive to aircraft maneuvers. A typical gyro mounting alignment accuracy of 100 arc seconds and a typical accelerometer alignment accuracy of 120 arc seconds produce a .96 nm/hr position error.

4.7.5 Gravity Anomalies

It has been suggested that the uncertainty in knowledge of the earth's gravitational field is the fundamental accuracy limitation in inertial navigation systems. The error associated with anomalies in gravity has been estimated to contribute a position error of .1 nm/hr. If all other instrument and mounting errors were significantly reduced through engineering advances, the error due to gravity anomalies would be the main source of error. In fact, accuracies of this order have been demonstrated by recently developed military inertial navigation systems utilizing highly accurate gyros and accelerometers.

4.7.6 Total System Error

The individual error components are combined in RSS fashion to determine the total position error. Since all the error components shown in Table 4-8 represent 2-sigma values, the total position error also represents the 2-sigma value.

4.8 ERROR BUDGET FOR STRAPDOWN INS

The error budget for a strapdown INS can be divided into four categories, as follows:

- Gyro errors Sensor axis mounting misalignment
- Accelerometer errors
 Gravity anomaly

Table 4-9 summarizes the effect of the various error sources on the navigation accuracy of the ring laser gyro (RLG) strapdown INS under current development. The main sources of error are related to the RLG characteristics. The various types of errors listed in Section 4.7 for conventional INS are also applicable to strapdown INS. As was the case for conventional INS, all strapdown INS errors are represented in terms of linear timedependency as a function of the error propagation model used to relate the effect of sensor error to position accuracy.

Table 4-9. ERROR BUDGET FOR STRAPDOWN RING LASER GYRO INS		
Source of Error	Effect on Navigation Accuracy (nm/hr)	
Gyro Errors		
Bias drift Scale factor Random walk	1.39 0.06 0.67	
Accelerometer		
Random bias Scale factor	0.05 0.01	
Sensor Mounting Orthogonality	0.10	
Gravity Anomaly	0.10	
Total System Error (RSS 2 drms)	1.55	

In strapdown INS, the errors due to gyro bias and accelerometer bias are most pronounced when the aircraft heading during flight is different from the aircraft heading during initial alignment. During initial alignment, these bias errors, which are related to the inertial space orientation of the gyros and the accelerometers, are partially counteracted by biases in computed aircraft heading and attitude. Since the orientation of the gyros and accelerometers changes relative to inertial space as a function of aircraft heading, the bias errors do not remain entirely compensated for when the aircraft heading is changed following initial alignment. The worstcase situation is a 180° difference between aircraft heading during alignment and heading during flight. The error values shown in Table 4-9 include the effects of this worst-case situation.

4.8.1 Gyro Errors

Three major sources of error in the RLGs used in current strapdown INSs are the following:

- Bias drift
- Scale factor instability
- Random walk

4.8.1.1 Bias Drift

Bias drift in an RLG is caused by a circulating flow of gas within the laser cavity, a result of the direct current used to excite the lasing action. The gas flow alters the refraction properties of the laser cavity, producing a bias in the measured angular rate. The angular rate bias causes an angular measurement error to accumulate as a function of time. Current RLGs have a typical bias drift of .01 degrees per hour (rms). This causes a 1.39 nm/hr position error in the strapdown INS.

4.8.1.2 Scale Factor

The scale factor in an RLG defines the relationship between the measured frequency difference of the two laser beams and the corresponding angular rate. Any instability in the scale factor produces position error in the strapdown INS. A scale factor stability of 5 parts per million is typical for current RLGs. This uncertainty causes a position error of .06 nm/hr.

4.8.1.3 Random Walk

The mechanical dithering used to prevent lock-in (discussed in Appendix A, Section 8.3.1) causes an angular random walk error that is expressed as angular error per square root of elapsed time. Current RLGs have a typical random walk error of $.003^{\circ}/\sqrt{hour}$ (rms), which causes a .67 nm/hr position error.

4.8.2 Accelerometer Errors

Inaccuracies in current accelerometers do not produce significant errors in strapdown INSs. The main sources of error in the accelerometers are random bias error and scale factor instability. For a typical rms random bias accuracy of $0.00005 \times g$ (g = acceleration of gravity), a position error of only .05 nm/hr results. A typical scale factor stability of 50 parts per million (rms) results in a .01 nm/hr position error.

4.8.3 Sensor Mounting Orthogonality

The gyros and accelerometers in a strapdown INS are typically mounted with an alignment accuracy of 10 arc seconds. The resulting position error is 0.1 nm/hr.
4.8.4 Gravity Anomaly

As described in Section 4.7.5, uncertainties in knowledge of the earth's gravitational field represent the fundamental accuracy limitation in INSs. This uncertainty has been estimated to produce a position error in INSs of 0.1 nm/hr.

4.8.5 Total System Error

The error effects of the individual sources in the RLG strapdown INS are combined in RSS fashion to determine the total position error (20). The total position errors listed for conventional and strapdown INS are indicative of accuracies achieved by commercially available units and do not represent the total spectrum of capabilities when including military developments. Whereas the accuracy of 1.61 nm/hr shown for conventional INS is representative of civil units, military designs have demonstrated accuracies of .08 nm/hr. This achievable accuracy for conventional INS is superior to the accuracies achieved by strapdown INS for both civil (1.55 nm/hr) and military (1.1 nm/hr) designs.

4.9 SUMMARY OF TECHNICAL CAPABILITIES

Tables 4-10 through 4-12 compare the technical capabilities of Loran-C, Omega/VLF, GPS, Doppler, conventional INS, and strapdown INS navigation systems with the requirements defined in FAA AC 90-45A. The nonprecision approach requirements specified in AC 90-45A are referenced to typical approach clearance zones. The FRP stipulates a nonprecision approach requirement of 100 meters, which is sufficient to allow currently available nonprecision approach capability. Since AC 90-45A is the currently acknowledged reference for certification of area navigation systems, it, rather than the FRP, will be used to provide the basis for comparing system capabilities. However, any navigation system considered as a replacement for VOR/DME as the national standard would have to demonstrate performance that equals or exceeds the performance of VOR/DME in all situations.

Flight technical error (FTE) refers to the accuracy with which the pilot controls the aircraft, as measured by his success in matching the indicated aircraft position with the indicated command or desired position on the display. (With autopilot coupling, FTE more appropriately refers to autopilot error and is typically lower than pilot-flown FTE.) FTE does not include blunder errors, which are gross errors in judgment or lapses in attentiveness that cause the pilot to stray far from his desired path.

The VOR/DME navigation system is not shown in the comparison, because the requirements were based on the capability of the VOR/DME system. Thus, that system, by definition, satisfies the requirements. The three levels of GPS capability shown in the table correspond to two possible levels of accuracy denial -- 300 meters or 500 meters -- and a nondegraded accuracy of 50 meters. All entries in the tables are 20 values.

Type of Accuracy	En-Route	Lenen-C	Cmega/		gps.		Connier	Conventional	Strapdown
	Requirementst	Noren-y	VLPTT	50m	300m	500m		INS	INS
Equipment									
Cross-track	1.50	0.25	1.60	0.03	0.16	0.27	9.54*	1.61**	1.55**
Along-track	1.50	0.25	1.60	0.03	0.16	0.27	3.25*	1.61**	1.55**
Flight Technical							ļ !		
Cross-track	2.00	2.00	2.00	2.00	2.00	2.00	2.00	2.00	2.00
Along-track									
Total System		1		.			! !		1
Cross-track	2.50	2.02	2.56	2.00	2.01	2.02	9.75	2.56	2.53
Along-track	1.50	0.25	1.60	0.03	0.16	0.27	3.25	1.61	1.55

Type of	Terminal Requirements†	Loran-C	Omega/ VLFtt	gps				Conventional	Strapdown
Accuracy				50m	300m	500m	Pobler	ins	ins
Equipment									
Cross-track	1.12	0.25	1.60	0.03	0.16	0.27	9.54*	1.61**	1.55**
Along-track	1.10	0.25	1.60	0.13	0.16	0.27	3.25*	1.61**	1.55**
Flight Technical									
Cross-track	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00
Along-track									
Total System									
Cross-track	1.50	1.03	1.89	1.00	1.01	1.04	9.59	1.90	1.84
Along-track	1.10	0.25	1.60	0.03	0.16	0.27	3.25	1.61	1.55

ttRanging, not hyperbolic, implementation.

E

E

As shown in the tables, Loran-C systems are capable of satisying enroute, terminal, and approach accuracy requirements, assuming availability of a signal. The equipment accuracy of .25 nm specified for Loran-C, which corresponds to a GDOP of 3.6, was chosen on the basis of reported operational experience. The value of 1.6 nm specified for Omega/VLF accuracy is representative of rho-rho Omega/VLF equipment accuracy, corresponding to a GDOP of 1 rather than 2, which is typical of hyperbolic Omega. Omega/ VLF is clearly unable to meet terminal and approach requirements and falls

Type of Accuracy		tomaned	Omega/	gps			Doppley	Queventional	Strapdow
	1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.		VLPTT	50m	100m	500m			INS
Squipment			in the second	,					
Cross-track Along-track	0.33	0.25	1.60	0.03	0.16	0.27	9.54*	1.61**	1.55**
Flight Technical	e e Altonia N			din a din d Lin A		н. н. — . К	ال معند ا معرف		
Cross-track Along-track	0.50	0.30	0.50	0.50	050	0.50	0.50	0.50 () 	0.30
Total System	an a						in to si and to s	4 \sim 2	ь - Т
Cross-track Along-track	0.60	0.56	1.60	0.50	0.52	0.57	9.55 3.25	1.69 1.61	1.63

slightly short of the en-route accuracy requirements. However, measured accuracy during flight has shown that Omega/VIF navigation systems are capable of an accuracy of ±1.5 nm. Even with signal accuracy degraded to ±500 meters, GPS can provide nonprecision approach capability at some sites, although this capability is marginal and does not take into consideration the effect on accuracy if fewer than four satellites are visible. Dopplar is unable to meat domestic en-route, terminal, and approach requirements. Both conventional INS and strapdown INS are unable to meet terminal and approach requirements, but can meet domestic en-route requirements for up to one hour in flight.

Table 4-13 summarizes the capabilities of the various navigation systems relative to operational environments. Suitability, as defined in the context of the table, reflects a possible rather than a certified capability to meet existing requirements. The capabilities of a particular system operating in a certain environment are judged on the basis of existing reference material. In addition, the suitability of a particular system to provide domestic navigation capability does not imply an ability to replace VOR/DME as the national navigation standard. Any navigation system considered as a replacement for VOR/DME must not only provide coverage and accuracy at the level that currently exists, but also demonstrate some degree of improvement, economic as well as operational.

Although Table 4-10 indicates that Loran-C is sufficiently accurate to meet en-route technical requirements, Loran-C chains do not provide oceanic coverage, as shown in Table 4-13. Also, there is currently no midwest Loran-C chain in the domestic United States, a situation that limits signal availability. The suitability of Loran-C for terminal and nonprecision approach navigation is contingent upon signal availability. Omega/VLF provides worldwide coverage with an accuracy suitable for enroute navigation but not sufficient for terminal or approach operations.

		Operational Environment										
Navigation	1	En Route										
bye cem	Domestic	Trans- Oceanic	Terminal	Approach								
Loran-C	Suitable ¹	No Coverage	Suitable	Suitable ¹	Suitable ¹							
Omega/VLF	Suitable	Suitable	Suitable ²	Insufficient Accuracy	Insufficient Accuracy							
GPS	Suitable	Suitable	Suitable	Suitable	Suitable ²							
VOR/DME	Suitable	No Coverage	No Coverage	Suitable	Suitable ¹							
Doppler	Insufficient Accuracy	Suitable ⁴	Insuffi- cient Accuracy	Insufficient Accuracy	Insufficient Accuracy							
INS	Suitable ³	Suitable	Suitable ²	Insufficient Accuracy	Insufficient Acouracy							

Table 4-13. SUMMARY OF NAVIGATION SYSTEMS' ABILITY TO PROVIDE ACCURACY AND COVERAGE

⁴Under conditions allowing attainment of sufficient accuracy. ³For flights of less than one hour. ⁴For ocean tracks less than 400 nm.

Accuracy requirements are specified in FAA AC-90-45A.

GPS, even with a denial of accuracy to a level of 500 meters, would provide sufficient accuracy for en-route, terminal, and some nonprecision approach navigation, given the continuous availability of four satellites. The FAA maintains that five satellites are required, each above a 10 degree masking angle, in order to provide sufficient reliability and to allow economical antenna design. Redundancy of sources of navigation signals is an important consideration when evaluating the potential of a navigation system to adequately replace the VOR/DME system and the level of redundancy which it provides. The limitation of the application of VOR/DME is the result of lack of coverage in oceanic, remote, and offshore areas and the consequent cost of maintaining a large number of transmitters for covered areas. Since Doppler and INS are self-contained navigation systems, they provide worldwide navigation capability. Doppler accuracy is suitable for oceanic en-route navigation for distances up to 400 nm. Doppler accuracy is inadequate for all domestic and offshore operations. Accuracies of both conventional INS and strapdown INS are adequate for oceanic en-route navigation for as long as about 8 hours in flight, and for domestic en-route navigation for up to 1 hour in flight. INS accuracy is not sufficient, however, for terminal or approach operations.

CHAPTER FIVE

OPERATIONAL EVALUATION OF NAVIGATION SYSTEMS

5.1 INTEROPERABILITY CONSIDERATIONS

Interoperability of navigation systems can be defined as the ability of systems to operate effectively in an environment where dissimilar systems are operating concurrently. The degree of compatibility among systems provides a measure of their interoperability. The following factors influence compatibility:

- · Coordinate systems
- Route structures
- Separation standards
- RF environment

- Equipment interface
- Flight procedures

Each of these is discussed in the following sections.

5.1.1 Coordinate Systems

All airborne navigation systems measure aircraft position in relation to a particular coordinate system. Basic VOR/DME employs a polar coordinate reference system with the VORTAC as the origin. The resulting distance and bearing (rho/theta) measurements define aircraft position in relation to the VORTAC. Knowledge of the geographical position coordinates of the VORTAC helps determine the geographical position of the aircraft. Geographically oriented navigation systems such as Omega, Loran-C, and GPS are referenced directly to latitude and longitude.

Any process for determining position on the earth's surface involves an assumption of the shape of the earth. One of the following three basic approximations to the earth's shape can be used:

- Flat
- Spherical
- Ellipsoidal

Through these approximations, the coordinate system used in the airborne navigation processor is defined.

The coordinate system accurately defines the surface of the earth to provide the capability to navigate between points on the earth's surface. Unfortunately, the shape of the earth is irregular and cannot be easily described mathematically. The geometrical figure that most closely approximates the shape of the earth is an oblate spheroid, or ellipsoid of revolution, created by an ellipse rotating about its minor axis. The irregularity of the earth's surface, however, prevents any one ellipsoid from approximating more than a particular section of the surface. Because of this restriction, a number of reference ellipsoids have been defined, each providing a fit only to localized areas. The geodetic and geophysical parameters used to define a reference ellipsoid are referred to as a datum. The datum origin is generally the point at which the reference ellipsoid is tangent to the earth geoid (the surface of the earth coinciding with mean sea level).

Navigation errors can be introduced when the navigation process involves different datums. The earth model used by the airborne navigation computer is based on one particular datum, such as the North American Datum. Charted locations of navaids, landmarks, airports, and other land sites may or may not be defined with respect to that same datum. The magnitude of differences between datums is primarily a function of the distance between datum origins. In addition, the accuracy within a given datum decreases as one travels progressively farther from the datum origin. Thus the error associated with nonstandardization of a coordinate system is not a constant bias, but depends on the actual ellipsoids used and the location of the user with respect to the datum origins of the ellipsoids. As an example, differences between the coordinates defined in the Tokyo Datum and a center-of-mass-referenced datum defining the World Geodetic System (WGS-72) can be as great as 500 meters (0.27 nm).

As the accuracy of navigation systems continues to improve, the effect of factors such as charting inaccuracies becomes increasingly more significant. Table 5-1 presents the effect that a charting error of 0.27 nm would have on the accuracy of a position fix determined by various navigation systems, in terms of the percentage of their respective system accuracy capability. The system accuracy of 2.5 nm listed for Omega corresponds to a GDOP of 1.6, which is considered typical for Omega navigation using hyperbolic techniques. Rather than specifying two values of Omega system accuracy, 1.6 nm for ranging implementation and 2.5 nm for hyperbolic implementation, the worst-case value of 2.5 nm will be used for the purpose of an operational evaluation.

5.1.2 Route Structures

The high cost of fuel emphasizes the need to provide for flexible routing of aircraft to maximize fuel efficiency. Direct routing between origin and destination provides the shortest distance path, but wind conditions may suggest selection of a less direct path.

Table 5-	Table 5-1. IMPACT OF CHARTING ERROR ON SYSTEM ACCURACY							
Navigation System	System Accuracy	Percentage of Accuracy Degradation Due to Charting Error of 0.27 nm						
Omega	2.5 nm	10.8						
Loran-C	0.25 nm	108.0						
RNAV VOR/DME	0.5 nm	54.0						
GPS	0.16 nm	167.0						
Doppler	9.5 nm*	2.8						
INS	1.6 nm**	16.9						
*Accumulated **Accumulated	over 300 nm during one	hour.						

Although it would be desirable to allow aircraft equipped with RNAV equipment to fly direct routes from origin to destination without having to report intermediate VOR position fixes, such freedom can create conflicts. RNAV units generally compute great-circle flight paths. A great circle is constructed under the assumption that the earth is a perfect sphere. Corrections are sometimes applied to account for the oblateness of the earth, but the spherical-earth assumption is frequently considered to be sufficiently accurate. However, for a flight between New York and Los Angeles, the differences between a spherical-earth assumption and use of the Clarke 1866 oblate spheroid model of the earth result in an eastwest discrepancy of 6.5 nm and a north-south discrepancy of 0.5 nm.*

Not only do the differences among the position-fixing processes used in airborne navigation systems present potential conflicts, but certain ATC computer capabilities can also create complications. The ATC computer can project flight paths forward in time to aid in the prediction of possible path conflicts. These extrapolations of current track are either great-circle projections or rhumb-line projections. The deviation between a great-circle path and a rhumb-line path connecting New York with Los Angeles is approximately 135 nm of lateral separation at the point of greatest divergence. This point occurs nearly midway between New York and Los Angeles. After this point, the paths begin to converge until they finally meet at the destination in Los Angeles. (It is interesting to note that there is a difference of only 1 percent in total distance traveled between the two paths.) Figure 5-1 is a comparison of the two flight

*Equations used to obtain values specified throughout this chapter are given in Appendix B.

Ľ

E





1

1.00

I

paths. The axes are, for simplicity, linear rather than representative of a Mercator or Lambert conformal projection. The computations used to determine the magnitude of greatest cross-track deviation betwee the rhumb-line and great-circle paths is presented in Appendix C.

The magnitude of deviation between a rhumb-line path and a greatcircle path is dependent on both direction and distance of travel. The shorter the distance, the less the deviation. Since the current ATC computer provides flight-path projections for only short distances, possible discrepancies between rhmub-line and great-circle paths are minimized, regardless of direction of flight.

The following calculations illustrate the application of equations (given in Appendix B) to determine the flight path latitude (ϕ) at a given value of longitude (λ) for each path. The value of longitude used as an example is 93.778 degrees. ϕ_1 equals 40.64 degrees, λ_1 equals 73.778 degrees, ϕ_2 equals 33.942 degrees, and λ_2 equals 118.407 degrees.

Rhumb Line (RL):

$$\phi_{RL} = \phi_1 + (\lambda - \lambda_1) \times \left(\frac{\phi_2 - \phi_1}{\lambda_2 - \lambda_1}\right)$$

= 40.64 + (93.778° - 73.778) ×
$$\left(\frac{33.942 - 40.64}{118.407 - 73.778}\right)$$
 = 37.64°

Great Circle (GC):

$$\phi_{\rm GC} = \tan^{-1} \left[(\tan Lv) \times \cos (\lambda - \lambda_1 - DLov) \right]$$

where

E

I.

$$Lv = \cos^{-1} (\cos \phi_1 \sin C) = 40.79^{\circ}$$

$$C = \tan^{-1} \left[\frac{\sin (\lambda_2 - \lambda_1)}{(\cos \phi_1 \tan \phi_2) - (\sin \phi_1 \cos (\lambda_2 - \lambda_1))} \right] = 86.1564^\circ$$

$$DLov = \sin^{-1} \left(\frac{\cos C}{\sin Lv} \right) = 5.8894^\circ$$
Thus $\phi_{cc} = \tan^{-1} \left[0.8629 \times \cos (\lambda - 79.6674) \right] = 39.9247^\circ$

If an aircraft is flying a great-circle route and the ATC computer provides the controller with a rhumb-line projection of that flight, a serious conflict may develop. Such a possibility could be avoided if the ATC computer were emulating the navigation techniques used by the aircraft. Unfortunately, standardization of a navigation technique does not exist. Radar surveillance is used to monitor deviations from intended routes, but the controller will not intervene unless a potential conflict is perceived.

5.1.3 Separation Standards

Separation standards have been developed historically on the basis of estimated system accuracies. The navigation system used as the reference standard in determining existing standards is VOR/DME. System accuracies are developed from estimates of the individual accuracies of the navigation signal, the receiver, and flight technical error and provide a measure of the bounds of position error.

Position error is defined in terms of the techniques used to determine position. GPS is a range system; VOR/DME is an angle and range system. Omega and Loran-C are hyperbolic navigation systems but can also be used in a ranging mode. The VOR contribution to VOR/DME position error is presented as an angle or a one-dimension distance at a defined range from the station. DME error is a function of a percentage of the distance from the station. Position error in relation to Loran-C, Omega, and GPS is an elliptical area, the dimensions of which are determined by the geometry of the aircraft position in relation to the transmitters. The problem of compatibility of accuracy specifications when different systems are compared should be resolved by adoption of the 2-drms position error probability method.

5.1.4 RF Environment

Of the radionavigation systems studied -- VOR/DME, Loran-C, Omega, and GPS -- only DME requires transmission of an RF signal by the airborne unit. There are no indications that DME signals cause interference with reception of the navigation signals of VOR, Loran-C, or Omega. However, the airborne DME interrogator may disrupt reception of GPS signals, depending on the proximity of the DME channel frequency to the frequency of the GPS signal.

Each radionavigation system is susceptible to certain types of external interference, but the question of compatibility is whether the existence of one navigation system adversely affects the use of another system. The only other known example of conflict is in the use of Loran-C when the user is in close proximity to the VLF transmitting stations used by Omega/VLF navigation systems. Test flights have indicated that reception of Loran-C signals is very poor when the user is within 15 nm of a VLF transmitter. The notch filters of airborne Loran-C units are unable to discriminate against the broad-band characteristics of the VLF signal. Doppler and INS systems are not susceptible to RF interference.

5.1.5 Equipment Interface

The purpose of a navigation system is to provide a means of establishing and maintaining a route by which one can travel from point A to point B. In most applications, display of the deviation from the prescribed route is more useful for maintaining correct course than display of actual geographical position along the route. For this reason, the course deviation indicator (CDI) has become the primary means of control display between the navigation system and the pilot. All the navigation systems studied provide capability for interface to a CDI. The navigation systems are not, however, interchangeable in terms of cockpit installation. Unique requirements of each system, such as antenna type, generally create the need for modifications to existing cockpit interface wiring when a change is made from one type of navigation system to another.

5.1.6 Flight Procedures

The question of compatibility with respect to flight procedures involves technique as well as capability. To assess adequately the operational performance of a particular navigation system, case studies were selected to represent operational environments in which the system must perform. It is not necessary to emulate all possible scenarios to provide an adequate measure of system interoperability. Interoperability is a measure of the ability of ATC to maintain safe separation of aircraft without giving special consideration to the types of navigation systems used. The methods by which ATC maintains lateral separation of aircraft are specified in the Air Traffic Control handbook as follows:*

- "Clear aircraft on different airways or routes whose widths or protected airspace do not overlap.....
- "Clear aircraft below 18,000 [feet] to proceed to and report over or hold at different geographical locations determined visually or by reference to NAVAIDs.
- "Clear aircraft to hold over different fixes whose holding pattern airspace areas do not overlap each other or other airspace to be protected.
- "Clear departing aircraft to fly specified headings which diverge by at least 45 degrees."

Degrees of effectiveness of system interoperability can best be measured through scenarios that highlight the distinctions between system capabilities in different environments. Five scenarios were devised that were considered effective in demonstrating these differences, and representative of each of the operational methods used by ATC to maintain lateral separation of aircraft. Additional scenarios were considered, but the resulting values of operational merit for respective systems duplicated those determined in a previously defined scenario. Duplication of results, regardless of differences in scenarios, was avoided so that the possibility of bias would be minimized. The scenarios were applied to a mix of navigation systems operating concurrently in common domestic en-route airspace.

The scenarios provide the basis for case studies, which are defined in terms of response to the following controller requests:

- 1. "Turn left heading zero two five."
- 2. "Cleared to fly direct Fargo."

*Air Traffic Control, Reference 31.

- 3. "Track outbound on the Green Bay VOR zero nine zero radial."
- 4. "Report crossing Jones intersection."
- 5. "Fly North Atlantic Track F."

The response is evaluated relative to the following criteria:

- · Means of establishing course or track
- Means of maintaining course or track
- Consequence of loss of signal
- Pilot workload
- Controller workload

5.2 CASE STUDIES

The following case studies are intended to provide insight into the operational effectiveness of the navigation systems by exercising those systems in realistic situations. Differences in the functional characteristics of airborne control/display units (CDUs), such as push-buttons versus knobs for data entry, must not be considered relevant when the effectiveness of types of navigation systems is compared. The computational, rather than the mechanical, effectiveness associated with use of a particular navigation system is of interest. Although the functional design of a CDU can affect pilot workload, it is not the intent of this study to determine the optimal design of a CDU. Rather, this study must establish the inherent limitations of a navigation system without regard to packaging considerations that could theoretically be accommodated by any system. The limitations identified through application of the case studies are summarized in Section 5.3.

sequent reference to INS in this report applies equally to both conventional INS and strapdown INS. Since the Doppler navigation system is shown in Chapter Four to have insufficient accuracy in domestic en-route applications, it will not be discussed in terms of its operational capability in Case Studies One through Four, which deal with domestic en-route applications.

5.2.1 Case Study One

5.2.1.1 Description

Situation

The first case study considers pilot compliance with an ATC request to "turn left heading zero two five." The aircraft is assumed to be flying at a true airspeed of 350 knots at 17,500 feet en route due east over the continental United States. Choice of aircraft airspeed and altitude is completely arbitrary and does not reflect any intentional discrimination relating to class of aircraft. The aircraft is being vectored under instrument flight rules (IFR) conditions to avoid a thunderstorm and is being monitored by radar. Wind is out of the northwest at 30 knots.

General Comments

I

adaatad tara in a

The request calls for a left turn to establish a magnetic heading of 25 degrees. When responding to an ATC vector request, navigation guidance becomes more a function of ground control and monitoring than a function of the airborne navigation unit. Since all aircraft are equipped with a magnetic compass, they can comply with a request to fly a specified heading vector without reliance on anything more sophisticated. However, since issuance of vectors is one of the means by which ATC maintains lateral separation of aircraft, any possible relationship between such a request and the type of navigation equipment used should be examined. Also, a discussion of the ability of navigation equipment to be used to respond to a heading vector request will serve to underscore the distinction between heading vectors and track vectors.

Applicable Systems

In this scenario, single VOR and non-RNAV VOR/DME systems are considered separately. The five RNAV navigation systems - VOR/DME, Loran-C, Omega, GPS, and INS - are not evaluated separately, since the response is common to all.

5.2.1.2 Application of Scenario

Single VOR

A VOR receiver indicates the magnetic bearing of the VOR transmitter with respect to the aircraft antenna. This information is not sufficient to establish a heading reference. Therefore, a magnetic compass is still required to establish heading, even when the aircraft is equipped with a VOR receiver.

Non-RNAV VOR/DME

En airborne VOR/DME system provides distance as well as bearing to a VORTAC. Addition of the distance information makes possible a relative position fix with respect to the VORTAC from which the signals emanate. Although a VOR/DME system does not provide heading information, successive manual plotting of relative position fixes provides an indication of ground track. Knowledge of ground track, however, does not indicate heading. Consequently, a magnetic compass is used to verify heading.

RNAV

All RNAV units, regardless of type, demonstrate the same degree of operational capability in responding to a request for a heading change. For this reason, the following description applies equally to RNAV VOR/DME, Loran-C, Omega/VLF, GPS, and INS.

Area navigation systems establish a ground-track reference, rather than a heading reference, by which the desired flight path is maintained. To establish a flight path, origin and destination must be entered into the RNAV unit. In many RNAV applications, the origin of a flight path is the aircraft's present position, which does not need to be manually entered into the RNAV unit. A ground track is then computed by the airborne computer, and deviations from that track are measured by processing the navigation signals received.

Destination entered into the RNAV unit can be defined in a variety of formats, one of which is bearing and distance from some location already known to the RNAV unit. If the RNAV unit has been successfully navigating before the request for a heading change, current position is a known location. In most RNAV units, current position is defined as Waypoint 0. Another waypoint, which can then be entered into the unit, is at a bearing of 025 degrees from Waypoint 0. The distance entered can be nearly anything, since the object is to establish a direction of track, not a specific destination. However, for practical reasons relating to computational overflow and underflow, the distance used should be reasonable -- 100 nm, for example. Directing the RNAV unit to initiate a direct-to computation from current position to a waypoint defined as 025 degrees bearing, 100 nm distance from current position, establishes a ground-track angle of 025 degrees as the desired track.

「日本のない」とないというというないになった。

A track of 025 degrees does not necessarily correspond to the heading of 025 degrees flown by aircraft using a magnetic compass to comply with the ATC request. The difference between track and heading is wind. Flying a heading of 025 degrees at a true airspeed of 350 knots in the presence of a 30-knot northwest wind results in a track of 029.6 degrees with a groundspeed of 341 knots, as illustrated in Figure 5-2. Therefore, to comply with a request to fly a heading of 025 degrees in this scenario, the pilot would have to input a track of 029.6 degrees when establishing his "to" waypoint. However, when a pilot is instructed to fly a heading, he is expected to fly a magnetic heading without regard to ground track.

Calculation of the required track of 029.6 degrees requires knowledge of wind speed, wind direction, and true airspeed, and application of the law of cosines. An alternative to this computation is to input the desired heading as the track angle and compensate for the wind by monitoring heading and modifying the track input accordingly. Although this alternative does not require manual calculations, it is no less demanding in terms of pilot workload because of the requirement to perform numerous operations with the RNAV unit over an extended time period.

Clearly it is difficult to use an area navigation system to fly a requested heading. Although heading may be a display parameter provided by area navigation systems, it is generally not a computational by-product of the area navigation process, but rather is the output of an external sensor. The source of heading information is a magnetic compass. It is therefore common practice for pilots to revert to the magnetic compass when responding to such requests. In aircraft equipped with area navigation systems, the magnetic compass is generally associated with a magnetic heading



Figure 5-2. RELATIONSHIP BETWEEN HEADING AND TRACK VECTORS

reference system stabilized by a gyro, referred to as a gyro-stabilized magnetic compass. The horizontal situation indicator (HSI) is a magnetic heading reference system that allows either heading selection via the magnetic compass or course selection when coupled with an area navigation system. An HSI can also be operated in a VOR-only mode, in which case course selection represents a desired VOR radial. In the area navigation mode, desired course is an input rather than an output and is displayed on the HSI as a function of the current track leg.

5.2.1.3 Discussion of Results

Ľ

[1

Relative Comparisons

None of the navigation systems evaluated are suited for efficient response to a request for a heading change, because their primary function is to compute and display track, not heading.

System Mix Conflicts

A conflict could arise if two or more aircraft were requested to fly heading vectors and each pilot programmed the heading into the respective hold modes appropriate to the navigation systems used. The navigation systems are referenced to ground track, whereas ATC requests are referenced to heading through the magnetic compass. Depending on winds and direction of flight, an aircraft holding track might not respond satisfactorily in accordance with the controller's intentions. A burden would then be placed on the controller, who would find it necessary to issue new vectors to compensate for the inconsistencies. (A pilot flying track when instructed to fly heading would be in violation of ATC regulations.)

5.2.2 Case Study Two

5.2.2.1 Description

Situation

Case Study Two considers pilot response to an ATC clearance to "fly direct Fargo." At the time of clearance the aircraft is near Green Bay, Wisconsin, 390 miles east of the Fargo, North Dakota, VORTAC. The aircraft is operating in visual meteorological conditions in daylight. The pilot has first filed a flight plan originating at Chicago, flying direct to Fargo. Because of traffic considerations, the aircraft has been vectored to Green Bay; upon arrival, direct clearance to Fargo is issued.

Applicable Systems

All systems are considered independently in this scenario -- single VOR, non-RNAV VOR/DME, RNAV VOR/DME, Loran-C, Omega/VLF, GPS and INS.

5.2.2.2 Application of Scenario

Single VOR

In this situation, the aircraft is initially out of range of the Fargo VOR signal, regardless of altitude. Therefore, a direct-to clearance cannot be accommodated and is not issued when a single VOR receiver is used and the "to" destination is beyond reception range.

Non-RNAV VOR/DME

Addition of a DME receiver is of no consequence in this scenario. Because the VORTAC defining the destination is beyond range of the VOR/DME unit, any possibility of a direct-to capability is eliminated.

RNAV Single VOR/DME

When a VOR/DME-based area navigation system is used, position offsets can be applied to appropriate VORTACs along the flight path to establish intermediate waypoints between origin and destination. The waypoints should be spaced along the flight path at distances that ensure overlapping of VORTAC signal coverage, thus making direct rounting possible between any two points. If a circular coverage area with a radius of 120 nm is assumed for each VORTAC, three VORTACs are necessary to provide continuous coverage between Green Bay and Fargo, as illustrated in Figure 5-3.

For the RNAV VOR/DME unit, all waypoints are entered in terms of bearing and distance from a particular navaid, which is defined by a frequency input. Waypoints other than origin and destination are optimally defined as points on the flight path that are directly abeam of the reference navaids. (An abeam point is defined as the intersection of the flight path and a line perpendicular to the flight path drawn to the navaid.) As the aircraft progresses along the flight path, the various waypoints defined in the stored flight plan list are sequentially activated.

As stated in the description of this scenario, the original flight plan was for a direct route between Chicago and Fargo. Appropriate navaids were manually selected from aeronautical charts to serve as waypoint references. Either manual calculations or a ground-based computer was then used to compute the magnetic bearing and distance of the navaid abeam points on the flight path. The diversion to Green Bay, which could not have been predicted, results in the necessity of manually charting a new course and defining new waypoints. If the pilot has access to an RNAV en-route chart for the area, an established RNAV route such as J976R can be used. The reference navaids and corresponding waypoint coordinates are already defined and need only to be entered into the RNAV unit. If such a chart is not available, however, or if none of the established RNAV routes are acceptable, the pilot must expend much effort to establish a new flight plan. This situation will now be considered.

The first task for the pilot is to draw a line connecting origin and destination on his aeronautical flight planning chart. A straight line on a Lambert conformal chart, the chart typically used, closely approximates a great-circle path with a constantly changing heading. If the only chart available is a Mercator projection, a straight line will define a rhumbline path, with constant true heading.

Once the path has been defined, the pilot must select navaids along the route that will provide adequate overlapping coverage. A line can then be drawn between the navaid and the flight path, optimally intersecting the flight path at a right angle. The point of intersection defines the waypoint. The magnetic bearing and distance from the navaid to the waypoint can then be approximated from the chart. On the flight path between Green Bay and Fargo, as indicated in Figure 5-3, at least two waypoints must be entered into the RNAV unit in addition to the destination (which has already been defined), so that adequate coverage is provided.

If, as the flight progresses, a navaid used to define a waypoint is not transmitting because of a sudden unannounced shutdown, a new navaid must be selected for use as soon as possible. Although this situation does not necessarily require charting a new course, it does require calculating new waypoint coordinates and entering them into the RNAV unit.

RNAV Multiple VOR/DME

The navaid data base typically included in RNAV multiple VOR/DME systems allows selection of any navaid and access to its identifying parameters (such as coordinates, frequency, elevation, and magnetic variation) through input of a three-letter navaid identifier (ident). The destination of Fargo, North Dakota, is entered simply as FAR. A waypoint list is used to enable definition of distinct flight legs. However, a directto clearance requires the input of only origin and destination. Waypoint 0 in the waypoint list is generally a reserved location that maintains current position. When a direct-to clearance is initiated, current position is designated the origin, and destination corresponds to the waypoint defining Fargo.

Selection of primary and secondary navaids to ensure continuous signal coverage throughout the flight is automatic. Station-selection algorithms first identify candidate navaids in terms of proximity to the flight path. Final selection of primary and secondary navaids is based on obtaining geometry that provides the greatest accuracy. Loss of a navaid while en route would pose no problem, since the station-selection algorithm would automatically search for and acquire a replacement.

Loran-C

Loran-C is, by definition, an area navigation system. The only inputs required are origin and destination. As with RNAV VOR/DME units, a waypoint list is used to define intermediate flight legs. Current position is always maintained and made available through a reserved location in the waypoint list. Under the assumption that a navaid data base does not exist, the destination (Fargo) must be entered in terms of latitude, longitude, and magnetic variation.

The flight from Green Bay to Fargo was selected in consideration of Omega and Loran-C coverage areas. The flight path passes through an area representing worst-case United States domestic coverage for Omega and Loran-C navigation. A midwest Loran-C chain, currently in the planning stages, would dramatically improve Loran-C coverage in the United States. The origin, Green Bay, Wisconsin, is well within the coverage area of the Loran-C Great Lakes chain. Fargo, North Dakota, however, is beyond the published coverage area of the Great Lakes chain and is also out of range of the Loran-C chain for the west coast of the United States. The published limits of Loran-C coverage are approximations based on a signal-to-noise ratio of 1:3 and a fix accuracy of 0.25 nm (95 percent, 2 drms). Therefore, as the pilot progresses toward Fargo from Green Bay, degradation in signal strength could cause loss of navigation integrity when the aircraft is 310 nm out of Green Bay, with 80 nm still remaining to Fargo.

Omega/VLF

Unlike Loran-C, Omega/VLF is considered a worldwide navigation system. As with Loran-C, however, the only inputs required for establishment of a direct-to great-circle route are specifications of origin and destination.



Source: Enroute Low Altitude - U.S. Flight Information Publication, Reference 27.

•



a stranger and the

Figure 5-3. Flight Path Between Green Bayer Misconsin, and Fargo, North Dakota

' de

Several Omega sets provide access to a data base, allowing input of waypoints in terms of alpha designators.

For this scenario, then, the pilot inputs "FAR" as the "to" waypoint and selects current position as the "from" waypoint. A great-circle route between Green Bay and Fargo is thereby established, and Omega and VLF stations within reception range are used to monitor progress along the flight path.

Seven Omega transmitters are currently operational. They are located in Norway, Argentina, La Reunion, North Dakota, Hawaii, Liberia, and Japan. Norway, La Reunion, and Argentina cannot provide coverage over the route described in this scenario. (The Australian transmitter, when declared operational, will provide coverage over this flight path only at night.) During the time of flight (daytime), Green Bay is within the coverage area of North Dakota, Hawaii, and Liberia. Liberia and Hawaii provide coverage throughout the flight (assuming they are not shut down for scheduled maintenance). The flight enters the outer fringe of the coverage area of the Japan transmitter only upon arrival at Fargo.

Published Omega coverage areas are generally based on a signal-tonoise ratio of 1:10 (-20 dB) or 1:32 (-30 dB). Omega units generally deselect use of a station that is within 300 nm of the receiver to avoid nsar-field effects. Although some units deselect stations within 600 nm, 300 nm is considered more typical. The North Dakota Omega station is only 76 nm southwest of Fargo. Therefore, the North Dakota station will be deselected when the aircraft is approximately 166 nm out of Green Bay, as illustrated in Figure 5-3.

Coverage along the flight path is provided by the following stations:

Destination	Available Omega Stations
400 nm	North Dakota, Hawaii, Liberia
300 nm	Hawaii, Liberia
200 nm	Hawaii, Liberia
100 nm	Hawaii, Liberia
0 nm	Hawaii, Liberia, Japan
0 nm	Hawaii, Liberia, Japan

Distance to

Satisfactory hyperbolic or rho-rho-rho navigation is possible when three Omega stations are being received. When only two Omega stations are available, hyperbolic navigation is not possible. Rho-rho navigation is possible with only two stations, but the accumulation of error associated with not having an independent measurement for determining user clock bias prevents extended use of this mode. These considerations make navigation using only Omega stations impossible along this flight path when the coverage charts published in Reference 21 are used as the basis for judgment.

However, use of VLF signals to augment Omega navigation provides capability not otherwise possible. Signals from several VLF transmitting stations can be received along the flight path between Green Bay and Fargo. Therefore, an Omega/VLF navigation system can provide satisfactory navigation in this case study. Addition of a VLF signal-processing capability to the Omega unit has no effect on the pilot interface with the unit.

GPS

GPS, a satellite-based navigation system, is also an area navigation system. The capabilities planned for a low-cost GPS navigation system are essentially similar to those previously described for the Loran-C receiver. It is assumed that a VOR/DME data base is not used. Origin and destination are entered in terms of latitude and longitude. A waypoint list is used to define intermediate flight legs if they are needed, and current position is available in a reserved location in the waypoint list. The route between Green Bay and Fargo is established by entering the latitude and longitude of Fargo into the waypoint list and defining that waypoint as the "to" waypoint; current position serves as the origin and is defined as the "from" waypoint. As has been assumed previously, the navigation unit in use has been operational for some time, and the received signals provide satisfactory navigation capability at the time the direct-to clearance is issued.

With an 18-satellite configuration of three satellites in six orbital planes, more than four satellites are not always in view over the continental United States above an elevation angle of 10 degrees. Four satellites are required for three-dimensional navigation by GPS; as long as four satellites remain in view, such navigation between Green Bay and Fargo is possible. After establishing origin and destination, all the pilot needs to do with the navigation system for the duration of the flight is to monitor and correct deviation from the flight path as indicated by the navigation system display. However, in a situation where fewer than four satellites are in view, the navigational capability of GPS is reduced dramatically. Even with an altimeter providing a vertical position measurement to the GPS receiver, having only three satellites in view would result in degraded position accuracy where satellite geometry, in conjunction with the altimeter measurement, creates a symmetrical arrangement resulting in poor HDOP. Without the altimeter input, user clock bias can not be estimated with only three satellites - again the result is degraded position accuracy.

INS

INS is a self-contained navigation system, which is by definition an area navigation system. Assuming that a VOR/DME data base is not used, origin and destination are entered in terms of latitude and longitude. Intermediate flight legs are defined by waypoints entered into a waypoint list. Current latitude and longitude are available in a reserved location in the waypoint list. The route between Green Bay (current position) and Fargo is established by entering the latitude and longitude of Fargo into the waypoint list and defining that waypoint as the "to" waypoint. The waypoint reserved for current position is then defined as the "from" waypoint. As the aircraft progresses from Green Bay to Fargo, the position accuracy of the INS degrades as a function of time. Figure 5-4 illustrates how this degradation of position accuracy interacts with the flight technical error to produce a total position error that exceeds the ±4 nm route width after 2.2 hours of flight. Assuming an airspeed of 200 knots or more, the flight time from the point of departure (Chicago) to Fargo is less than 2.2 hours. Therefore, INS provides sufficient accuracy in this case study.

5.2.2.3 Discussion of Results

Comparisons

Comparison of the units considered in this case study leads to the following conclusions:

- Non-RNAV systems provide direct-to capability only when the intended destination is a VORTAC and the airborne receiver is within range of the transmitted signal.
- For RNAV VOR/DME-based navigation systems, automatic station selection eliminates the need for the pilot to concern himself with station acquisition and signal monitoring. Such capability makes the RNAV VOR/DME unit comparable to existing Loran-C and Omega/VLF units in terms of pilot interface. If a standard man-machine interface is incorporated into the design of all navigation system units, pilots can ideally operate any navigation system unit without regard to the type of signals processed.
- Manual station selection of navaids, although less desirable than automatic station selection, can be made acceptable in VOR/DME navigation systems through the use of a navaid data base. Specification of a navaid by either its three-letter ident or its frequency eliminates the need for time-consuming and error-prone entry of latitude, longitude, frequency, and magnetic variation of each desired navaid.
- ⁴ Current designated airways are defined on aeronautical navigation charts in terms of VORs, justifying the use of existing airway descriptors in the design of non-VOR/DME-based systems. Such design philosophy includes the use of a VOR/DME data base, as well as versatility in input and output formats to include bearing and distance information.

System Mix Conflicte

This case study concerns an aircraft using the on-board navigation system to establish and maintain a direct course to a final destination. Assume that two aircraft, each with a different navigation system, are given clearance to the same destination from the same position fix. One aircraft is using an Omega navigation system, the other a VOR/DME RNAV system that has rhumb-line navigation capability. Obviously, both aircraft could not depart from the same fix at the same time. If the controller has



Deviation from Route Centerline (nm)

spaced the aircraft to ensure adequate longitudinal separation, the controller can expect the aircraft to for ow the same flight path, since they are traveling between the same origin and destination. However, differences in course-calculation techniques (rhumb-line versus great-circle), in the accuracy with which intermediate waypoints are defined (manual charting versus computer calculation), in the accuracies of systems (2.5 nm for Omega versus 0.5 nm for VOR/DME), and in the coordinate systems used (spherical versus Clarke 1866), all interact to cause varying degrees of deviation between flight paths. Again, the burden is on the controller to monitor the magnitude of the deviations and take appropriate corrective action when it is necessary. Table 5-2 shows the distance disparities of each of these conflicts for a flight of 392 nm between Green Bay, Wisconsin, and Fargo, North Dakota. (The difference in accuracy of path definition between manual charting and computer calculation is not easily quantified and is therefore not included in the table.) The equations used to obtain the distance differences associated with path selection and earth model are shown in Appendix B.

Table 5-2. ACCURACY CONFLICTS IN CASE STUDY TWO								
Element of Conflict	Cause of Conflict	Resultant Distance Disparity (nm)						
Path Selection	Rhumb line versus great circle	6.41						
Navigation System	Omega versus VOR/DME	3.00						
Earth Model	Spherical versus Clarke 1866	1.30						

Figure 5-5 illustrates the way the navigation distance disparities shown in Table 5-2 can combine to result in a distance difference of 9.5 nm. Therefore, although each aircraft has been cleared to the same destination from the same origin, the route width required to contain all flight paths is greater than the current standard route width of 8 nm for domestic en route flights when within 51 nm of a VOR.

Case Study Two was for a flight of 392 nm. For longer flights, the relative position displacements resulting from differences in path selection and earth models increase. As an example, for a flight from New York to Los Angeles, the maximum cross-track distance between a rhumb-line path and a great-circle path is 135 nm.

5.2.3 Case Study Three

I

5.2.3.1 Description

Situation

Case Study Three considers an ATC request to "track outbound on the Green Bay VOR zero nine zero radial."



Figure 5-5. PICTORIAL REPRESENTATION OF SYSTEM MIX RELATIONSHIPS FOR CASE STUDY TWO

1

1

)

1

r r

ļ

.....

-

Applicable Systems

The single VOR system is discussed separately; non-RNAV VOR/DME systems, both single and multiple, are discussed jointly, as are both single and multiple RNAV VOR/DME systems. Also discussed, as common systems, are Loran-C, Omega, GPS, and INS.

5.2.3.2 Application of Scenario

In this scenario, the single VOR performs satisfactorily. The VOR receiver is tuned to the frequency of the Green Bay VOR. The pilot then establishes his position in terms of bearing from the VOR. Depending on the magnitude of difference between present course and desired course along the 090-degree radial, the pilot determines an appropriate course-cut angle by which the 090-degree radial can be intercepted without too much overshoot. Once track on the 090-degree radial is established, the pilot will continue to monitor the deviation from selected course to maintain a track of 090 degrees until the controller provides further instructions. The only difficulty that is encountered is the fading of the VOR signal if the request to maintain track is not altered before the aircraft is out of range of the VOR.

Non-RNAV VOR/DME

Existence of the DME does not significantly affect the capability of the system to satisfy the ATC request. The aircraft is still flown so that a VOR bearing indication of 090 degrees is maintained. The DME readout is helpful only in monitoring distance from the VOR so that ATC can be notified (before complete loss of signal) if this distance becomes too great.

RNAV VOR/DME

The workload necessary to establish an RNAV route to comply with this request can be avoided by recognition that the full capability of the navigation system is not needed in this situation. Rather, the response should be as described for the non-RNAV VOR/DME navigation system.

Loran-C, Omega, GPS, INS

For navigation systems not referenced to VOR signals, a flight path coincident with the Green Bay VOR radial of 090 degrees must be defined. This can be achieved by first defining the origin of the RNAV flight path to be the latitude and longitude of the Green Bay VOR. Destination can then be entered as a bearing and distance offset from the origin; the bearing would be 090 degrees, and the distance could be, say, 100 nm. Once this flight leg is activated, a course deviation is indicated. Response to the deviation would establish track along the 090-degree radial. An advantage of these RNAV systems is that they are not susceptible to loss of the VOR signal, and therefore loss of reference, as the aircraft becomes increasingly distant from the VOR along the radial.

5.2.3.3 Discussion of Results

Relative Comparisons

By requiring the aircraft simply to maintain a constant relative bearing to a VOR, this case study illustrates the most basic use of VOR-based navigation systems and the premise on which airways were established. Therefore, VOR-based navigation systems surpass other navigation systems in the ease with which compliance with the request is achieved. That does not mean, however, that Loran-C, Omega, GPS, or INS navigation systems would have difficulty defining a VOR radial; such a procedure is, in fact, quite easy. The point is that a VOR navigation system can perform more naturally in this environment than any other system.

System Mix Conflicts

The VOR station, which provides the signal used by the VOR-based navigation system, is at a fixed location on the earth. VOR receivers can navigate in relation to the VOR without regard to the actual geographical coordinates of the VOR as long as their destination or flight path is also defined with respect to the VOR.

Non-VOR-based navigation systems must use geographical coordinates to represent a VOR used for position reference. Conflicts can occur if the geographical coordinates used by Loran-C, Omega, GPS, or INS navigation systems do not correspond to the actual location of the VOR. This can result from error in pilot input for navigation systems with no access to a data base or from the airborne unit processor's use of an earth model substantially different from the model defining the reference chart.

The possibility that various types of navigation systems may not agree upon the absolute position of a reference VOR could create conflict. The seriousness of this situation is a function of the degree of difference in perceived location of the VOR. Error in pilot input presents the greatest potential for conflict. A mistake of only one arc minute in latitude would cause a displacement of 1 nm between the basis of reference and the actual VOR. Availability of a data base would eliminate the potential for such errors (if the data base was error-free).

In this case study, as in the previous situations, the controller is responsible for monitoring compliance with requests. If an aircraft does not satisfactorily respond to a request, the controller must take corrective action. It would be confusing if a pilot with a non-VOR-based navigation system had established a track on a 90-degree bearing from a waypoint corresponding with the coordinates of the Green Bay VOR and was then told by ATC that he was flying a track parallel to the requested track but with an offset of 10 nm. The pilot would lose confidence in his navigation system, revert to magnetic heading, and request vectors, thus increasing the burden on the controller.

5.2.4 / Case Study Four

5.2.4.1 Description

Situation

Case Study Four considers an ATC request to "report crossing Jones intersection." The "Jones intersection" denotes a frequently used position fix defined in terms of two bearings from two VORs. The request is similar to "report crossing the Green Bay three one zero radial two five miles fix," which is referenced to a single VORTAC.

General Comments

An intersection is a defined waypoint in the National Airspace System (NAS). Intersections are referenced to VORs in terms of bearings, and/or bearing and distance, and are so specified on aeronautical charts.

Applicable Systems

Single VOR, non-RNAV single VOR/DME, and RNAV single VOR/DME are discussed separately; multiple VOR/DME, both non-RNAV and RNAV, are discussed jointly. Also discussed as common systems are Loran-C, Omega, GPS, and INS.

Single VOR

The VOR receiver can be tuned to one of the reference VORs and can provide a basis for navigation along the VOR radial on which the intersection is located. The pilot establishes track on the course radial, compensates for any drift resulting from wind, and estimates time of arrival at the intersection through knowledge of groundspeed. Before the estimated time of arrival at the intersection, the pilot tunes the VOR receiver to the off-course station and monitors radial crossings. Time of passage occurs when the desired intersecting radial is crossed.

Non-RNAV Single VOR/DME

Addition of DME to the VOR receiver allows navigation within the signal coverage of the VORTAC, in terms of both bearing and distance relative to the VORTAC. Two techniques are used to establish point of passage over the defined waypoint when a non-RNAV VOR/DME navigation system is used. One method is to establish track on the reference VOR radial on which the intersection is defined. The DME readout is then monitored for indication of convergence to the distance specification for the intersection. Coincidence of measured bearing and distance with desired bearing and distance signifies passage over the intersection and can be so reported to ATC. A second, less practical, method is used when a circular arc is flown about the reference VORTAC in which a DME range is held constant, corresponding to the specified distance of the intersection from the reference VORTAC. Monitoring of VOR bearing readout then indicates convergence on the intersection and subsequent passage. Both of these techniques are routinely used; the choice corresponds to the flight path for which ATC clearance was granted. ATC may not require a report of passage over an intersection if the flight plan is not along one of the VOR radials defining the intersection.

RNAV Single VOR/DME

Each RNAV waypoint in a single VOR/DME navigation system is defined with respect to a particular VORTAC. The Jones intersection can be designated a waypoint, because it is intended to be on the flight path; it can therefore be defined in accordance with the charted bearing and distance from the reference VORTAC. Distance to this waypoint can then be directly monitored via the navigation system display. However, the need to establish a new waypoint imposes on the pilot the additional task of inputting the frequency of the reference VORTAC and the bearing and distance from that VORTAC, which define the waypoint. Since the addition of a new waypoint in a multiple waypoint RNAV unit does not have to overwrite existing waypoints, the pilot can easily revert to the previously designated waypoint once the aircraft has passed over the intersection.

Multiple VOR/DME

A multiple VOR/DME navigation system is well suited to meet the requirements imposed in this case study, regardless of whether the system has RNAV capabilities. One of the VOR/DME receivers of the non-RNAV system can be dedicated to the VORTAC providing the reference for the intersection, while the other receiver can be available for primary navigation. For an RNAV multiple VOR/DME system, the unit can be used in the manner described for an RNAV single VOR/DME system.

Loran-C, Omega, GPS, INS

All remaining RNAV systems (Loran-C, Omega, GPS, and INS) provide the capability to define waypoints in terms of latitude and longitude coordinates. Therefore, the coordinates of the Jones intersection can be entered into the RNAV system to define a waypoint, and subsequent distance to the intersection can be directly monitored.

The ease with which RNAV systems can comply with the ATC request in this case study is diminished by the infrequency with which position fixes on charts are specified in terms of latitude and longitude in addition to bearing and distance. As an alternative to entering the latitude and longitude of the intersection, the coordinates of the VOR from which the Jones intersection is referenced can be entered to define a waypoint refence. A bearing and distance offset from this reference can then be entered to define the Jones intersection, the actual desired waypoint. However, such capability in input format is currently limited in availability.

5.2.4.3 Discussion of Results

Relative Comparisons

VOR/DME systems are better suited for response to this case study request than are non-VOR/DME-based navigation systems for the reasons stated in the discussion of Case Study Three (Section 5.2.3.3).

System Mix Conflicts

Because this case study presents the same potential for conflict as the last case study, the discussion of Case Study Three also applies to this situation.

5.2.5 Case Study Five

5.2.5.1 Description

Situation

The last case study to be considered is the pilot response to an ATC clearance to "fly North Atlantic Track F." Track F is the over-water portion of a daytime flight from London to New York. About 3.7 hours is required to fly the 1806 nm track at Mach .84 (about 490 knots). Procedures during the over-land portions of the flight are similar to those described in Case Studies One through Four.

General Comments

The system of defining the North Atlantic tracks is called the Organized Track System (OTS). A set of tracks is selected twice each day according to meteorologic conditions. The selected tracks are separated laterally by 60 nm (± 30 nm track width) and are labelled with alpha designators (e.g., Track A) sequentially from north to south. Over-water air traffic control procedures include reporting position in terms of latitude and longitude at specified intervals along the tracks. Reporting points are located at whole 10-degree meridians and at whole-degree latitude lines. A chart that provides the latitude and longitude of the reporting points defining the North Atlantic tracks on a given day is called a route chart. Each track comprises, then, a series of great-circle segments between specified reporting points and is illustrated on the route chart. A track is terminated at each end by a waypoint called a track gateway or anchor point. On the day of this flight, Track F is defined between the anchor points Cork in Ireland and Springdale in Newfoundland.

Applicable Systems

Neither VOR/DME nor Loran-C provides adequate coverage, and they are not discussed. Omega and GPS are discussed jointly. The Doppler navigation system and INS are each discussed separately.

5.2.5.2 Application of Scenario

Omega and GPS

The pilot enters the latitude and longitude of the anchor points and the specified reporting points into the navigation unit. The unit then establishes a route comprised of great-circle segments defined between the entered waypoints. This route coincides with Track F. Both Omega and GPS provide continuous navigation accuracy in excess of that required throughout the over-water portion of the flight. As has been assumed previously, the navigation unit in use has been operational for some time, and the received signals provide satisfactory navigation capability at the time of crossing the Cork anchor point. A display of current latitude and longitude is available for monitoring, providing a convenient means for reporting position at the required intervals.

Doppler

Doppler navigation systems used in recent years in civil applications use a course generation method that is neither rhumb line nor great circle. Instead, a desired track angle is computed for each flight segment that defines a constant magnetic course. The pilot enters the desired track angle so defined, along with distance-to-go, to establish each segment of his desired path. A series of such segments is used to establish Track F in this case study.

The display of distance-to-go is used to indicate when the aircraft crosses a reporting point. As the aircraft traverses the segment, the distance-to-go display counts down to zero. When the display reaches zero nm, the pilot knows the aircraft is crossing a reporting point. The latitude and longitude of this point can then be obtained from the route chart for reporting purposes.

INS

The latitude and longitude of the anchor points and the reporting points are entered into the INS unit. As described for GPS and Omega, the great-circle route established by the INS coincides with Track F, and a display of current latitude and longitude is available for use if reporting position along the track. Because of the position error accumulated in the INS, VOR/DME is generally used to update the INS after the aircraft passes from the over-water phase to the over-land portion of the flight. The transition between the two navigation systems is typically performed automatically in a gradual manner. Any position discrepancy between the two units is eliminated slowly in order to avoid sudden changes of input to the autopilot.

5.2.5.3 Discussion of Results

Relative Comparisons

Each segment of a North Atlantic track is a great-circle track defined between two reporting points. Each reporting point is specified by its latitude and longitude. The navigation systems that establish great-circle tracks permit navigation of the North Atlantic tracks more naturally than current Doppler navigation systems.

Since position reports are given in terms of latitude and longitude, the navigation systems discussed that can display current latitude and longitude provide a more convenient means of complying with position reporting requirements than most Doppler navigation systems that do not.

Many users of Doppler navigation systems have discontinued over-water use because of the pilot workload and the potential for errors in position reporting associated with the previously discussed accuracy deficiencies. However, Doppler navigation systems now under development provide both great-circle route generation capability and a display of current latitude and longitude. These features provide operational effectiveness for Doppler comparable to that demonstrated by INS in this case study.

System Mix Conflicts

In this case study an aircraft uses a navigation system to establish and maintain a course coincident with North Atlantic Track F. Consider two aircraft on adjacent Tracks E and F both at flight level 350. Both aircraft are westbound and are given clearance in Ireland at nearly the same time. The aircraft flying Track E is using a Doppler navigation system, and the aircraft flying Track F is using conventional INS. Although the track center lines are separated by one degree of latitude (i.e., 60 nm measured along a meridian), the actual lateral separation (i.e., perpendicular distance) between the two aircraft can be significantly less than 60 nm, as described below.

The required 60-nm track separation is based on one-degree intervals of latitude between adjacent tracks measured along a meridian. The perpendicular distance between the center lines of adjacent tracks can be somewhat less than 60 nm; it is a function of the course angles of the tracks. For example, the perpendicular distance between Tracks E and F midway between the reporting points at 40°W and 50°W is 56.8 nm, as shown in Figure 5-6. The maximum distance disparity between the course used by the Doppler navigation system and the corresponding great-circle track between these same two reporting points is 7.2 nm along Track E. The distance disparity resulting from differences in course generation methods interacts with the accuracies of the two navigation units to cause varying degrees of deviation from the intended lateral separation between the two aircraft. The distance from the anchor points in Ireland to points midway between the reporting points at 40°W and 50°W along Tracks E and F is about 1300 nm. The Doppler navigation system accumulates a cross-track error of 41.3 nm and an along-track error of 13.9 nm in traversing 1300 nm,



Figure 5-6. SYSTEM MIX RELATIONSHIPS FOR CASE STUDY FIVE

.....

forming the error ellipse illustrated in Figure 5-6. Traveling at 490 knots, the aircraft will take about 2.7 hours to traverse 1300 nm. The conventional INS accumulates 4.3 nm of position error in 2.7 hours and is shown as a circular error in Figure 5-6. Figure 5-6 illustrates the way in which the system errors interact with the distance disparity to yield a lateral separation of only 4.0 nm between the two aircraft.

5.3 SUMMARY OF CASE STUDIES

Table 5-3 summarizes navigation system performance in each of the five case studies as they relate to six performance rating criteria. Case Study One demonstrated that navigation systems cannot replace the need for heading devices in the current ATC environment. The heading sensors currently in use range from low-cost "wet" compasses to gyro-stabilized flux-value compasses. The inability of navigation systems to provide heading information does not, however, preclude the possibility of unique advantages and disadvantages being attributed to particular navigation systems during use in other operational scenarios. Although the ratings are intended to indicate the relative operational capability of the systems relative to each case study, totaling the numbers in each column leads to some interesting observations. The level of capability for each system can be ranked according to the value of the sum of the codes; the lower the value, the more capable the system. This method of ranking system capability assumes that all case studies are of equal significance in their relationship to performance. In addition, it must be remembered that although VOR/DME, Omega, Loran-C, Doppler, and INS are ranked in accordance with existing levels of performance and coverage, GPS is evaluated with respect to a projected level of performance capability.

	Table 5-3. RATING OF NAVIGATION SYSTEM PERFORMANCE IN CASE STUDIES*												
		Non-RNAV		RNAV									
Case Study	Single VOR	Single VOR/DME	Multiple VOR/DME	Single VOR/DME	Multiple VOR/DME	Loran-C	Omega/VLF	GPS	Doppler	INS			
1 (Fly Heading)	6	6	6	6	6	6	6	6	6	6			
2 (Fly Direct To)	5	5	5	3	2	5	1	1	5	1			
3 (Fly Radial)	1	1	1	1	1	2	2	2	5	2			
4 (Réport Position)	4	1	1	1	1	2	2	2	5	2			
5 (Fly Ocean Track)	5	5	5	5	5	5	1	1	2	1			
Total	21	18	18	16	15	20	12	12	23	12			

*Performance rating criteria for this table are as follows:

1 = Primarily requires pilot monitoring.

2 = Requires some manual calculations and/or routine interaction between pilot and unit.

B = Requires ancillary equipment or periodic manual corrections by pilot.

4 = Requires much manual calculation and/or a high level of pilot interaction with unit.

5 = Objective cannot be met because of lack of signal coverage or inadequate accuracy.

6 = Objective cannot be met because of limitations of system capability.

A Doppler navigation system is shown in the scenarios to be the least operationally capable system because of inadequate accuracy capability in en-route domestic applications. Omega/VLF, GPS, and INS are overall the most effective systems because of their transoceanic navigation capability, as discussed in the account of Case Study Five. The RNAV multiple VOR/DME system is ranked lower than Omega/VLF, GPS, and INS only because of lack of transoceanic coverage. It is otherwise the most effective system, since the current ATC environment over land is predicated on the VOR/DME system. Loran-C was given a lower ranking than Omega/VLF, GPS, and INS only because of lack of Loran-C coverage in the area of interest defined in Case Study Two and in the transatlantic route defined in Case Study Five. Recent developments in the design of Loran-C receivers indicate a potentially significant advancement in Loran-C operational capability. If the design goals are achieved and verified, there will be an impact on the performance rating of Loran-C as presented in this report.

As a result of the case study evaluations, the navigation systems may be ranked in descending order of effectiveness for world-wide applications as follows:

- Omega/VLF, GPS, INS
- RNAV multiple VOR/DME
- RNAV single VOR/DME
- Non-RNAV VOR/DME
- Loran-C
- Single VOR
- Doppler
CHAPTER SIX

CONCLUSIONS AND RECOMMENDATIONS

Currently available navigation systems offer numerous combinations of capabilities. The multitude of choices results from the different requirements imposed by different users, mandated by particular applications. The most basic decision to be made by a user is the selection of the particular type of navigation system (Loran-C, Omega/VLF, VOR/DME, GPS, Doppler, or INS) that best meets his needs. The most critical criteria considered in this initial selection process are signal accuracy and coverage. Table 6-1 summarizes each navigation system's ability to provide sufficient coverage and accuracy to be considered acceptable in different operational environments and flight phases. VOR/DME, Omega, Loran-C, Doppler, and INS were evaluated in terms of existing levels of performance and coverage, whereas GPS was evaluated with respect to a projected level of performance capability.

Loran-C has sufficient accuracy to meet all en-route requirements, but . the limited number and coverage area of Loran-C station chains restrict its current use to only a subset of domestic en-route applications. Although this restriction limits the suitability of Loran-C as an independent means of navigation, it does not impede its consideration for use in a system mix for domestic and off-shore applications. Recent developments in the design of Loran-C receivers indicate potential advances in operational capability that could improve the suitability of Loran-C for independent domestic en-route navigation. Use of Loran-C for terminal and nonprecision approach navigation, if approved, will be limited to areas where signals are available.

Omega/VLF provides worldwide coverage, with an accuracy suitable for en-route navigation but not sufficient for terminal or nonprecision approach operations.

GPS, even with a denial of accuracy to a level of 500 meters, would provide sufficient accuracy for en-route, terminal, and many nonprecision approach navigation operations, assuming continuous availability of four satellites.

の学校

6-1

Teble 6-1. SUMMARY OF NAVIGATION SYSTEMS' ABILITY TO PROVIDE ACCURACY AND COVERAGE									
Operational Environment									
Navigation		En Route							
System	Domestic	Trans- oceanic	Remote Offshore	Terminal	Approach				
Loran-C	Suitable ¹	No Coverage	Suitable	Suitable ¹	Suitable ¹				
Omega/VLF	Suitable	Suitable	Suitable ²	Insufficient Accuracy	Insufficient Accuracy				
GPS	Suitable	Suitable	Suitable	Suitable	Suitable ²				
VOR/DME	Suitable	No Cov erage	No Coverage	Suitable	Suitable ¹				
Doppler	Insufficient Accuracy	Suitable ⁴	Insuffi- cient Accuracy	Insufficient Accuracy	Insufficient Accuracy				
INS	Suitable ³	Suitable	Suitable ²	Insufficient Accuracy	Insufficient Accuracy				
¹ When covera ² Under condi ³ For flights ⁴ For ocean t	¹ When coverage is available. ² Under conditions allowing attainment of sufficient accuracy. ³ For flights of less than one hour. ⁴ For ocean tracks less than 400 nm.								

Accuracy requirements are specified in FAA AC-90-45A.

Because VOR/DME navigation is the basis upon which domestic navigation standards were developed, it provides an acceptable level of accuracy. Lack of coverage in oceanic, remote, and offshore areas resulting from line-of-sight limitations and unavailability of suitable sites limits the application of VOR/DME.

Loppler navigation systems can provide sufficient capability to meet oceanic en-route accuracy requirements for limited path lengths, but insufficient accuracy for all domestic requirements.

INS provides sufficient accuracy for oceanic en-route requirements. Because INS accuracy degrades as a function of time in flight, INS can meet domestic en-route requirements only for flight durations of less than one hour. However, INS can be used for extended periods of time when position updates are available from suitable radionavigation units. The time between updates must not exceed that which would result in the INS exceeding the allowable error. . On the basis of the results of the case studies of operational performance capability, the navigation systems are ranked in descending order of overall (i.e., both domestic and oceanic) effectiveness as follows:

- Omega/VLF, GPS, INS
- RNAV multiple VOR/DME
- RNAV single VOR/DME
- Non-RNAV VOR/DME
- Loran-C

T

- Single VOR
- Doppler

The operational effectiveness of each navigation system was measured by comparing the systems in terms of the ease with which they could be used to meet the stated objective of the particular case study. Emphasis was placed on the limitations of the navigation system, not on the functional capabilities of particular CDU designs.

Selection of GPS, Omega/VLF, and INS as the most operationally effective systems is predicated on their transoceanic navigation capability. When oceanic operations are not considered (Case Study 5) the navigation systems are ranked in descending order of effectiveness in domestic operations as follows:

- RNAV multiple VOR/DME
- RNAV single VOR/DME, Omega/VLF, GPS, INS
- Non-RNAV VOR/DME
- Loran-C
- Single VOR
- Doppler

The high ranking of RNAV VOR/DME systems is to be expected, since the current domestic ATC environment is based on VOR/DME. GPS is ranked lower than multiple VOR/DME RNAV solely on the basis of perceived functional capability as demonstrated in the case studies. It must again be emphasized, however, that there is no technical impediment to enhancing the functional features attributed in this report to GPS units. Loran-C is given a lower ranking than Omega/VLF and GPS only because of limitations in current Loran-C coverage.

The case studies identified an effect on pilot and controller workload when a geographically referenced navigation system was utilized in a station-oriented, procedural environment. Pilot workload was primarily affected by the differences between non-RNAV and RNAV navigation systems and the operation of those systems in situations not optimally suited to their specific capabilities. The case studies also revealed some potentially serious inadequacies regarding the characterization of performance of individual navigation systems. Accuracy of navigation systems has improved to the point where some systems are capable of a greater degree of accuracy then the reference against which accuracy is measured. Differences between the accuracies of earth models, which provide the basis for determining geographical position, can be greater than the accuracy capability of some navigation systems. Differences between possible flight paths (great-circle versus rhumb-line) when area navigation systems are used can far exceed specified route widths, even though the two paths have the same origin and destination. At the point of maximum deviation, there is a cross-track difference of 135 nm between the great-circle path and the rhumb-line path between New York and Los Angeles. Differences in sky wave propagation models used in Omega navigation systems can result in differences in accuracy ranging from 3.2 nm to 8 nm.

Unrestricted use of area navigation techniques could result in such diversity of routing that controllers would be overburdened in their attempts to maintain separation of aircraft without conflict. Two forms of action can be taken to promote safe and efficient use of the capabilities of area navigation systems:

- Adoption of standards that define accuracy reference parameters
- Modernization of the ATC system so that ATC can accommodate the capabilities of area navigation systems

Table 6-1 indicates which navigation systems are capable of providing service in each application area, reflecting a probable rather than certified capability to meet existing requirements. A navigation system and possible combinations of systems that can provide navigation capability in all operational environments and flight phases represented in Table 6-1 are as follows:

- GPS*
- GPS and VOR/DME
- GPS and Loran-C
- VOR/DME and Omega/VLF**
- VOR/DME and INS**
- VOR/DME and Doppler**
- VOR/DME, Omega/VLF, and Loran-C
- VOR/DME, INS, and Loran-C
- VOR/DME, Doppler and Loran-C

*Suitable only if approach accuracy requirement is achievable. **Suitable only if remote and offshore accuracy is achievable.

6-4

This list seems to suggest that GPS is the leading contender for selection as the standard navigation system. It could meet all existing requirements. However, to support the adoption of GPS as the primary navigation system of the future would be premature at this time. Acceptance of GPS depends on the resolution of the technical, operational, economic, and institutional issues relevant to its benefit to the civil aviation community. The absence of GPS in the role of a primary navigation system requires the selection of a suitable navigation system mix of existing systems. Since Doppler meets oceanic en-route requirements only for flights less than 400 nm, the VOR-DME-Doppler combination is suitable for only a limited number of conditions. Eliminating the combinations including GPS and Doppler from the list leaves the following combinations:

- VOR/DME and Omega/VLF
- VOR/DME and INS

I

- VOR/DME, Omega/VLF, and Loran-C
- VOR/DME, INS, and Loran-C

Although it would seem that the first two of these four system combinations would be appropriate in terms of minimizing the number of required systems, neither of them currently provides the accuracy desired for offshore navigation. Loran-C provides not only increased accuracy for offshore navigation but also terminal navigation capability in remote areas where VOR/DME is not suitable. Therefore, Loran-C should be included in the navigation system combinations. By the process of elimination, only two combinations of proven navigation systems are able to meet the technical and operational requirements in the context of existing conditions:

- VOR/DME, Omega/VLF, and Loran-C
- VOR/DME, INS, and Loran-C

It should be noted that each of the various navigation signals identified in this report do not have to be utilized in singular packages exclusive of one another. Inputs from multiple navigation systems can be combined to take advantage of those systems most appropriate for use at any given time.

APPENDIX A

DETAILED SYSTEM DESCRIPTIONS

1. LORAN C

1.1 <u>History of Development</u>

Loran-C (Long-Range Navigation) was developed in 1957 to provide greater accuracy and longer-range capability than Loran-A. The United States Coast Guard assumed responsibility for operation of the system in August 1958. In 1974, Loran-C was selected as the United States government-provided navigation system for civil marine use in United States coastal areas.

1.2 Facilities

All Loran-C transmitters operate at a fixed frequency of 100 kHz and must confine 99 percent of their radiated energy within the band of 90 to 110 kHz. The 100-kHz carrier frequency provides a nearly optimum signal-toatmospheric-noise ratio for a range of 1,000 to 2,000 kilometers (621 to 1,242 nautical miles). Loran-C signals are transmitted with a peak power of four megawatts via a single vertical antenna tower, which may be as high as 1,350 feet. The ground plane required for transmission of Loran-C ground waves consists of an extensive network of wires buried in the ground to a radius of about 1,000 feet. All Loran-C transmitting stations are equipped with cesium frequency standards, which allow each station to maintain synchronization to absolute time without the need of reference to another station.

Loran-C transmitting stations form chains, each of which consists of a master and two to four secondary (slave) transmitting stations separated from the master by 600 to 800 miles. The geometry of the various chains is shown in Figure A-1. Figure A-2 shows the coverage provided by Loran-C, with the exception of the Canadian East Coast and Commando Lion chains. (The Commando Lion chain has stations in Korea and Japan. Although it is used by the United States Air Force, it is not officially available to civilian users. The Canadian East Coast chain uses existing stations of other chains -- two from the Northeast United States chain and one from the North Atlantic chain.)

Current proposals are to expand Loran-C coverage in Spain, France, Norway, Canada, Russia, Hawaii, and mid-continent United States. It is doubtful that Loran-C will ever be made available in the southern hemisphere. ;1







10

C

Ľ

E

1



Loran-C navigation is based on time difference (TD) measurements between the user and the transmitters of the appropriate chain. Each chain is monitored by use of one or more system area monitor (SAN) stations within the coverage area to observe the TDs of each master-secondary pair. If an observed TD varies from the calibrated control TD by half of the prescribed control tolerance (±200 nanoseconds or better), the SAM directs a change in the timing of the secondary station to remove the error. If the observed TD differs from the control TD by more than the control tolerance, the transmitted signal is coded to advise users that the TD is unusable. For convenience, letters are used to designate the stations in a chain. The master station is referred to as M, and the secondary stations are denoted as W, X, Y, and Z.

1.3 Signal Characteristics

The transmitting stations of a Loran-C chain transmit groups of pulses at a specified group repetition interval (GRI). The master station transmits its pulses in groups of nine at a repetition rate of 10 to 25 groups per second. The transmissions of the secondary stations are delayed, with respect to the time of arrival of the signal from the master, by a specified time called the secondary coding delay. This delay ensures that signals from two or more stations in a chain cannot overlap in time anywhere in the coverage area. The GRI must be of sufficient length to allow time for transmissions of the master (10,000 microseconds [µs]) and each secondary station (8,000 µs per station) and for the secondary coding delays. The minimum GRI is therefore a direct function of the number of stations and the distance between them. The particular GRI specified for each chain is selected to minimize the mutual interference caused by adjacent chains.

Each station transmits one pulse group per GRI. The master pulse group consists of eight pulses spaced 1,000 μ s apart. A ninth pulse, transmitted 2,000 μ s after the eighth, is used for identification of the master. To warn of an error in the transmissions of a particular station, the ninth pulse is turned on and off in a specified code denoted as blink. Pulse groups for secondary stations contain eight pulses spaced 1,000 μ s apart and use the first two pulses for blink. All secondaries use the same pulse format, which is automatically recognized by most modern Loran-C receivers.

Each pulse within a group is designed to build up and decay slowly, defining the envelope shape shown in Figure A-3. The zero crossing point near 30 μ s from the start of the pulse is identified as the third-cycle zero crossing. The third cycle is typically identified by the slope of the pulse envelope. Confining transmission sampling to only the first three cycles eliminates interference from minimum time-delay sky waves. Distortion of the envelope shape, called cycle-to-envelope discrepancy (CED), can cause tracking of a cycle other than the third cycle. Tracking of a cycle zero crossing other than the third results in a bias error, the sign and magnitude of which are determined by the direction and number of cycle shifts. This form of error is commonly referred to as cycle-jump error. When it occurs, the error is typically a single cycle jump, which causes a constant error of $\pm 10 \ \mu$ s in TD.



Figure A-3. SHAPE OF LORAN-C PULSE

Because the amplitude of the signal envelope at the third-cycle zero crossing is only about 50 percent of the peak value, multiple pulses (eight per group) are used so that more signal energy is available at the receiver to improve the signal-to-noise ratio without the necessity of increasing the peak transmitted power capability of the transmitters.

Loran-C uses ground waves instead of sky waves, because the stable nature of the ground wave is not affected by ionospheric fluctuations, which affect the propagation characteristics of sky waves. Contamination of the ground waves by sky waves is eliminated in Loran-C by use of the pulse transmission technique. Receiver reception of a sky wave is delayed between 33 and 1,000 μ s after reception of the ground wave. Sky wave reception with a minimum delay time is discriminated against by receiver sampling of the Loran-C transmission at the zero crossing point near 30 μ s of the start of the pulse. This technique results in full ground wave stability. Longdelay sky waves could perturb the integrity of the transmission by overlapping the ground wave of the succeeding pulse. To prevent this from happening, the phase of the 100-kHz carrier of each pulse is changed in accordance with a predetermined pattern referred to as a phase code.

Phase coding of the pulses within a group also assists in identifying the master station. The phase of the 100-kHz carrier of each pulse is either in phase or 180 degrees out of phase with a defined reference carrier in accordance with the assigned code. Different phase codes are defined for the master and secondary pulse groups to identify master and secondary stations. Two sets of phase codes are used for master and secondary pulse groups, alternating between successive GRIs so that the phase code changes with each GRI and repeats every other GRI.

1.4 Signal Processing

A position fix is established by the intersection of two or more hyperbolic lines of position (LOPs), each of which represents a constant range difference from two transmitters. Range differences are based on TDs. TD measurements between the receiver and at least three transmitting stations are necessary to establish a position fix. A signal propagation model is used in the process of determining range differences from the measured TDs.

Although Loran-C receivers are designed for processing of ground waves, a sky wave mode can be used when the receiver is beyond the reception range of Loran-C ground waves. Although use of sky waves provides less accuracy than use of ground waves, a single-hop sky wave can be received at distances from the transmitter of about 2,300 nautical miles (nm) -- nearly double the range of the ground wave. Use of Loran-C in the hyperbolic mode does not require a precision oscillator in the receiver for measurement of TDs, because synchronization of time references is maintained by the transmitting stations. Therefore, user clock bias does not affect the measurement of TDs.

Another positioning technique used in some Loran-C receivers is called the ranging, range-range, or rho-rho mode. In the ranging mode, a time measurement provides a circular rather than a hyperbolic LOP, which allows a position fix using two individual stations rather than two station pairs. Another advantage of the ranging mode is the elimination of the geometric dilution due to using the hyperbolic mode at extended ranges. Geometric dilution is a function of the user's position relative to the transmitting stations. The gradient, or spacing between consecutive LOPs per unit of time difference, such as 1 $\mu s,$ increases with the divergence of the hyperbolic LOPs. The effect is most pronounced along baseline extensions, which are beyond the two stations that define the end points of the baseline. When the gradient is large, a relatively small TD error will result in a relatively large position error. In the ranging mode, the gradient is a constant equal to the propagation velocity, thus eliminating the effect of gradient on geometric dilution. The ranging mode can therefore be used to extend the coverage area beyond that possible in the hyperbolic mode by overcoming the geometric dilution at extended ranges and by requiring the user to be within range of only two transmitting stations. Use of three stations enhances the accuracy of a position fix, because redundant information is used to estimate errors.

All of these advantages of the ranging mode are seriously compromised by one disadvantage -- the need for a very precise and stable time reference in the receiver. The high cost of this type of equipment limits the use of the ranging mode for Loran-C navigation systems.

Before any navigation mode is used, the necessary signals must be acquired by the airborne receiver. The time required for signal acquisition

is typically thirty seconds to two minutes, with a maximum delay of five minutes -- depending on geometry, signal strengths, and the accuracy with which current position is known. Use of Loran-C in the hyperbolic mode requires three stations (a triad); the loss of any single station requires selection of a new triad. Although only three stations are used for navigation, continuous tracking of all stations in a chain eliminates the loss of time during signal acquisition when a new triad is selected. The eventual capability of using all signals from a chain for navigation will provide sufficient redundancy to ensure continual tracking stability and accuracy, even during station loss.

A back-up mode is available on most receivers to allow master independence when the master station fails. Loss of the master station automatically initiates the blink code on all stations in the affected chain, since the integrity of master-secondary TDs can no lenger be ensured through monitoring. However, because each station uses an independent clock for synchronization to absolute time, it is reasonable to assume that interstation synchronization will not degrade rapidly. A problem is encountered when trying to interpret the blink code. If only one station in a chain is blinking, it can safely be assumed that only that station has a problem and should not be used. If all stations are blinking, loss of the master is definitely indicated, but the level of integrity of individual stations is not obvious. However, if caution is used and signal parameters are monitored, the blinking signals can still be used for navigation.

Loran-C receivers are susceptible to localized interference created by such things as VLF transmitters, arc-welding operations, and 100-kHz communication channels carried on power lines. Notch filters can be used to minimize continuous wave interference from Decca navigation chains and communication stations within the same band, but they are not effective in eliminating broad-spectrum interference.

The achievable accuracy of Loran-C depends on a number of factors, including the following:

- Number of stations processed
- Signal quality

t

- Geometry of stations in relation to aircraft
- Definition of propagation model
- Region of operation

As a function of these factors, typical observed absolute accuracy ranges from a few hundred feet to a few miles. Repeatable accuracy is highly stable, usually within 100 to 200 feet.

The high degree of repeatability is an indication of the significance of the propagation model. With accurate modeling of the propagation characteristics, absolute accuracy can approach repeatable accuracy. Biases

result from the relative differences between the fixed propagation model and the actual propagation characteristics of the signals received from each station.

1.5 Operational Characteristics

Because Loran-C was initially implemented for marine use, it had some operational disadvantages that prevented acceptance for airborne navigation. Many of those disadvantages have since been eliminated through system modifications and improvements in user equipment. The Loran-C system was initially master-dependent, in that only the master of each chain had a clock, to which each slave was reference. If the master went off the air, the slaves lost their time reference, and the entire chain was out of operation. Independent synchronization to absolute time is now possible, since all stations are equipped with cesium clocks.

When Loran-C coverage provided only three stations in a given area, loss of any station led to system unavailability during hyperbolic navigation, because of the lack of redundancy. The addition of more stations now ensures coverage from at least four stations in a given area.

Advances in user equipment have evolved from the necessity of ID overlay charts to today's fully automatic Loran-C RNAV systems. The advent of microprocessor technology has brought not only increased technical capability but also increased reliability and low cost.

The first airborne antennas used with Loran-C were tail-cap and other electric dipole configurations. Signal loss and difficulty in reacquisition were significant problems, because of the antenna's susceptibility to precipitation (P) static. The tail-cap antenna also exhibited a null pattern problem. Use of H-field orthogonal loop antennas has resolved these problems.

2. OMEGA

2.1 History of Development

The Omega navigation system was developed to extend the range of a system like Loran-C to about 5,000 miles. For the past 25 years, the United States Navy has conducted intensive research and development for Omega. The first experimental Omega stations were established in Norway, Hawaii, Trinidad, and New York by 1964, when the Naval Research Laboratory conducted the first evaluation flights of a prototype airborne Omega receiver. Seven of the eight permanent Omega stations are now operational in Norway, Liberia, Hawaii, North Dakota, La Reunion, Argentina, and Japan; the eighth station, located in Australia, is not yet operational. A temporary station was operating in Trinidad until December 31, 1980. In July 1978, the United States Coast Guard assumed full responsibility for the operation and maintenance of United States-based Omega. Until the eighth station is completed and resulting system accuracy and coverage can be measured and validated, the Omega network cannot be declared an operational system. However, coverage and accuracy of Omega are being verified on a regional basis.

2.2 Facilities

Omega stations transmit very low frequency (VLF) (10 to 14 kHz) continuous wave (CW) signals on a common carrier frequency on a time-shared basis. Except for the antenna, the complement of electronic equipment in each transmitting station is identical. The antenna system is either a vertical tower about 450 meters high supporting an umbrella of transmitting elements or a valley span typically 3,500 meters in length. Each Omega station has a transmission power of 10 kW. The major elements of each station are timing and control, transmitter, and antenna tuning.

Each Omega station synchronizes its transmissions with highly stable cesium-beam frequency standards, which are referenced to the atomic time scale. Monitor systems provide phase measurement data between stations. The data are used in an advanced optimal-estimation and control algorithm to provide corrections for any offset or divergence of a transmitter from mean Omega system time.

2.3 Signal Characteristics

All Omega stations transmit four frequencies (10.2, 13.6, 11.33, and 11.05 kHz) on a semicontinuous basis, with a basic repetition period of 10 seconds, as in the format shown in Figure A-4. The notations f_1 through fg in the figure denote transmitted frequencies unique to each station, as follows:

fl		12.1	kHz	f4	**	13.1	kHz	f ₇	-	13.0	kHz
f2	-	12.0	kHz	f ₅	***	12.3	kHz	£	-	12.8	kHz
f ₃	-	11.8	kHz	f	*	12.9	kHz	•			

The 0.2-second interval between successive transmissions eliminates the possibility of overlap of signals received from different stations and allows for a margin of error in alignment of the receiver commutator. The Omega signal format was designed so that each station could be identified by the transmission of a particular frequency at a prescribed time.

2.4 Signal Processing

Position determination for Omega can be obtained by either the hyperbolic or the circular (ranging) technique. Hyperbolic systems utilize phase differences so that any drift in the local oscillator will affect both phase measurements equally and will therefore be canceled. Therefore, a highprecision oscillator reference is not required for the hyperbolic mode. The ranging mode requires individual phase measurements rather than phase differences. The level of precision required in the receiver oscillator is determined by the mode of operation in which it will be used.

A	10.2	13.6	11.33	£1		11.05	f1		10.2
B	- f 2	10.2	13.6	11.33	f		11.05	f2	
с	-f 3		10.2	13.6	11.33	- f 3		11.05	- f 3 -
D	11.05	£4		10.2	13.6	11.33	£4		11.05
E	-f 5-	11.05	f 5		10.2	13.6	11.33	£ 5	
F	- f 6		11.05	f		10.2	13.6	11.33	f ₆
G	11.33	f ₇		11.05	f 7		10.2	13.6	11.33
H	13.6	11.33	f 8		11.05	f ₈		10.2	13.6
	100	1.0	1 1	1.2	1.1	0.9	1.2	1.0	0.9

Figure A-4. SIGNAL FORMAT FOR OMEGA

2.4.1 Ranging Mode

Several methods of position determination can be implemented in the ranging mode. The rho-rho method requires direct measurement of the phase from only two stations. The intersection of the circular LOPs resulting from the phase measurements of the two stations defines position. The ambiguity caused by the existence of two points of intersection is resolved by proper initialization. Since there is no redundant information available for monitoring errors, the local oscillator must be highly precise - preferably of the atomic standard type - to achieve acceptable accuracy and reliability. In addition, because each phase measurement is used directly, the system is more susceptible to propagation anomalies, which are unpredictable.

Use of three stations in a rho-rho-rho mode provides sufficient redundancy to permit limited self-calibration of the local oscillator. However, the assumption that any discrepancy in all LOPs meeting at a single point is wholly the result of oscillator drift may not be correct.

2.4.2 Hyperbolic Mode

Position fixing for Omega in the hyperbolic mode requires comparison of phase values obtained from signals of several transmitting stations. Loran-C provides hyperbolic LOPs defined by differences between the arrival times of signals from two transmitters. With Omega, hyperbolic LOPs are formed by contours of constant phase differences in the signal fields of

the transmitting stations. All points along a given contour have the same difference in distance from the two transmitting stations.

Identical phase angles emanating from each transmitter are repeated at every wavelength of the transmitted signal. Therefore, the same difference in phase angles occurs at the intersection of all integralwavelength wavefronts radiating as circles from each transmitting station. The locus of these intersections is a hyperbolic contour of constant phase, called an isophase contour. The midpoint of the hyperbolic isophase contour line segment joining two points of intersection is a half-wavelength from either wavefront; since it is part of the isophase contour, it also has the same difference in phase angles as at the intersections of integral wavefronts. Thus, identical phase differences repeat not only every wavelength, but every half-wavelength, as illustrated in Figure A-5.







2.4.3 Laning

In the ranging mode, all circular LOPs spaced one wavelength apart correspond to the same phase angle measurement. Phase differences, the form of measurement in the hyperbolic mode, repeat along the baseline at intervals of a half-wavelength. The repetition interval distance defines an Omega lane.

A representative baseline in the Omega system is about 5,000 nm in length. At a signal frequency of 10.2 kHz, with a propagation velocity of 161,829 nm per second, the wavelength corresponds to a distance of approximately 16 nm. For a baseline of 5000 nm, then, there are approximately 315 full wavelength lanes, 16 nm in width, which are applicable to the ranging mode. Since lane widths in the hyperbolic mode are defined by intervals of a half-wavelength, the lane widths along the baseline in the hyperbolic mode are approximately 8 nm. Divergence of the hyperbolic contours causes lane width in the hyperbolic mode to increase as a function of distance from the baseline.

Each lane between transmitting stations is numbered for identification. To aid in establishing position within a lane, phase measurements are expressed as the percentage of a cycle, with each 360 degrees constituting 100 centicycles. The difference between phase measurements in centicycles is numerically equal to the percentage value of the lane defining the LOP, generally expressed as centilanes. An observed phase difference corresponds to an LOP in each lane for a given station pair (600 to 700 lanes at 10.2 kHz in the hyperbolic mode). This ambiguity is overcome by initialization of the Omega receiver to current position accurate to within ±0.5 lanewidth and continuous count of lane changes resulting from vehicle motion. For the example of 10.2 kHz, initial position must be known to within ±4 nm (±8 nm in the ranging mode).

The four frequencies used for Omega navigation (10.2, 13.6, 11.33, and 11.05 kHz) provide lane widths ranging from 6 to 8 nm in the hyperbolic mode. Additional frequencies can be formed to yield coarser lane widths by "beating" together Omega signals on two different frequencies. As an example, combined processing of 10.2-kHz and 13.6-kHz signals yields a phase difference value of 3.4 kHz, resulting in a hyperbolic mode lane width of 24 nm. The net effect is a relaxation of required initial position accuracy to only ±12 nm. Table A-1 shows the various lane widths possible with multiple frequency techniques for both hyperbolic and ranging modes of processing.

2.5 Propagation Modeling

Accuracy, and therefore practicality, of the Omega system depends on the inherent stability and predictability of the phase variations of a VLF signal over a very long propagation path. Propagation models characterizing VLF signals propagating in the earth-to-ionosphere waveguide are generally based on one of the following three approaches: hop theory, zonal harmonic series, and mode theory. Hop theory assumes that the VLF signal is composed of a ground wave in addition to a series of "hops" (or rays) generated by successive reflections between the ionosphere and the

Table A-	I. INITIAL PO	SITION ACCURA	CY DETERMINATI	AN POR DUA	L-PREQUERCY O	ND AN
Frequency 1 (kHz)	Frequency 2 (kHz)	Frequency Difference (KHz)	Lane Width	(aac)	Inita Positi Accura Requirem	1 ion acy ments
			Hyperbolic*	Ranging	Hyperbolic	Banging
10.20	1	10.20	7.94	15.88	3.97	7.94
11.05	1	11.05	7.33	14.66	3.67	7.34
11.33	1	11.33	7.14	14.28	3.57	7.14
13.60	1	13.60	5.95	11.90	2.98	5.96
10.20	13.60	3.40	23.80	47.60	11.90	23.80
10.20	11.33	1.13	71.61	143.22	35.81	762
10.20	11.05	0.85	95.20	190.40	47.60	95.20
11.05	13.60	2.55	31.73	63.46	15.87	31.74
11.05	11.33	0.28	288.98	577.96	144.49	288.96
11.33	13.60	2.27	35.65	71.30	17.83	35.66
*Along basel	ine.					

Γ

Ľ

and the second se

A CONTRACTOR

A--13

「日本市の学」の湯

earth. The zonal harmonic series approach applies rigorous mathematical analysis to the physical interpretation of hop theory. Mode theory describes the total signal at any point within the waveguide in terms of the natural or characteristic modes of propagation. The prediction models used by Omega Navigation Systems Operations Detail (ONSOD) are based on the mode theory approach. Omega signal coverage diagrams are generated by predicting signalto-noise ratio (SNR) and occurrences of excessive mode interference. Global computations of predicted phase corrections are used to develop predicted propagation corrections (PPCs), which are published in tables to provide phase corrections for various regions, times of day, and months.

The prediction models use propagation characteristics that are fairly well understood and highly repeatable. The most obvious influence on Omega signals is the effective height of the ionosphere. Increases in solar radiation advance ionization of the atmosphere, thereby effectively lowering the height of the ionosphere to about 70 km soon after sunrise; at night, the height is about 90 km. Because Omega phase is inversely proportional to phase velocity, the phase increases or decreases in step with the effective height of the ionosphere, referred to as diurnal phase shift.

The phase of the Omega signal received at a given point in space is determined by the phase velocity and distance to the source. If distance is held constant and phase velocity is increased, the previously measured phase will appear to have advanced beyond the given point in space, creating the impression that the point has moved closer to the transmitter.

Numerous other phenomena that influence signal propagation cannot be wholly taken into account because of their unpredictable nature. Geophysical parameters that affect propagation include ground conductivity, the earth's magnetic field, solar activity, latitude, and solar zenith angle. Sudden ionospheric disturbances (SIDs) are produced by solar flares. Those solar flares that generate significant X-ray radiation cause ionization changes to the ionosphere, an effect that manifests itself as an Omega phase anomaly. This effect is referred to as a sudden phase anomaly (SPA). Polar cap absorption (PCA) events are also caused by solar flares, but only those that produce significant quantities of solar protons, which concentrate at the earth's magnetic poles. These protons tend to absorb the effective energy of radio waves. However, in the case of low-frequency Omega transmissions through the polar regions, the dominant effect of the proton activity is manifested as another phase anomaly. Both SIDs and PCA events are random and unpredictable but occur more frequently during years of peak solar activity. A significant SID can cause position errors of about 2 to 3 nm only during daylight hours, whereas a severe PCA event could cause a position error of 6 to 8 nm, lasting for several days.

Another source of Omega signal error is modal interference, created by phase errors from additional ground and sky waves combining with the primary wave. This form of interference is minimized by automatic deselection of a station within a defined distance from the Omega receiver. This distance varies from 300 nm to 600 nm, depending on the equipment used. Beyond the prescribed distance, the limitation on range of ground and sky waves dilutes their impact on the primary wave.

The prediction models applied to the Omega system are based on monitor data taken for at least one year to exploit annual correlation fully. Therefore, at least this amount of time will be required following completion of the Australian station before Omega can be declared a fully operational navigation system.

2.6 Operational Characteristics

Use of all available Omega frequencies can provide greater accuracy in determination of position. Since each frequency propagates through the air with different characteristics, off-nominal conditions will not affect all frequencies the same way. Off-nominal conditions can therefore be isolated, and their effects can be compensated for, by proper filtering of multiple frequencies.

Even more important to accuracy than the number of frequencies processed is the number of Omega stations within reception range and their geometry in relation to the aircraft receiver. Hyperbolic navigation requires a minimum of three stations at all times, Whereas navigation in the ranging mode can be accomplished with only two stations and a sufficiently accurate local oscillator. The best geometry for greatest accuracy results when the LOPs intersect at right angles. This configuration provides equal cross-track and along-track accuracy. As the LOPs approach a parallel configuration, the resulting accuracy in position fixing is correspondingly degraded. When more than three Omega stations can be received, sufficient redundancy exists to provide excellent accuracy. Generally good Omega signal coverage throughout the conterminous United States provides availability of a minimum of three stations. When an aircraft encounters loss of Omega signals, preventing satisfactory Omega navigation, the area in which this condition persists is referred to as a "hole" in signal coverage.

As was mentioned earlier, the Omega transmission format repeats every 10 seconds, with each Omega station transmitting different frequencies at any given time. It would therefore require a full 10 seconds to receive all transmitted frequencies from all Omega stations. As a result, when all phase information is processed to determine position, some information is no longer current, but is up to 10 seconds old. This loss of processing time synchronization is overcome by projecting each signal forward in time in accordance with its respective position in the processing cycle. This form of processing constitutes a dead-reckoning system with ground-referenced updates. The dead-reckoning system uses aircraft heading and speed information, with the ground-referenced updates provided by the Omega system. Aircraft heading and speed information can be supplied by on-board sensors or manual data entries or can be derived from progressive Omega measurements.

When Omega measurements are used to estimate aircraft heading and speed, filter time-constants of sufficiently long duration must be employed to minimize unwanted deviations resulting from noise and other spurious activity. A common predicament encountered in establishing a filter timeconstant is that, although stability can be achieved with use of a long timeconstant, quick response to maneuvers requires a short time-constant. Since the most prevalent use of Omega is for en-route navigation with minimal heading or speed changes, a long time-constant has generally been preferred and is therefore the standard for Omega receivers. With increased use of Omega in the airborne environment, however, there is greater interest in the possible use of Omega in approach operations, where a great deal of maneuvering can occur. Acceptance of Omega for approach navigation will be predicated on the ability of the airborne Omega navigation system to respond quickly (a short filter time-constant) and accurately to course and speed changes.

Omega antennas are of either the E- or H-field type. The H-field antenna is an orthogonal loop that captures the magnetic (H-field) component of the signal. Various configurations of the H-field antenna have been designed to accommodate the many different installation requirements for both large and small aircraft. H-field antennas are desirable because of their immunity to the effects of precipitation static (P-static). However, since H-field antennas are designed for high sensitivity to magnetic fields, they are susceptible to magnetic noise.

On some aircraft, 400-Hz interference from power generation is so widespread that a suitable location for an H-field antenna cannot be found. In these cases, a capacitive (E-field) antenna is used. The E-field whip antenna is immune to magnetic noise but is susceptible to P-static.

Some manufacturers have offered custom antenna design to maximize antenna efficiency for the particular installation environment. One such design uses heading information to orient an H-field antenna electrically.

2.7 VLF Signal Processing

All Omega stations are VLF transmitters. There are additional VLF transmitters that are not part of the Omega system. The United States Navy maintains several VLF communication stations in various countries around the world to provide a worldwide military communications network. Although these stations were not intended to be used for navigation purposes, they are technically suitable for use as navigation aids. However, since the stations are not within the framework governing compliance with international navigation standards, they are used solely for secondary-mode navigation. VLF stations provide a means of navigating through regions of unacceptable Omega coverage without the necessity of reverting totally to dead reckoning. VLF can be implemented as the sole signal source or can be used in combination with Omega signal processing. In general, when the VLF mode is activated, one of the Omega frequency channels is deactivated, and the VLF signals are superimposed in a fixed sequence in the Omega transmission time slots of the deactivated Omega frequency channel. The phase of the VLF signal is initialized at the position of the receiver at the time the VLF station signal is first received. Subsequent phase measurements indicate position change.

ì.

The operational capabilities of Omega navigation systems heavily influence the potential accuracy of position fixing. Table A-2 presents some typical expected accuracies as a function of operational capability.

Table A-2. OPERATIONAL CAPABILITY VERSUS EXPECTED ACCURACY FOR OMEGA NAVIGATION SYSTEM								
Number of Stations Received	Number of Frequencies Processed	VLF Proc essing	Accuracy of Position Fixing (nm)					
0	0	x	<u><4</u>					
2	3	x	<u><</u> 2					
>4	4		1 to 1.5					
>4	3	x	1 to 1.5					

2.8 Differential Omega

Omega signal accuracy is primarily dependent on the inherent stability and predictability of the phase variations of a VLF signal over a very long propagation path. This accuracy can be improved locally by taking into account localized Omega signal phase variations, which can be determined by differencing predicted phase values with measured phase values, a technique referred to as differential signal processing.

The differential Omega concept requires local area ground monitors, which receive the standard signals from the Omega transmitters. The measured phase values are then compared with predicted values, any difference being considered a phase error. The phase error measured at the ground monitor is relayed to aircraft in the vicinity of the monitor station. The airborne Omega unit compensates for the error by applying it as a differential correction to the standard Omega signal being received directly from the Omega transmitters.

The predicted Omega phase values are computed at the monitor stations as a function of assumed signal propagation speeds and precise knowledge of the distance between the monitor station and each transmitter. The assumed signal propagation speed is a function of the propagation model used at the monitor site. The differential correction term relayed to aircraft is the difference between the predicted and measured values of Omega phase as computed at the monitor site.

Ideally, application of the differential correction term would result in a position accuracy equivalent to the accuracy with which the geographical position of the monitor site had been surveyed. A number of error sources,

A~17

however, affect the accuracy capability of differential Omega. The instrumentation error of the monitor facilities will obviously have a direct influence on the accuracy provided by differential Omega. The most significant of the error sources other than those that are equipment-related are modal interference, non-modeled variations in phase velocity, and distance displacement of the user from the monitor site. Also, differential corrections cannot compensate for the effects of lack of signals or poor geometry.

A variation of the differential Omega concept, called "gradientcorrected" differential Omega, has been suggested as a means of reducing the effects of modal interference and phase velocity variations. In this technique, the phase correction terms provided by basic differential Omega are supplemented with phase velocity (phase gradient) correction terms. Implementation of this scheme requires the use of two neighboring ground monitor stations to enable measurement of local phase velocity. Again, the computation of the correction terms is dependent on the propagation model used at the monitor station.

The accuracy improvement provided by differential corrections is most pronounced when the user is directly over the monitor site, where the correlation with the differential correction terms is at its maximum. A position accuracy of 0.1 to 0.2 nm (10) is typically achievable in marine applications when the user is within 50 nm of the ground monitor station. This accuracy then degrades with increasing distance from the station in a roughly linear fashion. Although there is insufficient evidence to prove it conclusively, it is generally believed that application of differential corrections to Omega signals will improve position accuracy even when the user is 250 nm from the monitor station. Use of differential corrections beyond 500 nm of the monitor station could actually degrade the position integrity of the standard Omega signal being received by the user.

All Omega receivers utilize some form of propagation modeling to predict signal propagation characteristics. These models can be quite complex, taking into account diurnal and seasonal phase shifts, or be as simple as predicting a constant value of phase velocity. The differential corrections are computed relative to the propagation model used by the monitor station and are therefore most effective when applied as compensation to a receiver using the same propagation model. Differences between the airborne and ground monitor propagation models will compromise the effectiveness of the differential corrections in improving Omega accuracy. One solution to this problem would be to require all airborne receivers to utilize the model used by the ground monitor stations. However, such a solution could create another problem. As propagation anomalies become better understood, prediction models could be improved. But there would be lack of incentive to improve propagation models in an environment where standardization is mandated to ensure consistency in the effectiveness of application of differential corrections. Improvements in accuracy would have to be weighed against the cost of retrofitting all existing Omega receivers with the new propagation model.

a a ser a

1. 22.

An alternative approach would be simply to relay the Omega signal received by the ground monitor station directly to the aircraft for processing by the airborne receiver. Knowledge of the exact geographical position of the monitor station would enable the airborne processor to determine phase errors with the airborne propagation model. This form of implementation eliminates potential differences between the propagation model used to determine differential corrections and the one in which the corrections are applied to compute position. Also, there would be no impediment to improving propagation models, since there would be no need for standardization. There would, however, be an impact on the processing and memory requirements imposed on the airborne Omega unit.

3. VOR

Γ

3.1 History of Development

VHF Omnidirectional Range (VOR) was developed in response to the growth in air traffic in the 1930s. The unpredictable propagation characteristics of the low- and medium-frequency range navaids in use at that time severely limited their practical service area. As aircraft began flying at higher and higher altitudes, line-of-sight distances increased, making VHF frequencies useful to over 100 miles. The only directional guidance navaid available before VOR that did not require a direction finder on board the aircraft was limited to selection of one of four courses. VOR became the United States standard in 1946 and the international standard in 1949, combining communication and navigation within one band.

3.2 Facilities

The VOR ground station transmits two 30-Hz signals: one is a constant phase omnidirectional reference signal, and the second signal varies in phase in relation to the magnetic bearing of the aircraft from the VOR station.

VOR antenna systems use Alford loops, which generate horizontally polarized signals with the same field pattern as a vertical dipole. Conventional VOR systems use four Alford loops arranged in a square. The plane containing the Alford loops is horizontal and is located a half-wavelength above a circular and conducting ground plane, or counterpoise, that is approximately 30 feet in diameter. The counterpoise also acts as the roof of the transmitter house, with the loops protected from the weather by a radome. The radome is generally hemispherical in shape; if a Tactical Air Navigation (TACAN) antenna is included in a VOR installation (a VORTAC), the radome is conical. The VOR antenna system is designed to operate in two distinct modes, corresponding to the two distinct signals radiated -- the carrier mode and the sideband mode, described in the following section.

3.3 Signal Characteristics

VOR operates in the VHF band of 108 MHz to 118 MHz, divided into 50 kHz channels. Originally, the channel width was 100 kHz. The 30-Hz reference signal is generated by frequency modulation of a 9.960-kHz subcarrier signal, which amplitude-modulates the radio frequency (RF) carrier signal. This reference signal is radiated when the antenna is operated in the carrier mode, where all four loops are simultaneously driven in phase with carrier frequency currents, the result being an omnidirectional pattern. The variable phase signal is generated when the antenna is operating in the sideband mode, which causes each diagonal pair of loops to be excited so that at any instant of time the horizontal plane pattern of each pair of loops above the counterpoise is a figure eight. The relative phase between the pairs of loops is such that the combined effect of these phases in space produces a single figure-eight azimuthal pattern rotating at 30 revolutions per second. Combination of the carrier field with the total sideband field radiates a cardioid pattern that rotates at 30 revolutions per second, generating a 30-Hz sine wave at the output of the airborne receiver. The 30-Hz reference signal and the rotating pattern sine wave are synchronized so that they are exactly in phase when viewed from magnetic north. Therefore, the phase between the reference and the rotating cardioid varies directly with the bearing of the aircraft.

VOR signal reception depends on the standard service volume of the class of facility used and can range in radial distance from 40 nm to 130 nm from the VOR, depending on altitude, as shown in Figure A-6. The primary difference between Figure A-6 and the currently existing specifications for standard service volumes as defined in FAA Advisory Circular (AC) 00-31 (U.S. National Aviation Standard for the VORTAC System) is the FAA-proposed additional stipulation of a defined coverage area between 1,000 and 14,500 feet altitude for the high-altitude station.

3.4 Signal Processing

The airborne receiver detects the VOR signal and passes it through an amplitude modulation (AM) detector. The AM detector distinguishes the identity tone and any voice frequencies present and broadcasts them through the aircraft audio system. These components of the signal are removed via a 30-Hz filter, and the resulting 30-Hz amplitude modulation produced by the rotating pattern of the VOR is fed to phase comparison circuitry. The original signal is also passed through a 9.960-kHz filter, a limiter to remove 30-Hz amplitude modulation, and a frequency modulation detector, which then outputs the 30-Hz frequency-modulated reference. After one more stage of filtering, the reference frequency is compared with the variable phase signal in the phase comparator. The output of the phase comparator is the bearing of the aircraft with respect to magnetic north.

3.5 Operational Characteristics

Use of VHF eliminates sky wave contamination of VOR signals and prevents interference from stations beyond line of sight. The most significant error sources are site errors and errors in measuring phase shifts at 30 Hz.

3-20







Standard Low-Altitude Service Volume Standard Terminal Service Volume

I.

E.

1

[]

[]

.....

Source: FAA Order 9840, Reference 13.

- Contract of an in Buran

Figure A-6. VOR STANDARD SERVICE VOLUMES

Site errors are caused by multipath effects at the ground station, resulting from signal reflections off trees, buildings, and other scattering objects near the transmitter. The multiple signal paths created by the reflections combine with the desired signals to alter the phase difference measured at the aircraft. This results in an error in the VOR bearing indicated by the airborne receiver. As the aircraft moves, the phase difference between the desired signal path and the reflected paths changes. The phase difference error fluctuates with a measurable frequency depending on the rate of change of path difference. This phenomenon, referred to as scalloping, becomes more pronounced with decreasing distance from the VOR. Coverage of VOR is restricted because of line-of-sight limitations. The coverage pattern of a VOR is additionally restricted by a limited elevation-angle of transmission, above which no VOR signal is available. VOR signals currently are guaranteed to be available within 60 degrees of elevation from the ground plane of the transmitter. In practice, however, signals can generally be received up to an elevation angle of nearly 80 degrees. The airspace directly above the VOR, in which no signal exists, defines a cone, the apex of which is at the transmitter with a central angle of 60 degrees. This cone, generally referred to as the cone of confusion, is illustrated in Figure A-7. As the figure shows, the effective area of the cone increases with altitude. At 5,000 feet altitude, the area of concern above the VOR has a radius of 0.4 nm. At an altitude of 30,000 feet, the radius is 2.5 nm.



Figure A-7. VOR CONE OF CONFUSION

4. DME

4.1 History of Development

Distance measuring between aircraft and ground stations is provided through Distance Measuring Equipment (DME) and Tactical Air Navigation (TACAN). Since TACAN uses the same pulses and frequencies for distance measurement as are used by the standard DME system, all references to the operational design and performance of DME correspondingly apply to TACAN. The history of DME dates back to the Rebecca-Eureka pulse ranging system developed during World War II, which operated at 200 mHz. By international agreement in 1946, DME was assigned for operation in the 960- to 1,215-mHz L-band. The exact frequencies and pulse techniques were not established until 1959, however.

4.2 Facilities

Ground-based DME consists of a receiver-transmitter and an antenna. DME is typically collocated with some other navigational aid, including the following station types:

- VOR/DME
- ILS/DME
- TACAN
- VORTAC

VOR/DME is a DME station located at the same site as a VOR station. ILS/DME denotes the presence of a DME at the site of an instrument landing system (ILS). TACAN provides both azimuth and distance information to suitably equipped aircraft. Although the azimuth signal from a TACAN is not the same type as from a VOR, the distance measuring function of TACAN is equivalent to DME. A VORTAC facility consists of a VOR station collocated with a TACAN station.

4.3 Signal Characteristics

The receiver-transmitter channels of DME stations located with a VOR station are paired with the respective VOR channel in accordance with a published table. The air-to-ground transmitted frequencies of DME are from 1,025 to 1,150 MHz; the ground-to-air transmitted frequencies are from 962 to 1,213 MHz. The VOR frequencies with which DME frequencies are paired are between 108 MHz and 118 MHz. VOR frequencies are divided into 100-kHz-wide channels, resulting in 100 channels. There are 126 DME channels between 1,025 and 1,150 MHz, spaced 1 MHz apart, allowing frequency pairing with the 100 VOR channels and leaving 26 unpaired DME channels. When the VOR channel width was changed to 50 kHz, an additional 100 VOR channels were made available, bringing the total to 200. As a result of this action, additional DME channels were made available by the designation of X and Y channels. The frequency spacing between DME channels was not changed.

X channels correspond to the original 100-kHz-spaced VOR frequencies and the corresponding DME frequencies. Spacing of the ground reply pulse for X channels is 12 μ s. VOR channels that are offset 50 kHz from the X channels are designated Y channels; they have a ground reply pulse spacing of 30 μ s. The airborne interrogating pulse-pair spacing is 12 μ s on X channels and 36 μ s on Y channels. With use of X and Y channels, 252 DME channels are available; 200 of these are paired with the available VOR frequencies. The transmit and receive frequencies of any one channel are separated by 63 MHz.

DME ground stations are capable of handling approximately 100 aircraft interrogations simultaneously. When more than 100 aircraft interrogate the ground station, the ground station limits its sensitivity and replies only to the 100 strongest interrogations. Most airborne DMEs are designed to operate down to a reply efficiency of 50 percent, reflecting the situation where the DME receives replies to only half of its transmitted interrogations.

The ground station continuously transmits a squitter signal (filler signal) of 2,700 pulse pairs per second (pp/s), with an identification code signal of 1,350 pp/s at 30-second intervals. When interrogated by the airborne DME pulse pair, the ground station transmits a reply pulse pair that replaces a squitter pulse pair 50 µs after interrogation. Replies are considered valid by the airborne receiver when they occur at approximately the same time after every transmitted pulse pair. The pulse rate of the airborne DME transmissions is varied randomly (jittered) so that no two airborne DMEs are transmitting at the same rate. This jitter technique prevents transmitted pulse pairs from another aircraft DME being mistaken for reply pulses.

The airborne DME decreases the frequency of interrogations when it has established the validity and regularity of the reply pulses. These variations in level of frequency of interrogation are referred to as either the search mode or track mode -- the track mode indicates less frequent interrogation. Use of the track mode relieves the loading of the ground station to allow service of more aircraft.

The identification signal of 1,350 pp/s can be converted to an audio signal so that the pilot can monitor the signal to confirm that the airborne DME is tracking the station the pilot has selected.

4.4 Signal Processing

The DME system employs a pulse-ranging technique whereby an airborne unit, referred to as an interrogator, transmits a pair of pulses that the DME ground transponder retransmits after a fixed time delay. Paired pulses are used to reduce interference from other pulse systems. The peak pulse power of the airborne unit is about 50 watts to 2 kilowatts. The DME ground station (or transponder) receives these pulses and, after a fixed delay of 50 μ s, retransmits them back to the aircraft on a frequency 63 MHz below or above the airborne transmitting frequency. The peak power of the ground station is between 1 and 20 kilowatts.

The airborne interrogator automatically compares the elapsed time between transmission and reception, subtracts the fixed delay of 50 μ s, and displays the result in terms of nautical miles. The range thus computed is slant range, the actual distance between the aircraft and the DME ground station, rather than ground distance. If the aircraft is more than one mile

from the station for each 1,000 feet of altitude, the difference between slant range and ground distance is negligible. Correction of slant range to obtain ground distance is a function only of altitude above the station.

4.5 Operational Characteristics

If the DME signal is lost after the track mode has been established, the DME retains the last value of computed distance for a short period of time. If the signal is not reacquired within this period, the airborne DME enters the search mode.

The most potentially serious problem encountered when DME is used is false lock-on. This occurs when the airborne DME acquires and tracks transponder replies to a multipath signal instead of the direct signal. "Confirm/track" circuits in the airborne unit eliminate the possibility of false lock-on by scanning the interval just before the tracked pulse pair to confirm that the tracked signal is not an echo. The magnitude of error caused by false lock-on can be several miles. Multipath and siting errors other than false lock-on are small and nearly random.

5. GPS

5.1 History of Development

The Navigation System using Timing and Ranging (NAVSTAR) Global Positioning System (GPS) is currently being developed and evaluated as an advanced satellite-based positioning and navigation system. Although it was originally scheduled to be operational in 1982, current estimates reflect an initial 18-satellite deployment by 1987.

The concept of NAVSTAR GPS, being developed by the Department of Defense, evolved from the 1973 merger of the Navy's Time Navigation (TIMATION) Program and the Air Force 621B Project. Both projects were established in the mid-1960s to investigate satellite-ranging techniques to satisfy military navigation requirements. The Air Force used ground stations to simulate satellite signals, whereas the Navy actually launched satellites.

The Navy TIMATION satellites provided more accuracy and greater coverage through use of ranging techniques than was possible through use of previously developed satellite navigation systems, such as TRANSIT, which were based on frequency measurements. TRANSIT is the Navy Navigation Satellite System (NNSS or NAVSAT), currently used primarily for navigation of submarines and surface ships, both civil and military.

When the Department of Defense's GPS Joint Service Project Office was created, the Air Force was given the responsibility of program management through the Space Division in El Segundo, California. The Navy established the Navigation Technology Program to provide technical support and design of the navigation technology satellites (NTSs) currently used to evaluate

the components and systems being developed for the navigation development satellites (NDSe) of NAVSTAR GPS.

5.2 Facilities

NAVSTAR GPS involves the following segments of operation, each dependent on the others to provide precision worldwide navigation capability:

- · Ground control segment
- Space system segment
- · User segment

The ground control segment tracks the satellites, monitors the navigation signals, and provides correction terms to the satellites. The space system segment includes the satellites themselves and the navigation signals they transmit. The user segment comprises numerous receivers representing diverse applications of the navigation data.

5.2.1 Ground Control Segment

The primary function of the ground control segment is to maintain the precision of the navigation data supplied to the user. This function is performed via a network of four monitor stations (MSs), an upload station (ULS), and a master control station (MCS). The four unmanned MSs are remotely located from but directly linked to the MCS. Each MS receives and decodes satellite navigation data and collects local meteorological data. Data from each MS are relayed to the MCS, where they are processed to determine satellite ephemeris and clock errors. The MCS then generates correction terms to compensate for identifiable biases and provides tham to the ULS for upload transmission to the satellite.

The ability of GPS to provide highly accurate navigation capability depends on precision timing and frequency control. The importance of precision timing in terms of navigation accuracy is illustrated by the fact that one nanosecond of time error corresponds to 0.3 meter (0.984 foot) of distance error (assuming the speed of light to be 3×10^6 m/s).

The MCS maintains GPS system time through the use of cesium clocks. Because the precision of GPS system time must be continuous, it is therefore not offset by leap-year seconds, which cause periodic adjustments to universal coordinated time (UTC). The MCS monitors the satellite clocks to determine their degree of synchronization with GPS system time. Clock corrections are uploaded by the ULS to the satellites and become part of the navigation message transmitted by the satellite to users.

5.2.2 Space System Segment

Great effort has been expended in the search for an optimal 18-satellite configuration. The most current, but not necessarily final, choice uses six orbital planes, with three satellites in each orbit. Each satellite orbits at 10,900 nm above the earth in a 12-hour period. The ascending nodes of

each orbit are equally spaced by 10 degrees in longitude around the equator. Each orbital plane is inclined by 55 degrees with respect to the equatorial plane, and the three satellites in each orbit are equally spaced by 120 degrees. The satellites are designed for an operating life of five years.

The four atomic clocks contained in each satellite are accurate to one second per 30,000 years. Two levels of navigation accuracy are provided by the GPS signals transmitted from the satellite. The greatest degree of accuracy is provided through acquisition of the precision code (P-code). Access to this code is limited to military users. A second code, the coarse acquisition (C/A) code, is less accurate; it will be made available to all users. The accuracy capability of the C/A code may be further degraded for reasons of national security.

5.2.3 User Segment

Several configurations of GPS user equipment are currently being developed and evaluated, as follows:

•	X-set	•	FAA Experimental Dual
•	Y-set		Channel Receiver
		•	DOD Phage II liger

- Z-set Equipment
- Manpack
- GPSPAC

The X-set was developed as a means of evaluating the speed and accuracy of GPS position-determination under the most severe jamming and dynamic conditions that might be experienced on high-performance military aircraft. The X-set uses four receiver channels, which enable it to process data from four satellites simultaneously. It receives GPS signals on two frequencies, allowing measurement of propagation errors.

The Y-set is more compact than the X-set and costs less. It has only one receiver channel and therefore sequences between satellites. Performance of the Y-set is comparable to that of the X-set except under conditions of high dynamics.

The Z-set is a low-cost, compact configuration that sacrifices high accuracy and dynamic response. As with the Y-set, the Z-set sequences between satellites, using only one receiver channel. But unlike the Xand Y-sets, which process two frequencies, the Z-set processes only a single frequency. Because the Z-set is able to acquire only the C/A code, it provides less accuracy than the X- and Y-sets, which can also acquire the P-code.

Although Manpack does not require high dynamic capability, it does require high-accuracy, small-size, low-weight, low-power, and antijam capabilities. Manpack is essentially the Z-set with the addition of dual frequency and P-code processing. GPSPAC is intended primarily for use in satellites and shares the design features of the Z-set and Manpack.

An experimental dual-channel GPS receiver is providing a means for the FAA to develop engineering requirements and cost estimates for a lowcost general aviation receiver. The Phase II user equipment is being designed and evaluated in preparation for production of equipment for the Department of Defense.

The many possible applications of NAVSTAR GPS include the following:

- Strategic aircraft and cruise missile navigation
- Battlefield operations
- Submarine navigation (update of INSs while surfaced)
- Tactical navigation
- Aircraft carrier navigation
- Harbor and sea lane operations
- Maritime shipping
- Search and rescue operations
- Spacecraft operations
- Surveying
- Oil exploration
- Air traffic control
- Civil air navigation

5.3 Signal Characteristics

5.3.1 Modulation

The signal transmitted by the satellite to the user is modulated by a sequence of bits in a pattern unique to each space vehicle. These patterns are referred to as pseudorandom noise (PRN) codes. PRN codes exhibit the characteristics of random thermal noise until the pattern is decoded and the receiver locks onto it. The signals are coded for two reasons. The first is to allow denial of access to the signal by withholding details of the code patterns. The user receiver must be able to match the incoming code to establish a lock-on. The second reason for using a code is that when the sequence generated by the user receiver is matched with the incoming signal, measurement of the phase shift required to maintain match of the codes provides a measurement of the transit time of the navigation signal.

Two forms of PRN codes are used to modulate the NAVSTAR GPS navigation signal. The first and most precise PRN code is the precision code (P-code). The precision navigation accuracy attained from use of this code is made

λ-28

possible by its transmission rate of 10.23 megabits per second, or 97.8 nanoseconds per bit. At the speed of light $(3 \times 10^{6} \text{ m/s})$, each bit of the P-code corresponds to a distance resolution of 29.34 meters (96.8 feet). However, the repetition rate of the P-code pattern is only once per seven days, making it effectively impossible to lock onto it without knowledge of the code pattern. Even with such knowledge, aiding is generally required for rapid convergence on pattern synchronization. For this reason, a second PRN code is used -- the coarse acquisition, or clear access, (C/A) code. The C/A code repeats itself every millisecond, enabling easy acquisition and lock-on. The transmission rate of the C/A code is one-tenth that of the P-code, resulting in a distance resolution of 293.4 meters (968 feet). By locking onto the C/A code, the receiver can extract the navigation information contained in the signal. Included in the navigation message is a parameter referred to as the handover word (HOW), which is used in transferring from the C/A code to the P-code.

5.3.2 Format

The navigation message contains the handover word, data relating to space vehicle (SV) status, clock correction parameters, corrections for signal propagation delays, the ephemeris of the SV whose signal is being received, and almanac information that defines the approximate ephemerides and status of all other space vehicles.

The navigation message is contained in a 1,500-bit, 30-second data frame. The data frame is subdivided into five subframes, each of which contains 10 words, each word 30 bits in length. The first two words of each subframe are a telemetry word (TLM) and the C/A-to-P-code HOW. The TLM serves as an identifier to facilitate pattern synchronization and contains information primarily for use by the ground control segment, relating to upload operations. The HOW contains information that allows transition from the C/A code for users with access to the P-code. The remaining eight words in each subframe constitute a block of data, as illustrated in Figure A-8.



*TLM - Telemetry word.

HOW - Handover word.

Figure A-8. DATA FRAME FORMAT FOR GPS

Block 1, corresponding to subframe 1, contains the SV clock correction parameters and ionospheric propagation-delay model parameters. Blocks 2 and 3 contain the ephemeris of the SV. Block 4 is reserved for special alphanumeric messages, and Block 5 contains almanac data for all space vehicles. Only a single SV almanac is available per data frame. Therefore, it takes 18 frames, nine minutes, for the receiver to cycle through the almanac for a system complement of 18 satellites. The almanac information is required for use in signal acquisitions; the other information is necessary for accurate processing of the navigation signal.

5.4 Signal Processing

As previously mentioned, each satellite generates a particular C/A code. The user receiver generates replicas of these codes and crosscorrelates the locally generated signals with the signals received from the satellites. A tracking loop is used to establish synchronization between the two signals. Tracking error is indicated by the output of the correlator, with minimum error representing the peak of the autocorrelation function. Maximum code tracking is achieved by advancing and retarding the phase of the locally generated signal with respect to user clock timing. If the user clock is synchronized to GPS system time, the time delay associated with the phase difference measured at a common clock time between the local and received signals indicates the travel time of the signal between satellite and receiver. The travel time measured corresponds with satelliteto-receiver range.

In reality, the quality of the user clock is not high enough to maintain synchronization to GPS system time. Therefore, the time, or range, measurement includes clock bias errors and is referred to as a pseudorange measurement.

By measuring range from any individual satellite, the user places himself on a sphere centered at that satellite. Two range measurements create two spheres, the intersection of which defines a circular LOP. The intersection of a third sphere under ideal circumstances would provide a single point of intersection and thereby a relative position fix with respect to the center of each sphere. Knowledge of the earth-referenced X, Y, and Z position coordinates of each sphere center -- that is, of each satellite -would result in a three-dimensional fix of user position.

Satellite position can be accurately computed from the ephemeris parameters included in the navigation message of the received GPS signal. Data contained in the message are accessible once the format identifiers have been detected after track lock-on.

Although three satellites generate three spheres, they do not provide an accurate three-dimensional position fix. The clock biases included in the pseudorange measurements introduce errors into the navigation computations. Each pseudorange measurement is therefore composed of the three position parameters and a clock bias term. The existence of four unknowns requires four pseudorange measurements to obtain four equations, which can then be solved for the position in three dimensions and the user clock offset.

Before the four equations are simultaneously solved, however, each pseudorange measurement is compensated for ephemeris errors, propagation delay, and satellite clock offset. The terms providing these corrections are included in the navigation message. Additional accuracy can be achieved when the accumulated phase difference between the received signal and the locally generated reference code is measured over a fixed time interval. This delta range measurement has the accuracy and resolution of a fraction of the carrier wavelength.

The GPS signal from each satellite is transmitted over two carrier frequency links, both of which are in the lower microwave region called the L-band. Link 1 (L1) has a center frequency of 1,575.42 MHz, and Link 2 (L2) has a center frequency of 1,227.60 MHz. When the two frequencies are processed, propagation errors that vary with frequency can be identified, and appropriate corrections can be generated and applied. L1 currently carries both the C/A code and the P-code, whereas L2 carries only the P-code. Inability to process the C/A code on two frequencies prevents the civilian user receiver from determining propagation errors. However, propagation corrections available in the navigation message do improve the accuracy of the singlefrequency signal processing.

5.5 Operational Considerations

It was stated earlier that four range measurements are necessary to obtain a three-dimensional position fix. This requirement for a minimum of four satellites to be visible at all times throughout the world is difficult to satisfy with only 18 satellites. Failure of a single satellite would result in a large area of the United States not having coverage by four satellites during certain times of the day. As an alternative to the severe degradation of accuracy resulting when clock biases cannot be accounted for via the fourth range measurement, an altimeter can be used to supply the Z (vertical) component of the range equations. Any inaccuracies in the vertical measurement will translate into lateral and longitudinal position errors. Although barometric altimeters provide satisfactory relative accuracy, they are a poor substitute for the fourth satellite in terms of absolute position measurement. The usefulness of altimeter aiding is further limited because of the geometry between the satellite and the user. Studies have indicated that the altimeter measurement and one of the satellite range measurements would generally lie in the same plane with the user, a situation seriously diluting the effectiveness of the measurement.

6. DOPPLER NAVIGATION

6.1 History of Development

Doppler navigation was first developed for use in long-range bombers and other military aircraft around the mid-1950s. By 1962, several international airlines were using Doppler navigation systems in over-water navigation and in other remote areas where VOR/DME coverage was not provided.
These early Doppler navigation systems do not provide some of the more convenient man-machine interfaces provided by more recent over-water navigation systems such as INS and, later, Omega. For example, the early Doppler systems do not use great-circle course generation and do not provide a display of current latitude and longitude. Doppler navigation systems under current development provide both great-circle route generation capability and a display of current latitude and longitude.

6.2 System Overview

A Doppler navigation system is a self-contained system that is used to navigate between position fixes. A Doppler radar is used to measure the aircraft's velocity vector relative to the aircraft body-axes. This velocity vector, together with an input of heading data, is then used for navigation from waypoint to waypoint.

The complete Doppler navigation system includes the following:

- Doppler radar
- Heading reference system
- Attitude reference system

The Doppler radar provides a measure of the aircraft's velocity vector in three dimensions. A simple attitude reference system is used to resolve the velocity vector into locally level coordinates. This velocity vector is used in conjunction with the heading provided by the heading reference system. The heading reference system is the limiting element in the crosstrack accuracy of a Doppler navigation system.

A Doppler navigation system provides current position and cross-track deviation for use by displays and the autopilot. It also provides a display of drift angle, cross-track distance, and ground speed.

6.3 System Inputs

The Doppler navigation system requires inputs from a Doppler radar, a heading reference system, and an attitude reference system. The operation of each of these is discussed below.

6.3.1 Doppler Radar

The Doppler radar is used to measure the aircraft's velocity vector. The operation techniques and principles are discussed below.

6.3.1.1 Doppler Radar Operation

The Doppler radar comprises a radar transmitter, a receiver/processor, and an antenna. The transmitter antenna directs toward the ground two to four radar beams, which are scattered by the ground surface. Some of the scattered radar energy that is reflected back toward the aircraft is then detected by the receiver antenna. The detected signal includes a frequency shift, called Doppler shift, which is proportional to the aircraft velocity relative to the ground surface. The receiver/processor then measures this Doppler shift in each beam and computes the corresponding component of velocity. The geometric relationship existing between the radar beams and the aircraft fuselage is then used to resolve these velocity components into the aircraft body-axis coordinates.

Since three velocity components are required to determine the aircraft's velocity in three-dimensional space, at least three radar beams are required. Two radar beams are sufficient only if barometric altitude rate is used to supply a third velocity component. A variety of beam configurations are possible. A Janus configuration directs radar beams both forward and rearward with respect to the aircraft and may use three or four beams. A non-Janus configuration uses two beams and directs both either forward or rearward. The non-Janus configuration requires barometric altitude rate as the third velocity component input.

6.3.1.2 Doppler Radar Principles

The backscatterer radar energy is reflected from discrete surfaces on the ground (e.g. or waves, rocks, and leaves). The total signal received at the Doppler rau ceiver antenna is composed of many separate reflections from these . es. These separate signal components have different amplitudes and phases, because of the randomly positioned reflecting surfaces. In addition, the angle of incidence of the various signal components varies slightly over the finite width of the radar beam. Since the relative velocity is dependent on the angle of incidence, the signal components have slightly different Doppler shifts. The resulting frequency spectrum of the received signal is roughly Gaussian. The mean frequency is estimated by the receiver/processor and compared with the transmission frequency to obtain the Doppler shift.

The ground velocity component corresponding to the Doppler shift in a given Doppler radar beam is determined by the following equation:

 $V_{\rm B} = \frac{CV}{2E}$

where C is the speed of signal propagation (i.e., speed of light), \vee is the Doppler shift, and f is the frequency of transmission. Two frequencies, 8.8 GHz and 13.325 GHz, have been allocated for Doppler navigation operation by U.S. and international agreement. The factor of 2 in the above relationship reflects the concept that each radar beam experiences two Doppler shifts, which are additive, during the radar signal's round trip from the transmitter antenna to the ground and back to the receiver antenna. The first Doppler shift is due to the relative velocity between the transmitter and the ground. The second shift, which is added to the reflected signal, is due to the relative velocity between the ground and the receiver antenna.

6.3.2 <u>Heading Reference System</u>

The heading reference system provides a heading input to the Doppler navigation system. Because the Doppler navigation system accuracy is principally dependent on the accuracy of the heading reference, the heading reference system is required to be highly accurate. A flux-valve compass is commonly used to measure the nagmetic heading of the aircraft. The earth's magnetic field is passed through a three-phase coil fixed to the aircraft frame. The coil senses the components of the earth's magnetic field, which are then resolved to determine the aircraft's magnetic heading. The measured heading is generally gyro-stabilized to enhance the short-term response.

6.3.3 Attitude Reference System

The attitude reference system provides pitch and roll information to the Doppler navigation systems. Two different methods can be used:

- Body-fixed
- Stable-mounted

The distinction between the two methods lies in the manner in which the Doppler radar antenna is mounted to the aircraft.

The body-fixed antenna is mounted directly to the aircraft frame. The beam geometry is therefore fixed with respect to the aircraft body-axes. The attitude reference system provides pitch and roll data that are used to resolve the velocity vector into locally level coordinate frame components.

A stable-mounted antenna is controlled relative to the pitch and roll axes so that the beam geometry remains fixed with respect to a locally level coordinate frame, regardless of aircraft attitude. The orientation of the antenna is maintained by pitch and roll commands supplied by the attitude reference system.

With either method, the attitude reference system is not required to be highly accurate. The navigation error due to errors in the pitch and roll inputs is generally small. For example, the velocity error of a typical Janus configuration is only .014 percent per degree of error in pitch. The effect of aircraft roll angle on the navigation accuracy is also not considered to be significant, since the average roll for a typical flight tends to be zero. For this reason, some Doppler navigation systems only consider the effect of pitch and disregard roll effects. A low-cost, low-quality attitude reference system is typically used and is not considered to be a major component of the Doppler navigation system.

6.4 Navigation Processing

The navigation computer uses the inputs from the Doppler radar, the heading reference, and the attitude reference to navigate between position fixes. The Doppler radar provides the aircraft groundspeed vector in bodyaxis coordinates. The attitude input is then used to resolve the velocity

vector into suitable navigation coordinates (e.g., along-track and crosstrack components of the locally level coordinate frame). This velocity vector defines the drift angle with respect to the aircraft body axes. By adding this relative drift angle to the true heading, the actual track angle is computed. Comparing the actual track angle with the desired track angle produces the track angle deviation, which can be used as a steering input to the autopilot to maintain track. The magnitude of the velocity vector is the groundspeed. The groundspeed is then integrated to provide distance displacement information.

6.5 Operational Characteristics

Doppler navigation systems in current civil use were developed in the late 1950s to early 1960s for navigation in areas where VOR/DME coverage was inadequate or non-existent. These Doppler navigation systems use a constant magnetic course generation method and do not provide a display of current latitude and longitude. In over-water environments such as North Atlantic tracks, which are great-circle tracks and which require position reporting in terms of latitude and longitude, current Doppler navigation systems are not convenient. However, Doppler navigation systems under current development use great-circle route generation and provide a display of current latitude and longitude, thereby providing a more convenient manmachine interface.

The characteristics of the detected return Doppler signal are not the same over water as they are over land. Current systems provide a land-sea switch that must be appropriately selected. If the switch is inadvertently left in the wrong position, the system will provide erroneous navigation data.

Another operational deficiency in over-water navigation applications is the potential for total signal loss when the water is very calm. Although total signal loss renders Doppler unsuitable for navigation, such situations occur infrequently.

7. CONVENTIONAL INS

7.1 History of Development

Conventional inertial navigation system (INS) techniques have evolved from principles used in aircraft gyro instruments and the marine gyrocompass. The earliest applications of INS techniques were for guidance in ballistic missiles and for ship navigation in the mid-1950s. The first INS aircraft applications were for navigation in various military aircraft, beginning in the late 1950s. The first civil airborne applications were for overwater navigation in the new wide-body "jumbo jets" introduced in the late 1960s. Conventional INSs were selected over Doppler navigation systems because of their better accuracy capability, more convenient man-machine interfaces, and provision of heading and attitude data.

7.2 System Overview

A conventional INS is another self-contained navigation system. Accelerometers are used to measure the aircraft's acceleration vector. The measured acceleration is then integrated twice to compute the aircraft's velocity vector and position displacement. By relating the position displacement to initial position, the INS determines current position.

The accelerometers are mounted on a platform that is so mechanized that the accelerometer axes are aligned with the axes of the local coordinate frame, regardless of the aircraft's orientation. The platform orientation is controlled by gyros to provide an angular reference. The navigation accuracy of the INS is highly dependent on the accuracy of these gyros.

The initial orientation must be established while the aircraft is stationary on the ground prior to flight. It is at this time that the pilot manually enters initial position. This mode of operation is called the initial alignment mode or, simply, alignment.

7.3 Platform Orientation

Gyros measure angular deviations with respect to inertial space. The local coordinate frame at a given location on the earth continually rotates with respect to inertial space because of earth rotation. Also, at any instant of time, the locally level geographic coordinate frames at different locations on the earth are not coincident, since the different "up" axes are not parallel. The INS must compensate for these variations between inertial space and the locally level geographic coordinate frame in order to use gyros as an angular reference in aligning the platform with the local coordinate frame. The compensation rates are computed relative to two effects:

- Earth rotation rate
- Aircraft transport rate

Two different methods have been used in mechanizing the platform:

- North-pointing method
- Wander-azimuth method

Each of these methods computes the compensation terms in slightly different ways and will be discussed separately.

7.3.1 North-Pointing Method

The platform axes in a north-pointing INS are parallel to the east, north, and "up" axes, respectively, of the locally level geographic coordinate frame. The input axes of the accelerometers (the axes along which the accelerations are measured) are parallel to the platform axes. The gyro input axes (those about which the gyros measure angular deviations) are also parallel to the platform axes.

The earth rotates at an angular rate of 15.041 degrees per hour with respect to inertial space. A portion of the earth rotation rate is detected by the north axis and the "up" axis (also called the azimuth axis) gyros, as follows:

 $\omega_{north} = \Omega \cos \lambda$

I

 $\omega_{azimuth} = \Omega \sin \lambda$

where Ω is earth rate (15.041 degrees per hour), ω is the component of earth rate detected by the gyros, and λ is the latitude. No component of earth rate is detected by the east axis gyro, since it is perpendicular to the earth's spin axis.

The aircraft's transport rate can be thought of as a rotation about the earth's center of the "up" axis with respect to inertial space as the aircraft moves from point to point over the earth. The "up" axis rotates at an angular rate of V/R with respect to inertial space, where V is the aircraft's velocity and R is the radius of the earth. The north-south and east-west components of aircraft transport will be discussed separately.

North-south transport causes an angular rate about the east gyro axis as follows:

 $\omega_{\text{east}} = \frac{\text{Vn}}{\text{R}}$

where Vn is the north-south component of aircraft velocity. The north and azimuth gyros do not measure an angular rate, where these axes are perpendicular to the rotation axis.

East-west transport causes angular rates, which are measured by both the north and azimuth axis gyros as follows:

 $\omega_{\text{north}} = \frac{Ve}{R}$ $\omega_{\text{east}} = \frac{Ve}{R} \tan \lambda$

where Ve is the east-west component of aircraft velocity.

The combined effects, then, of both earth rotation and aircraft transport are obtained by summing the individual effects to obtain the following equations:

 $\omega_{\text{east}} = \frac{Vn}{R}$ $\omega_{\text{north}} = \Omega \cos\lambda + \frac{Ve}{R}$ $\omega_{\text{azimuth}} = \Omega \sin\lambda + \frac{Ve}{R} \tan\lambda$

These terms are continually computed to complement the gyro measurements and are called the compensation rates. The compensation rates provide the interassociation between the local coordinate frame of the platform and inertial space.

North-pointing INSs are not suitable for navigation at latitudes above 80° . The azimuth-axis compensation rate ($\omega_{azimuth}$) is prohibitively large near the polar regions. For this reason, north-pointing INSs are not widely used.

7.3.2 Mander-Azimuth Method

The problem of excessive componention rates near polar regions is solved by using the wander-asimuth method, in which the platform is allowed to assume an arbitrary azimuth angle. The azimuth-axis compensation rate $(\omega_{azimuth})$ is not applied to the platform. Instead, this term is integrated to determine the arbitrary azimuth angle, called the wander angle or alpha angle (α). The platform azimuth axis is designated the "Z" axis in a wander-azimuth INS. The two other axes, which are rotated about the "Z" axis with respect to true north by the angle α , are called the "X" and "Y" axes respectively. The compensation rates in a wander-azimuth INS are derived from the compensation rates computed for a north-pointing mechanization by resolving them as follows:

$$\begin{split} & \underset{\mathbf{w}_{\mathbf{x}}}{\overset{\mathbf{w}_{\mathbf{e}ast}}{\overset{\mathbf{w}_{\mathbf{e}ast}}{\overset{\mathbf{v}_{\mathbf{n}\alpha}}{\overset{\mathbf{v}_{\mathbf{n}}}{\overset{\mathbf{w}_{\mathbf{n}}}{\overset{\mathbf{v}_{\mathbf{n}}}}{\overset{\mathbf{v}_{\mathbf{n}}}{\overset{\mathbf{v}_{\mathbf{n}}}}{\overset{\mathbf{v}_{\mathbf{n}}}{\overset{\mathbf{v}_{\mathbf{n}}}}{\overset{\mathbf{v}_{\mathbf{n}}}}{\overset{\mathbf{v}_{\mathbf{n}}}{\overset{\mathbf{v}_{\mathbf{n}}}}}}}}}}}}}}}}}}}}}}}}$$

The wander-azimuth method is the most widely used mechanization in current conventional INSs.

A variation of the wander-azimuth method intentionally rotates the platform about the "Z" axis at some known rate ($\dot{\alpha}$). This rate is added to the "2" axis compensation term, and the sum is integrated to compute α . Each gyro input axis alternately points north and then south, thus cancelling some of the adverse effects of gyro drift.

7.4 Platform Mechanization

The preceding section describes methods of attaining the desired platform orientation. This section describes the mechanical implementation used to achieve the desired platform orientation. The platform mechanization comprises the following:

- Gimbal system
- Angle-measuring gyros

The gimbal system provides the platform with total angular freedom about the aircraft body-axes. The gimbal system includes gimbal motors for

positioning the platform and gimbal resolvers to measure aircreft attitude and heading. The angle-measuring gyros are used to measure angular deviations of the platform from the desired orientation. These measurements are used to position the platform properly with the gimbal motors. The gyros also include gyro-torquers, which are used in providing earth-rate and transport-rate compensation.

7.4.1 Gimbal System

The platform is mounted to the aircraft frame with a multi-order gimbal system. Because it is simple and its operation is basic to all higher-ordered gimbal systems, a third-order gimbal system will be discussed. However, a fourth-order system is typically used to avoid gimbal-lock, which occurs in third-order systems when the aircraft is in a 90° pitch attitude.

The platform is pivot-mounted to the pitch gimbal along the azimuth axis. The pitch gimbal is then pivot-mounted to the roll gimbal along the pitch axis. The roll gimbal, finally, is pivot-mounted to the aircraft frame along the roll axis. Torque motors are used at the pivot mounts to rotate the platform about each axis independently. The torque motors are then driven appropriately to orient the platform properly. Resolvers are used at each pivot mount to measure the aircraft's pitch, roll, and heading. These measurements are supplied to cockpit displays and to the autopilot.

7.4.2 Gyros

Three gyros are mounted orthogonally on the INS platform with their input axes parallel with the platform axes. The gyros are used to measure the angular deviations of the platform about the platform axes. The deviation measurements are used to drive the torque motors to maintain the desired platform orientation. Since the gyros measure rotation with respect to inertial space, the compensation terms (see Section 7.3 of Appendix A) are needed to define the relationship of inertial space to the platform axes. To apply the compensation to the gyro measurements, each gyro is provided with a gyro-torquer, which commands the orientation of the gyro's input axis according to the gyro-torquer input signal. In addition to providing a means of applying the compensation terms, the gyro-torquers are used in the initial alignment mode (discussed in Section 7.5.2 of Appendix A).

7.5 Operating Modes

INS operates in two basic modes:

- Navigation mode
- Initial alignment mode

The navigation mode is the normal operating mode, in which navigation data are provided by the INS to the autopilot and cockpit displays. The initial alignment mode is required to orient the platform properly before the navigation mode is selected.

7.5.1 Navigation Mode

In the navigation mode, the INS output data includes the following:

- Current position
- Groundspeed
- Altitude
- Track angle
- · Wind speed
- Wind angle
- True heading
- Attitude (pitch and roll)
- Steering commands

Groundspeed and position displacement are derived from the acceleration vector measured by the accelerometers. This derivation can be thought of as integrating the acceleration vector twice. Current position is determined from initial position displacement. The vertical component of acceleration must be complemented with barometric altitude to provide stable outputs of vertical speed and altitude.

Track angle is the angle formed by the east and north components of groundspeed. Wind speed and wind angle are computed by comparing true airspeed (TAS) and true heading with groundspeed and track angle. True heading, pitch, and roll are provided by the resolvers used at the pivot mounts of the gimbal system. Whereas true heading is directly provided by the azimuth axis resolver in a north-pointing INS, the alpha angle must be added to the azimuth resolver measurement in a wander-azimuth INS to provide true heading. Steering commands are generated by comparing the current position with the desired course. The desired course is a great-circle track defined between waypoints entered into the INS.

7.5.2 Initial Alignment Mode

Before the navigation mode can be used, the initial alignment mode must be selected to orient the platform properly. Alignment comprises three phases:

- Manual entry of initial position
- Platform leveling
- Gyrocompassing

Initial position is used as a reference for computing compensation terms and for subsequent determination of position in the navigation mode. Platform leveling aligns the vertical axis of the platform with the direction of the gravity vector. Gyrocompassing is a technique for aligning the platform azimuth with true north. The entire alignment, which requires

from 8 to 15 minutes for completion, must be performed while the aircraft is stationary on the ground.

7.5.2.1 Initial Position

The initial latitude and longitude of the aircraft is manually entered into the INS. Initial latitude is used in the gyrocompassing phase of the alignment. Both the latitude and the longitude of the initial position are used later in the navigate mode as references for computing current position.

7.5.2.2 Platform Leveling

The platform is leveled with respect to the direction of the gravity vector ("up"). This places the X and Y axes (or east and north axes in a north-pointing INS) within the local horizontal plane. During the leveling phase, the components of gravity measured by the two horizontal accelerometers are monitored. The portion of gravity measured by the horizontal accelerometers is a function of unwanted platform tilt and is used to drive the gyro-torquers. The gyro-torquer signals cause the input axes of the gyros to rotate, producing angular deviation errors. These errors, measured by the gyros, are used to reposition the platform so that the platform tilt is removed. The olatform is level when the horizontal accelerometers no longer measure any component of the gravity vector.

7.5.2.3 Gyrocompassing

Gyrocompassing orients the platform azimuth with true north. The implementation in a north-pointing INS is different from that in a wander-azimuth INS. The two implementations will be discussed separately.

North-Pointing

The platform is rotated about the azimuth axis to align the north platform axis with true north. The earth-rotation compensation rates (discussed in Section 7.3 of Appendix A) are applied to the appropriate platform axes. Any misalignment between the north platform axis and true north, however, causes the compensation to be misapplied, thus tilting the platform. The magnitude and direction of tilt are detected by the horizontal accelerometers. This information is used to rotate the platform about the azimuth axis so that the misalignment between the platform north axis and true north is removed. The platform is rotated by applying the detected tilt signal to the azimuth gyro-torquer. The resulting gyro error signal is then used to rotate the platform.

Wander-Azimuth

In a wander-azimuth INS, the platform "Z" axis is slaved to the aircraft heading during the gyrocompassing phase. The earth-rotation compensation is applied with the alpha angle assumed to be zero (i.e., aircraft heading is exactly true north). A nonzero alpha angle results in misapplied earth-rotation compensation and causes the platform to tilt. The

tilt rate, measured by the horizontal accelerometers, is used to estimate the actual alpha angle. A correct estimate of the alpha angle will cause the earth-rotation compensation to be correctly applied; therefore there is no further tilt in the platform.

In both the north-pointing and the wander-azimuth INSs, a fine-align mode is entered in which the platform tilt incurred during the gyrocompassing process is removed and further gyrocompassing is executed to obtain a precise true-north reference.

Some systems calculate the initial latitude, which is a function of the total tilt incurred during gyrocompassing. The computed latitude can then be used in further fine alignment and as a reference for a reasonableness check of entered initial latitude.

7.6 Operational Characteristics

Since INS is a self-contained navigation system, it can be used in over-water navigation and in remote areas in which VOR/DME coverage is inadequate. Although the position and groundspeed errors increase as a function of time, the navigation accuracy of INS is sufficient for most over-water flights of long duration without requiring position updates from a radionavigation system. However, to further enhance the accuracy, most INSs provide some form of update capability. Some provide a manual update capability in which the pilot can insert values of latitude and longitude for current position. Other systems mix DME/DME RNAV information with INS data, using the INS to smooth noisy DME/DME derived position. Still other INSs provide interface capability between the INS and an Omega receiver. The Omega unit automatically provides a position update when the Omega accuracy is estimated to be suitable, and the INS provides position data between updates. The INS can also provide position data during periods of inadequate Omega coverage, such as during scheduled transmitter maintenance or when the aircraft is within 300 nm of a transmitter station upon which the Omega navigation process is dependent.

8. STRAPDOWN INS

8.1 History of Development

Strapdown inertial navigation system (INS) development began in the late 1950s. Significant advances were impeded until recent years by technology limitations in gyros and computers. Development of nonconventional gyros such as the ring laser gyro and the electrostatically suspended gyro has overcome gyro dynamic range limitation. Recent minicomputer technology has supplied a throughput capability that is more than adequate for strapdown INS.

The accuracy capability of strapdown INS was first demonstrated to be suitable for aircraft navigation when the U.S. Air Force conducted evaluation flights at Holloman Air Force Base in 1975. The first civil strapdown INS applications will comprise a strapdown inertial reference system (IRS) and the use of a flight management computer (FMC). This capability will be commercially available in the early 1980s.

8.2 System Overview

Γ

1

A strapdown INS is a self-contained navigation system with basic principles similar to those of conventional INS. Accelerometers are used to measure the aircraft's acceleration, which is then integrated twice to calculate the aircraft's velocity and position displacement. Current position is then determined by relating the position displacement to initial position.

The accelerometers are mounted in fixed reference to the aircraft frame so that their input axes are aligned with the aircraft body-axes. The acceleration components measured by the accelerometers must be transformed into the locally level geographic coordinate frame before velocity and position displacement can be determined. The transformation is performed with gyros to measure the total angular attitude changes of the aircraft. The navigation accuracy of a strapdown INS is highly dependent on the accuracy of the gyros.

The initial orientation of the aircraft body-axes with respect to the locally level geographic coordinate frame is established during the initial alignment mode, while the aircraft is stationary on the ground prior to flight. During the initial alignment mode, the pilot manually enters the initial latitude and longitude.

8.3 Strapdown INS Gyros

The gyros used in a strapdown INS must be capable of measuring the total range of aircraft attitudes. In a conventional INS, the gyros are required to measure only the small incremental deviations of the platform and are not appropriate for use in a strapdown INS. Three types of gyros have been developed that are particularly suited for use in strapdown INS applications:

- Ring laser gyro
- Electrostatically suspended gyro
- Tuned-rotor gyro

8.3.1 Ring Laser Gyro

The ring laser gyro (RLG) can be used to measure the angular rate about a single axis. The RLG has no spinning mass and has the potential for very high reliability compared with gyros that do have a spinning mass. The operating principle of the RLG is based on relativistic principles first demonstrated in 1963.

The RLG uses two laser beams traveling around a single closed path in opposite directions from each other. If the RLG is rotated with respect to inertial space in the direction of one of the beams, the theory of general relativity predicts that the length of the closed path will decrease in the direction of rotation and increase in the opposite direction, in proportion to the rate of rotation. Since the oscillation frequency of a

X-43

laser is proportional to the length of the path along which it travels, the changes in path length due to rotation will cause the frequency of each beam to change in proportion to the rotation rate. Since the frequency of one beam increases while that of the other decreases, a difference frequency can be measured that is also proportional to the rotation rate. By applying the appropriate scale factor to the difference frequency measurement, the rotation rate can be determined.

The laser path is either a triangular or square shape, with reflecting surfaces at each corner. The reflecting surface at one corner allows a small amount of energy from each of the two beams to pass through to a special difference frequency detector. The size of the laser path and the quality of the reflecting surfaces affect the overall accuracy capability of the RIG. Generally, increasing the path length and improving the quality of the reflecting surface will improve the accuracy of the angular rate measurement capability.

A condition called lock-in occurs in an RLG because of interaction between the two beams. Lock-in places a lower bound on the measurable angular rate when it occurs. Lock-in can be prevented by mechanically dithering the entire laser cavity enclosure. Dithering is a small rotational vibration that has a zero angular mean. This technique is commonly used in most RLGs currently being developed to prevent lock-in.

8.3.2 Electrostatically Suspended Gyro

The electrostatically suspended gyro (ESG) can be used to measure large angular deviations about two orthogonal axes. ESG technology was first demonstrated in the late 1950s.

An ESG uses an electrostatic field to suspend a spherical rotor in a vacuum. The rotor is spun by rotating the field at a very high rate. The rotor is made to wobble as it spins by offsetting the center of mass slightly from the geometric center of the sphere. Modulation in the electrostatic field caused by the wobbling is detected by sensors mounted in the enclosure. The direction of the modulation is determined from the sensor signals and is a measure of the relative orientation of the rotor spin axis and the enclosure. Since there is no physical contact with the rotor, no unwanted torques are introduced. The ESG therefore exhibits a very low erroneous drift rate.

Because of the low drift rate capability, the ESG is also being developed for use in conventional INS military applications that require a high level of accuracy. The angle detection technique used in some of these developments uses an optical sensor. Rather than using a wobbling rotor, this technique uses a rotor with an optical pattern etched into it that is detected by the optical sensor.

8.3.3 Tuned-Rotor Gyro

The tuned-rotor gyro is a mechanical gyro with a spinning mass. It was originally developed for use in conventional INS, in which it measures

only small incremental angular deviations. By means of a special caging technique, the tuned-rotor gyro can measure angular rate, so it is suitable for use in strapdown INS.

The spinning mass (i.e., rotor) is connected to the shaft of the spinmotor by a special flexible coupling. The plane of the rotor is free to rotate slightly about the two axes perpendicular to the spin axis. Angular deviations of the rotor about these axes are detected by sensors attached to the gyro case. The gyro contains a two-axis electric torquer that can be used to introduce calibrated torques into the rotor. The rotor orientation can thus be commanded by an electric signal, which drives the torquer. The torquer is used to electrically servo the rotor to null, using the deviation signals from the case-mounted sensors. The signal level required to drive the torquer is, then, proportional to the angular rate of the gyro case. By using the appropriate scale factor, this signal provides a measure of angular rate. Using the torquer in this manner is called caging.

8.4 Navigation Mode

Three accelerometers are mounted orthogonally to the aircraft frame, with their input axes parallel to the aircraft body-axes. Aircraft acceleration components are therefore measured in body-axis coordinates and must be transformed into the locally level geographic coordinate frame in order to determine aircraft position.

A transformation matrix is used to perform the acceleration vector transformation and defines the relationship between the body-axis and locally level geographic coordinate frames. This relationship is defined in two stages. First, the relationship between the aircraft body-axis coordinate frame and inertial space is defined by the angular measurements of the gyros. Second, the relationship between inertial space and the locally level geographic coordinate frame is defined in calculated earthrotation and transport-rate compensation terms. (Compensation terms are discussed in detail in Section 7.3 of Appendix A). These two relationships are then used to determine and continuously update the transformation matrix.

Once the acceleration vector is transformed from aircraft body-axis coordinates to the locally level geographic coordinate frame by means of the transformation matrix, it is integrated twice to obtain the aircraft's velocity vector and position displacement. Subsequent computations of the strapdown INS output data are identical to those described in Section 7.5.1 of Appendix A for conventional INS, except for the heading and attitude outputs, which are derived from the gyro input data in a strapdown INS.

8.5 Initial Alignment Mode

Before the strapdown INS can be used to generate navigation information, it must be initialized while the aircraft is stationary on the ground. This process is analogous to the initial alignment mode in the conventional INS. It requires up to 10 minutes. Initial latitude and longitude is entered into the strapdown INS. The latitude is used to estimate the earth rate, which is then compared with that measured by the gyros. The error, together with the gravity vector measured by the accelerometers, is used to initialize the transformation matrix by means of an iterative technique. When the transformation matrix is properly initialized, no error will be detected between the computed and measured earth rate components.

8.6 Operational Characteristics

Strapdown INS is operationally the same as conventional INS, with the addition of acceleration output data that are referenced to body-axis coordinates. Strapdown INS is more reliable than conventional INS, since it has few moving mechanical parts.

APPENDIX B

EQUATIONS

This appendix contains the equations used to obtain the values of navigational errors specified in this report.

1. DEFINITION OF TERMS

E

E

Ε

Ľ

Ľ

Ē

L

E

Γ

1

- ϕ_1 = Latitude of origin in degrees
- λ_1 = Longitude of origin in degrees

 ϕ_2 = Latitude of destination in degrees

 λ_2 = Longitude of destination in degrees

$$\overline{\phi} = \left(\frac{\phi_1 + \phi_2}{2}\right)$$

$$DLo = \lambda_2 - \lambda_1$$

DLox = Interval of longitude measured from point of departure in degrees

2. RHUMB-LINE EQUATIONS

Rhumb-line course angle (a in degrees):

$$\alpha = \tan^{-1} \left(\frac{\phi_2 - \phi_1}{DLo \cos \overline{\phi}} \right)$$

Latitude of points on the rhumb-line track ($\phi_{\rm RL}$ in degrees):

 $\phi_{RL} = \phi_1 + DLox \cos \phi \tan \alpha$

B-1

Rhumb-line distance (R in nautical miles):

$$R = 60 \sqrt{\left[\left(\phi_2 - \phi_1 \right)^2 + \left(DLO \ \cos \overline{\phi} \right)^2 \right]}$$

3. GREAT-CIRCLE EQUATIONS

Initial great-circle course angle (C in degrees):

$$C = \tan^{-1} \left[\frac{\sin(DLo)}{\cos\phi_1 \tan\phi_2 - \sin\phi_1 \cos(DLo)} \right]$$

Latitude of the great-circle vertex (Lv in degrees):

The vertex of a great circle is defined as the point of greatest latitude.

$$Lv = \cos^{-1}[\cos\phi_1 \sin(C)]$$

Difference of longitude between the vertex and the point of departure (DLov in degrees):

 $DLov = \sin^{-1} \left[\frac{\cos(C)}{\sin(Lv)} \right]$

Latitudes of points on the great-circle track (ϕ_{GC} in degrees):

 $\phi_{GC} = \tan^{-1} [\tan(Lv) \cos(DLox - DLov)]$

Great-circle distance (D in nautical miles):

 $D = 60 \times \cos^{-1} [\sin\phi_1 \sin\phi_2 + \cos\phi_1 \cos\phi_2 \cos(DLo)]$

4. GEODESIC ERROR EQUATIONS

Difference between a spherical-earth model and a Clarke 1866 ellipsoid model:

$$E_{x} = \left[9.12951 \cos\overline{\phi} - 2.92495 \cos(3\overline{\phi})\right] \left(\lambda_{2} - \lambda_{1}\right) \left(\frac{\pi}{180}\right)$$

$$E_{y} = 0.37414 \left(\phi_{2} - \phi_{1}\right) \left(\frac{\pi}{180}\right) - 8.88543 \left[\sin\left(2\phi_{2}\right) - \sin\left(2\phi_{1}\right)\right]$$

` **B**−2

where

Ľ

T

1

 $\left(\right)$

Ū

E = East error in nautical miles

Tell Manager and the second second second

NAMES IN COMMENT

 E_y = North error in nautical miles

APPENDIX C

COMPUTATIONS

The following paragraphs illustrate the application of equations (given in Appendix B) to determine the maximum cross-track deviation between the rhumb-line and great-circle paths between New York and Los Angeles as defined in Figure 5-1. The cross-track deviation between the two paths is defined as the length of the line drawn perpendicular from the rhumb-line path to the great-circle path.

£

1.

The rhumb-line course angle is determined from the following equation:

$$\alpha = \tan^{-1} \left[\frac{\phi_2 - \phi_1}{(\lambda_2 - \lambda_1)\cos(\frac{\phi_2 + \phi_1}{2})} \right].$$

$$\alpha = \tan^{-1} \left[\frac{-6.698}{(44.629)\cos(37.291)} \right] = -10.683^{\circ}$$

This course angle is relative to the predominant cardinal direction of travel, which in the case of JFK to LAX is west.

The maximum deviation in latitude between the rhumb-line path and the great-circle path occurs at a longitude of 93.778°. The latitude corresponding to this longitude along the rhumb-line path is found from the following equation:

$$\phi_{RL} = \phi_1 + (\lambda - \lambda_1) \cos \left(\frac{-\phi_1 + \phi_2}{2} \right) \tan \alpha \qquad (1)$$

where $\lambda = 93.778^{\circ}$ and α is defined as the angle whose tangent is the change in latitude divided by the change in longitude, also taking into account the convergence of the meridians. The course angle of -10.683° computed earlier is the appropriate angle to be used as α in the above equation. Hence,

C-1

The line of maximum cross-track deviation, a rhumb-line drawn perpendicular to the rhumb-line path toward the great-circle path, makes an angle of 79.317° (90° - 10.683°) with respect to west. The latitude and longitude of the point on the line of cross-track deviation which intersects the rhumb-line path have been defined as 37.64° and 93.778° respectively. Therefore, the latitude of any point on the line of cross-track deviation ($\phi_{\rm XTRK}$) can be determined by specifying a longitude ($\lambda_{\rm XTRK}$) in equation (1). In this case, the appropriate value for α is 79.317°.

$$\phi_{\text{XTRK}} = 37.64^\circ = (\lambda_{\text{XTRK}} - 93.778^\circ) \cos\left(\frac{37.64^\circ + \Psi_{\text{XTRK}}}{2}\right) \tan(79.317^\circ)$$
(2)

The latitude (ϕ_{GC}) along a great-circle path corresponding to a given longitude (λ) is defined as:

$$\phi_{GC} = \tan^{-1} \left[\tan(Lv) \cos(\lambda - \lambda_1 - DLov) \right]$$

For the great-circle path between JFK and LAX,

$$Lv = \cos^{-1}(\cos\phi_1 \sin C) = 40.79^{\circ}$$

where

$$C = \tan^{-1} \left[\frac{\sin (\lambda_2 - \lambda_1)}{(\cos\phi_1 \tan\phi_2) - (\sin\phi_1 \cos(\lambda_2 - \lambda_1))} \right] = 86.1564^{\circ}$$

$$DLov = \sin^{-1} \left(\frac{\cos C}{\sin Lv} \right) = 5.8894^{\circ}$$

$$Thus \phi_{GC} = \tan^{-1} \left[0.8629 \times \cos (\lambda - 79.6674) \right]$$
(3)

At the point of intersection of the rhumb line of cross-track deviation with the great-circle path, the solution of the great-circle and rhumb-line equations in terms of latitude as a function of longitude will be equal. We therefore have two equations with two unknowns (equations 2 and 3). Unfortunately, a closed-form solution of these two equations does not exist, so a graphical or iterative solution is required. Convergence of an iterative solution occurs at a longitude of 94.31° and a corresponding latitude of 39.85°.

Hence, the maximum cross-track deviation between the rhumb-line and great-circle paths between New York and Los Angeles is found to be the following:

Cross-track deviation (nm) = $60 \sqrt{(\Delta \phi)^2 + (\Delta \lambda \cos \phi)^2}$

C-2

 $\Delta \phi = 39.85^{\circ} - 37.64^{\circ} = 2.21^{\circ}$ $\Delta \lambda = 94.31^{\circ} - 93.78^{\circ} = 0.53^{\circ}$ $\overline{\phi} = \frac{39.85^{\circ} + 37.64^{\circ}}{2} = 38.75^{\circ}$

Cross-track deviation = 134.90 nm

where

L

C-3

APPENDIX D

REFERENCES AND BIBLIOGRAPHY

Г

- 1. Bowditch, N., American Practical Navigator. Defense Mapping Agency Hydrographic Center, 1977.
- 2. Kayton, M., and Fried, W., Avionics Navigation Systems. John Wiley and Sons, Inc., New York, 1969.
- Milliken, R. J., and Zoller, C. J., "Principle of Operation of NAVSTAR and System Characteristics," Navigation, Vol. 25, No. 2, 1978.
- 4. Easton, R. L., "The Navigation Technology Program," Navigation, Vol. 25, No. 2, 1978.
- 5. Spilker, J. J., Jr., "GPS Signal Structure and Performance Characteristics," *Navigation*, Vol. 25, No. 2, 1978.
- Van Dierendonck, A. J., Russell, S. S., Kopitzke, E. R., and Birnbaum, M., "The GPS Navigation Message," Navigation, Vol. 25, No. 2, 1978.
- 7. Russell, S. S., and Schaibly, J. H., "Control Segment and User Performance," Navigation, Vol. 25, No. 2, 1978.
- B. Glazer, B. G., "GPS Receiver Operation," Navigation, Vol. 25, No. 2, 1978.
- 9. Martin, E. H., "GPS User Equipment Error Models," Navigation, Vol. 25, No. 2, 1978.
- Johnson, C., and Ward, P., "GPS Application to Seismic Oil Exploration," Navigation, Vol. 26, No. 2, 1979.
- 11. Braff, R., Shively, C., and Bradley, J., Navigation System Integrity and Reliability for Civil Aviation. The MITRE Corporation, McLean, Virginia. Paper presented at IEEE Position Location and Navigation Symposium, Atlantic City, New Jersey, 8 to 11 December 1980.
- Shively, C., "Use of Altimeter with GPS 18 Satellite System." The MITRE Corporation, McLean, Virginia. Letter to Jerry Bradley, W47-0641, 22 May 1980.
- 13. United States (U.S.) National Aviation Standard for the Very High Frequency Omnidirectional Radio Range (VOR)/Distance Measuring Equipment (DME)/Tactical Air Navigation (TACAN), FAA Order 9840.

D-1

- Meer, S., Study of the VORTAC System and Its Growth Potential. The MITRE Corporation, McLean, Virginia, Publication FAA-EM-74-9, November 1973.
- 15. An Introduction to Omega -- A Worldwide Navigation System. Litton Aero Products, Canoga Park, California, 30 January 1978.
- 16. ONEGA Navigation. Canadian Marconi Company, Avionics Division, Montreal.
- 17. Kasper, J., and Hutchinson, C., "The Omega Navigation System -- An Overview," Ortung und Navigation, February 1978.
- Aviation Use of Omega. ICAO Circular 139-AN/95/2, Second Edition, 1979.
- 19. Loran-C User Handbook. Department of Transportation, United States Coast Guard, May 1980.
- 20. Van Etten, J. P., LORAN-C as an International Aid to Navigation. Paper presented at International Conference on Electronic Navigational Aid Systems for Aircraft, Paris, 14 to 18 November 1977.
- 21. Radionavigation Systems. Department of Transportation, United States Coast Guard, 1980.
- 22. Federal Radionavigation Plan, Vols. I-IV. Department of Transportation, Publication DOT-TSC-RSPA-80-16, July 1980.
- 23. Approval of Area Navigation Systems for Use in the U.S. National Airspace System. FAA Advisory Circular AC-90-45A, 21 February 1975.
- 24. What is DME? Instruction Guide, Rockwell International, Collins Air Transport Division, May 1979.
- 25. Vogel, T. J., Horizontal Datums for Nautical Charts. DMA Hydrographic/ Topographic Center, Washington, D.C., July 1980.
- 26. U.S. National Aviation Standard for the VORTAC System. FAA Advisory Circular AC-00-31, 10 June 1970.
- 27. Enroute Low Altitude -- U.S. Flight Information Publication. National Oceanic and Atmospheric Administration, Chart L-10, 18 May 1978.
- 28. Dodge, S. M., A Comparative Analysis of Area Navigation Systems in General Aviation. Massachusetts Institute of Technology, NASA Publication CR-132504, June 1973.
- 29. Economic Requirements Analysis of Civil Air Navigation Alternatives. Systems Control, Inc., Publication FAA-ASP-78-3, April 1978.
- 30. Economic Analysis of Future Civil Air Navigation Systems. MITRE Corporation, METREK Division, Publication FAA-EM-78-6, December 1977.
- 31. Air Traffic Control. Department of Transportation, FAA Air Traffic Service, Publication 7110.65B, 1 January 1980.

D-2

- 32. Laskowski, E., Terrain and Neather Effects on Doppler Radar Navigation Systems, FAA Report No. RD-64-15, January 1964.
- Mayer, Robert H., Study of the Performance Characteristics of the Bendix Type DRA-12 Airborne Doppler Radar System, FAA Report No. RD-70-73, November 1970.
- 34. "DRA-12 Doppler Radar System Maintenance Manual," Bendix Avionics Division, January 1971.
- 35. "Airborne Doppler Radar," ARINC Characteristics 540, Aeronautical Radio, Inc., 1 June 1958.
- 36. "Doppler Radar Mark I Computer," ARINC Characteristic 543, Aeronautical Radio, Inc., 6 February 1959.

Ē

E

- 37. Macomber, G. R., Inertial Guidance Engineering, Prentice Hall, 1962.
- 38. "Technical Description Litton LTN-51 Inertial Navigation System," Litton Aero Products, December 1972.
- 39. "System Technical Description Carousel IV & IVA Inertial Navigation Systems," Delco Electronics, March 1974.
- 40. "Air Transport Inertial Navigation System INS," ARINC Characteristic 561-11, Aeronautical Radio, Inc., 17 January 1975.
- 41. Savage, Paul G., "Strapdown Sensors," AGARD Lecture Series No. 95, June 1978.
- 42. Killpatrick, Joseph, "The Laser Gyro," IEEE Spectrum, October 1967.
- 43. "Preliminary Technical Description of LTN-90 Inertial Reference System," Litton Aero Products, February 1981.
- 44. "Inertial Reference System," ARINC Characteristic 704-1, Aeronautical Radio, Inc., 1 April 1980.
- 45. Swanson, E. R., et al, "Differential Omega Navigation for the U.S. Coastal Confluence Region. Part I: Overview," Naval Electronics Laboratory Center, Publication AD-777 041, 2 January 1974.
- 46. Swanson, E. R., et al, "Differential Omega Navigation for the U.S. Coastal Confluence Region. Part II: Requirements, Accuracies, and System Definition," Naval Electronics Laboratory Center, Publication AD-777 043, 2 January 1974.
- Swanson, E. R., "Omega Chart Base Selection for Differential and Standard Use," Naval Ocean Systems Center, Publication AD A070380, 21 March 1978.
- 48. "Performance Standards for Differential Omega Correction Transmitting Stations," Inter-Governmental Maritime Consultative Organization, Resolution A.425(XI), 15 November 1979.
- 49. Foltz, J. M., et al, "Differential Omega in the Alaska/Yukon Region: An Analytic Assessment," the Analytic Sciences Corporation, Publication AD A078517, August 1979.

D-3

50. Mackenzie, F. D., and Haroules, G. G., "Preliminary Results of Loran-C Flight Testing in the State of Vermont," Department of Transportation, Publication DOT-TSC-RSPA-80-21, November 1980. 「日本のため」

I

1

]

]

Ţ

I

Ţ

I

]

]

]

「

ł